# UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

\_\_\_\_\_

### FORM 8-K

#### **CURRENT REPORT**

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934.

Date of Report: (Date of Earliest Event Reported): December 6, 2021

### PHX MINERALS INC.

(Exact name of registrant as specified in its charter)

OKLAHOMA	001-31759	73-1055775
(State or other jurisdiction of incorporation)	(Commission File Number)	(I.R.S. Employer Identification No.)
1601 NW Expressway,		
Suite 1100		
Oklahoma City, OK		73118
(Address of principal executive offices)		(Zip code)
(Regis	(405) 948-1560 strant's telephone number including area	code)
(Former na	Not Applicable ame or former address if changed since l	ast report)
	e Form 8-K filing is intended to simultaning provisions (see General Instruction A	
<u> </u>	suant to Rule 425 under the Securities Ac	
	o Rule 14a-12 under the Exchange Act ( ications pursuant to Rule 14d-2(b) under	,
* **	ications pursuant to Rule 13e-4(c) under	the Exchange Act (17 CFR 240.13e-
Securities registered pursuant in Section	on 12(b) of the Act:	
Title of each class		Name of each exchange on which
Class A Common Stock, \$0.01666	par value Symbol(s) PHX	registered New York Stock Exchange
	gistrant is an emerging growth company is chapter) or Rule 12b-2 of the Securition	
Emerging growth company		
	rate by check mark if the registrant has el ny new or revised financial accounting s	

#### Item 1.01 Entry into a Material Definitive Agreement.

#### Entry into Purchase and Sale Agreements

On December 6, 2021, PHX Minerals Inc. (the "Company") entered into two separate Purchase and Sale Agreements (collectively, the "Purchase Agreements") with affiliated sellers (the "Sellers") to acquire certain mineral interests, royalty interests and overriding royalty interests underlying certain lands located in Caddo Parish, Louisiana (the "Assets"). The Company entered into one Purchase Agreement with Merrimac Properties Partners, LLC and Quarter Horse Energy Partners, LLC (the "Merrimac Purchase Agreement") to acquire a portion of the Assets for consideration equal to \$5,185,475 in cash, and a separate Purchase Agreement with Palmetto Investment Partners II, LLC (the "Palmetto Purchase Agreement") to acquire the remainder of the Assets for consideration equal to \$601,797 in cash. The Assets include mineral and royalty interests totaling approximately 426 net royalty acres in the Haynesville play. The transactions contemplated in the Purchase Agreements are conditioned upon both Purchase Agreements closing concurrently and are expected to close by December 15, 2021. As of December 6, 2021, the Sellers and their affiliates collectively own 3,549,207 shares of common stock of the Company, or approximately 10.8% of the issued and outstanding shares of common stock of the Company.

The terms and conditions of both Purchase Agreements are substantially similar, and the purchase price under each Purchase Agreement is subject to customary adjustments, including adjustments based on due diligence performed by the Company prior to closing. Both Purchase Agreements contain customary representations, warranties, covenants and indemnities by each of the applicable parties thereto. The obligations of the Company and the Sellers to close each acquisition is subject to certain customary closing conditions as set forth in the Purchase Agreements. There can be no assurance that the conditions to closing the acquisitions of the Assets will be satisfied.

The above description of the material terms and conditions of the Purchase Agreements does not purport to be complete and is qualified in its entirety by reference to the full text of the Merrimac Purchase Agreement, which is filed as Exhibit 10.1 hereto, and the Palmetto Purchase Agreement, which is filed as Exhibit 10.2 hereto.

#### First Amendment to Credit Agreement

On December 6, 2021 (the "Amendment Closing Date"), the Company entered into the First Amendment to Credit Agreement (the "Amendment"), which amends and modifies the Credit Agreement dated as of September 1, 2021 among the Company, Independent Bank, as Administrative Agent and L/C Issuer, and certain other lenders (the "Credit Agreement").

The Amendment provides for an increase to the Company's Borrowing Base from \$27.5 million to \$32.0 million effective on the Amendment Closing Date. The Borrowing Base will remain at \$32.0 million until the next scheduled semi-annual redetermination, which is scheduled to occur on or about June 1, 2022, unless otherwise redetermined pursuant to an Unscheduled Redetermination. In addition, the Amendment amends the commitment schedule to reallocate the Committed Sum and Commitment Percentage of each Lender under the Credit Agreement. All capitalized terms in this description of the Amendment that are not otherwise defined in this Current Report on Form 8-K have the meaning assigned to them in the Credit Agreement.

The above description of the material terms and conditions of the Amendment does not purport to be complete and is qualified in its entirety by reference to the full text of the Amendment, which is filed as Exhibit 10.3 hereto.

### Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

The information set forth under Item 1.01 of this Current Report with respect to the Amendment is incorporated herein by reference.

#### Item 7.01 Regulation FD Disclosure.

On December 9, 2021, the Company issued a press release announcing the Company's entry into the Purchase Agreements and the Amendment. A copy of the press release is attached as Exhibit 99.1 hereto and is incorporated herein by reference.

In accordance with General Instruction B.2 of Form 8-K, the information in Item 7.01 of this Current Report on Form 8-K, including the attached Exhibit 99.1, is being furnished pursuant to Item 7.01 and shall not be deemed to be "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liabilities of that section, and shall not be deemed to be incorporated by reference into any of the Company's filings under the Securities Act of 1933, as amended, or the Exchange Act, whether made before or after the date hereof and regardless of any general incorporation language in such filings, except to the extent expressly set forth by specific reference in such a filing.

#### Item 9.01 Financial Statements and Exhibits.

#### (d) Exhibits

#### **Exhibit Description** No. Purchase and Sale Agreement, dated December 6, 2021, by and among Merrimac Properties Partners, 10.1\* LLC and Quarter Horse Energy Partners, LLC, as Sellers, and PHX Minerals Inc., as Buyer. Purchase and Sale Agreement, dated December 6, 2021, by and between Palmetto Investment Partners II, 10.2\* LLC, as Seller, and PHX Minerals Inc., as Buyer. First Amendment to Credit Agreement dated as of December 6, 2021, by and among PHX Minerals Inc., 10.3 each lender party thereto, and Independent Bank, as Administrative Agent and L/C Issuer. Press Release of PHX Minerals Inc. dated December 9, 2021. 99.1 104 Cover Page Interactive Data File (embedded within the Inline XBRL document).

<sup>\*</sup> The Purchase and Sale Agreement contains schedules and exhibits that have been omitted pursuant to Item 601(a)(5) of Regulation S-K. The Company agrees to furnish a supplemental copy of any such omitted exhibit or schedule to the Securities and Exchange Commission upon request.

#### **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

#### PHX MINERALS INC.

By: /s/ Chad L. Stephens

Chad L. Stephens Chief Executive Officer

DATE: <u>December 9, 2021</u>

### PURCHASE AND SALE AGREEMENT

**BY AND AMONG** 

MERRIMAC PROPERTIES PARTNERS, LLC

**AND** 

QUARTER HORSE ENERGY PARTNERS, LLC

(as Sellers)

**AND** 

PHX MINERALS INC.

(as Buyer)

Dated as of: December 6, 2021

### **TABLE OF CONTENTS**

Article 1	PURCHASE AND SALE	4
1.1	Purchase and Sale.	4
1.2	Effective Time.	4
1.3	Excluded Assets	4
Article 2	PURCHASE PRICE; REVENUES AND EXPENSES	4
2.1	Purchase Price.	4
2.2	Allocation of Purchase Price.	4
2.3	Adjustments to the Purchase Price at Closing.	4
2.4	Adjustments to the Purchase Price Following Closing	5
2.5	Payment Method	7
2.6	Principles of Accounting.	7
	SCOPE OF DUE DILIGENCE AND TITLE DEFECTS	7
3.1	Scope of Due Diligence Review and Access	7
3.2	Title Defects and Purchase Price Adjustment.	7
		12
4.1		12
4.1	$\mathcal{E}$	12
4.3		12
4.4	•	13
4.5	•	13
4.6	C	13
4.7	1 7	13
4.8		13
4.9		13
4.10	, , , , ,	14
4.11		14
4.12	Payment of Royalties; No Suspense Accounts; No Deferred Payments	14
4.13	Environmental Matters.	14
4.14	No Cost-Bearing Interests.	15
4.15	· · · · · · · · · · · · · · · · · · ·	15
Article 5		15
5.1		15
5.2	$\mathcal{E}$	15
5.3		15
5.4		15
5.5	· · · · · · · · · · · · · · · · · · ·	16
5.6		16
5.7	$\mathbf{c}$	16
5.8		16
	1 2	
		16
6.1	$\epsilon$	16
6.2		16
6.3		17
6.4		17
6.5		17
6.6	·	18
		18
7.1	Transfer Taxes.	18
7.2	Asset Taxes	18

7.3	Tax Allocation.	
7.4	Refunds of Taxes.	
7.5	Cooperation.	
	CONDITIONS PRECEDENT AND TERMINATION	
8.1	Conditions to Obligations of Sellers	
8.2	Conditions to Obligations of Buyer.	
8.3	Frustration of Closing Conditions.	
8.4	Termination.	
8.5	Effect of Termination and Remedies.	
Article 9 (	CLOSING	
9.1	Date and Place of Closing.	
9.2	Closing Obligations.	. 23
Article 10	ASSUMPTION AND RETENTION OF OBLIGATIONS AND INDEMNIFICATION;	24
DISCLAII	MERS	
10.1	Buyer's Assumption of Liabilities and Obligations.	. 24
10.2	Indemnification	
10.3	Survival of Warranties, Representations and Covenants	26
10.4	Reservation as to Non-Parties	. 27
10.5	Acceptance of Title Condition.	
10.6	Express Negligence Rule	
10.7	Waiver of Certain Damages.	
10.8	Redhibition Waiver	
10.9	Determination of Losses.	
10.10	Notice and Defense.	
	MISCELLANEOUS	
11.1	Expenses.	
11.2	Notices	
11.3	Governing Law; Jurisdiction; Venue: Jury Waiver; Binding Arbitration.	
11.4	Further Assurances.	
11.5	Amendments: Waivers.	
11.6	Assignment.	
11.7	Counterparts/Fax Signatures.	
11.7	Construction.	
11.9	Entire Agreement.	
11.10	Successors and Permitted Assigns.	
	Parties in Interest.	
	Severability	
	Joint Preparation.	
	Publicity	
	Disclosure Schedules.	
	Specific Performance	
	Confidentiality.	
11.18	No Recourse.	. 33
ANNEX -	- DEFINED TERMS	A

### **List of Exhibits**

Exhibit A-1	Tracts and Allocated Values
Exhibit B	Form of Conveyance
Exhibit C	Form of Cartificate of Non For

Exhibit C Form of Certificate of Non-Foreign Status

### Exhibit D Leases

### <u>List of Disclosure Schedules</u>

Section 4.8	Litigation
Section 4.9	Taxes
Section 4.11	Contracts
Section 4.13	<b>Environmental Matters</b>
Section 4.14	No Cost-Bearing Interests

#### PURCHASE AND SALE AGREEMENT

This Purchase and Sale Agreement (the "Agreement") is dated as of this 6<sup>th</sup> day of December, 2021 (the "Execution Date"), and is by and among Merrimac Properties Partners, LLC a Delaware limited liability company ("Merrimac"), Quarter Horse Energy Partners, LLC, a Delaware limited liability company ("QHEP"), (each, a "Seller," and collectively, the "Sellers"), and PHX Minerals Inc., an Oklahoma corporation ("Buyer"). Sellers and Buyer are each individually referred to in this Agreement as a "Party" and collectively as the "Parties."

#### RECITALS

WHEREAS, Sellers own certain rights, title and interests in and to certain mineral interests, royalty interests and overriding royalty interests in the oil, gas, and other minerals underlying certain lands located in Caddo Parish, Louisiana, and

WHEREAS, Sellers desire to sell such rights, title and interests to Buyer, and Buyer desires to purchase such rights, titles and interests from Sellers on the terms and subject to the conditions set forth in this Agreement.

NOW THEREFORE, in consideration of the mutual promises, representations, warranties, covenants and agreements contained in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Buyer and Sellers hereby agree as follows:

#### Article 1 PURCHASE AND SALE

- 1.1 **Purchase and Sale.** Subject to the terms and conditions of this Agreement, at the Closing, Buyer agrees to purchase from each Seller, and each Seller agrees to sell, assign, transfer, convey and deliver to Buyer, all of such Seller's right, title and interest in and to the Assets for the Purchase Price specified in <u>Article 2</u>.
- 1.2 **Effective Time.** If the Closing occurs, the purchase and sale of the Assets shall be effective as of the date the respective Seller was vested with its interest in the respective Assets (the "**Effective Time**").
- 1.3 **Excluded Assets.** Notwithstanding any other provision of this Agreement to the contrary, the Assets to be conveyed and assigned under this Agreement do not include the Excluded Assets, all of which are reserved by Sellers.

# Article 2 PURCHASE PRICE; REVENUES AND EXPENSES

- 2.1 **Purchase Price.** Subject to the adjustments set forth below and the other terms and conditions of this Agreement, the total consideration for the purchase, sale and conveyance of the Assets to be paid by Buyer to Sellers is Five Million One Hundred Eighty-Five Thousand Four Hundred Seventy-Five Dollars and Forty-Six Cents (\$5,185,475.46) (the "**Purchase Price**").
- 2.2 **Allocation of Purchase Price.** The unadjusted Purchase Price shall be allocated among each tract of Land identified on Exhibit A-1 (each, a "Tract"), with each such Tract having the "Allocated Value" set forth for such Tract on Exhibit A-1.

#### 2.3 Adjustments to the Purchase Price at Closing.

- At Closing, without limiting Section 2.4, the Purchase Price will be adjusted in accordance with this Section 2.3. No later than three (3) Business Days prior to the Closing Date, Sellers will in good faith prepare (as a reasonable and prudent Person) and provide to Buyer a preliminary settlement statement (the "Preliminary Settlement Statement") identifying all adjustments to the Purchase Price, without duplication, known by Sellers to be made at Closing, in accordance with this Agreement and as of the time of the preparation of the Preliminary Settlement Statement (including reasonable estimates of any such amounts where actual amounts are not available). After the delivery of the Preliminary Settlement Statement to Buyer and prior to the Closing, Sellers shall consult in good faith with Buyer regarding any reasonable changes to the Preliminary Settlement Statement proposed by Buyer prior to the Closing. Without limiting Section 2.4, the Preliminary Settlement Statement, as agreed upon in writing by Sellers and Buyer, will be used to adjust the Purchase Price at Closing; provided, however, that if Sellers and Buyer cannot agree in writing on the Preliminary Settlement Statement prior to Closing, then the Preliminary Settlement Statement proposed by Sellers, taking into account any adjustments thereto agreed to by the Parties, will be used to adjust the Purchase Price at Closing. Sellers and Buyer acknowledge that some items in the Preliminary Settlement Statement may be estimates or otherwise subject to change in the Final Settlement Statement.
  - (b) The Purchase Price will be increased (without duplication) by the following:
  - (i) the amount of any proceeds (including any lease bonuses, royalties and other proceeds) received by Buyer (and not otherwise paid or remitted to any Seller) to the extent attributable to (A) any of the Assets and (B) the time period before the Effective Time:
  - (ii) subject to <u>Section 3.2</u>, the Title Credit Amount of any Title Credit (if applicable);
  - (iii) the amount of all Asset Taxes allocable to Buyer pursuant to <u>Article 7</u> but paid or otherwise economically borne by Sellers; and
    - (iv) as otherwise agreed in writing by the Parties.
  - (c) The Purchase Price will be decreased (without duplication) by the following:
  - (i) the amount of any proceeds (including any lease bonuses, royalties and other proceeds) received by any Seller (and not otherwise paid or remitted to Buyer) to the extent attributable to (A) any of the Assets and (B) the period beginning on and following the Effective Time:
  - (ii) subject to <u>Section 3.2</u>, the Title Defect Amount of any uncured Title Defect that is not waived in writing by Buyer (if applicable);
  - (iii) the Allocated Value of the Assets excluded from the transactions contemplated hereby pursuant to Section 3.2(f);
  - (iv) the amount of all Asset Taxes allocable to Sellers pursuant to <u>Article 7</u> but paid or otherwise economically borne by Buyer; and
    - (v) as otherwise agreed in writing by the Parties.

#### 2.4 Adjustments to the Purchase Price Following Closing.

- (a) Final Settlement Statement. Within ninety (90) days after the Closing Date (such 90th day, the "Final Settlement Date"), Sellers will prepare a proposed final settlement statement (the "Final Settlement Statement"), which shall take into account all final adjustments made to the Purchase Price and shows the resulting final adjusted Purchase Price ("Final Price"). The Final Settlement Statement shall set forth the actual adjustments required under Section 2.3. As soon as practicable, and in any event within thirty (30) days, after receipt of the Final Settlement Statement (the "Review Period"), Buyer shall deliver to Sellers a written report containing any proposed changes to the Final Settlement Statement and an explanation of any such changes and the reasons therefor (the "Dispute Notice"). If Buyer fails to timely deliver a Dispute Notice to Sellers containing changes Buyer proposes to be made to the Final Settlement Statement, the Final Settlement Statement as delivered by Sellers will be deemed to be correct and mutually agreed upon by the Parties, and will, without limiting Section 7.2(c), be final and binding on the Parties and not subject to further audit or arbitration. If the Final Price set forth in the Final Settlement Statement is mutually agreed upon by Sellers and Buyer, the Final Settlement Statement and the Final Price shall, without limiting Section 7.2(c), be final and binding on the Parties hereto. Any disputed items included in a Dispute Notice regarding the Final Settlement Statement shall be resolved as provided in Section 2.4(c).
- (b) Payment of Post-Closing Adjustments. Any adjustments to the Purchase Price as reflected in the Final Settlement Statement will be offset against each other so that only one payment is required between the Parties. The Party owing payment will pay the other Party in cash the net post-Closing adjustment to the Purchase Price set forth in the Final Settlement Statement within five (5) Business Days following the date upon which all matters in the proposed Final Settlement Statement (excluding any Title Disputes) are either (i) agreed upon by the Sellers and Buyer in writing, (ii) deemed accepted by Buyer at the end of the Review Period pursuant to Section 2.4(a), or (iii) finally determined by the Accounting Consultant in accordance with Section 2.4(c), as applicable.
- Resolution of Disputed Items. After the completion and delivery of the proposed Final Settlement Statement, the Parties shall negotiate in good faith to attempt to reach an agreement in writing on the amount due with respect to any disputed items included in the Dispute Notice. If the Parties agree in writing on the amount due with respect to any disputed items in the proposed Final Settlement Statement, and any payment adjustment is required, the Party owing payment will pay the other Party in the time period provided in Section 2.4(b). If the Parties are unable to agree in writing on the amount due with respect to any disputed items in the proposed Final Settlement Statement within thirty (30) days after Sellers receive Buyer's written Dispute Notice, then Buyer and Sellers shall engage an independent national accounting firm mutually agreed upon by the Parties (the "Accounting Consultant") to exclusively and finally resolve any such disputed matters set forth in the Dispute Notice in accordance with the terms of this Agreement. Each of Buyer and Sellers shall within fifteen (15) Business Days after either Party's written request for arbitration summarize its position with regard to such dispute in a written document and submit such summaries to the Accounting Consultant, together with a copy of the Dispute Notice, the Final Settlement Statement, this Agreement and any other documentation such Party may desire to submit. If there is more than one or more Dispute Notice or disputed matter, all such disputes shall be consolidated into the same arbitration process pursuant to this Section 2.4(c). The Parties shall cooperate diligently with any reasonable request of the Accounting Consultant in an effort to resolve the matters submitted to the Accounting Consultant as soon as reasonably possible after the Accounting Consultant is engaged. The Accounting Consultant, once appointed, shall have no ex parte communications with any of the Parties concerning the determination required under this Section 2.4(c). All communications between any Party or its Affiliates and the Accounting Consultant shall be conducted in writing, with copies sent simultaneously to the other Parties in the same manner, or at a meeting or conference call to which each of the Parties and their respective Representatives have been invited and of which such Parties have been provided at least five (5) days' written notice. In no event shall the Accounting Consultant select an amount (i) for any

upward adjustment to the Purchase Price that is greater than Sellers' proposed amount or less than Buyer's proposed amount or (ii) for any downward adjustment to the Purchase Price that is greater than Buyer's proposed amount or less than Sellers' proposed amount. Within thirty (30) days after receipt of such materials from the Parties and after receipt of any additional information required by the Accounting Consultant, the Accounting Consultant shall make its determination, which shall be in accordance with the terms of this Agreement and shall be final and binding upon the Parties, without right of appeal, absent Fraud or manifest error. The Accounting Consultant may not award damages, interest or penalties to either Buyer or Sellers with respect to any matter. Sellers and Buyer shall each bear their own respective legal fees and other costs of presenting its case. Sellers shall bear fifty percent (50%) and Buyer shall bear fifty percent (50%) of the costs and expenses of the Accounting Consultant.

- (d) <u>Further Revenues and Expenses</u>. From and after the Closing Date, if a Party or Parties receive revenues that belong to another Party under this Agreement, the Party or Parties receiving the revenues agrees to remit those revenues to the applicable Party promptly (but no later than thirty (30) days) following such first Party's receipt thereof.
- 2.5 **Payment Method.** Unless the Parties otherwise agree in writing, all payments made or otherwise required under this Agreement (a) to Sellers, will be made in United States dollars by wire transfer in immediately available funds to such account(s) as are designated in writing by the Sellers to Buyer and (b) to Buyer, will be made in United States dollars by wire transfer in immediately available funds to such account(s) as are designated in writing by Buyer to Sellers.
- 2.6 **Principles of Accounting.** The Preliminary Settlement Statement and Final Settlement Statement will be prepared in accordance with GAAP, consistently applied in the petroleum industry, and applicable Laws.

## Article 3 SCOPE OF DUE DILIGENCE AND TITLE DEFECTS

3.1 **Scope of Due Diligence Review and Access.** In order to allow Buyer to conduct title due diligence with respect to the Assets, from and after the Execution Date until 5:00 P.M. Central Time on December 8, 2021 (the "**Title Claims Date**"), Sellers shall provide Buyer with reasonable access to the Records; provided, however, (a) such access does not unreasonably interfere with the normal operations and business of Sellers or their Affiliates, (b) such access shall occur in such a manner as Sellers reasonably determine to be appropriate to protect the confidentiality of the transactions contemplated by this Agreement, and (c) nothing herein shall require Sellers to provide access, or to disclose any information, to Buyer if such access or disclosure (i) would waive any legal privilege (except with respect to any title opinions covering any Asset), (ii) would be in violation of applicable Laws or regulations of any Governmental Authority, (y) would be in violation of any confidentiality obligations to a third party or (iii) would constitute access to the Excluded Records. Further, Buyer shall have no right (A) to access any of the Assets or (B) to perform or conduct any environmental sampling or other invasive environmental investigation on or about or with respect to any of the Assets. Buyer shall keep confidential all information made available to Buyer by or on behalf of Sellers or their Representatives until the Closing.

#### 3.2 Title Defects and Purchase Price Adjustment.

(a "**Defect Notice**. Buyer shall give Sellers written notice of alleged Title Defects (a "**Defect Allegation**") promptly upon Buyer's discovery thereof, and in any event, prior to the Title Claims Date. To be effective, such Defect Allegation shall be in writing, delivered to Sellers before the Title Claims Date and shall include (i) a reasonably detailed description of the alleged Title Defect, including Buyer's detailed findings and reasons for concluding that such alleged Title Defect exists, (ii) the

Allocated Value of the Tract affected by the Title Defect, (iii) the amount by which Buyer reasonably believes that the Allocated Value of the affected Tract is reduced by the alleged Title Defect and the computations and information upon which Buyer's belief is based, and (iv) all supporting information and documents in Buyer's possession or reasonable control relating to such asserted Title Defect necessary for Sellers to verify such Title Defect and the value thereof; provided, however, that an alleged failure to comply with subsections (i) through (iv) above shall not cause any such Defect Allegation to be invalid or any Title Defect to be waived so long as the Defect Allegation is timely delivered and it provides reasonably sufficient notice and supporting documents to Sellers of the existence of and reasonable details regarding the nature of the alleged Title Defect. Except for any claim under the Special Warranty, Buyer shall be deemed for any and all purposes to have waived, and Sellers shall have no liability for, any Title Defects and other defects of title of which Sellers have not received a Defect Allegation from Buyer on or before the Title Claims Date meeting the requirements of this Section 3.2(a). Notwithstanding the foregoing, Buyer shall be entitled to modify or amend any Defect Allegation or other communication until the Title Claims Date, and Buyer shall not be deemed to have waived any Title Defects that are ultimately asserted in a Defect Allegation from Buyer on or before the Title Claims Date meeting the requirements of this Section 3.2(a). Upon Sellers' receipt on or prior to the Title Claims Date of any Defect Allegation, Sellers shall review such Defect Allegation in good faith and determine whether Sellers believe such Defect Allegation meets the requirements of this Section 3.2(a). If Sellers determine that any Defect Allegation fails to meet the requirements of this Section 3.2(a), Sellers shall provide written notice to Buyer of such determination (and reasonable detail of the basis for such determination) (a "Deficiency Notice") as soon as reasonably practicable following Sellers' receipt of such Defect Allegation. Notwithstanding anything to the contrary, with respect to each Defect Allegation received by Sellers on or prior to the Title Claim Date, Sellers may not dispute any alleged Title Defect set forth in any such Defect Allegation on the basis that it did not meet the requirements of this Section 3.2(a) unless Sellers provide prompt written notice to Buyer (no later than five (5) Business Days following the Title Claim Date) of the basis upon which Sellers (acting in good faith) have determined that the Defect Allegation failed to meet the requirements of this Section 3.2(a).

Title Credit Notice. Sellers or Buyer, as the case may be, as a "Disclosing Credit Party," may (but are not obligated to), give the other Party written notice of alleged Title Credits (a "Credit Notice") promptly upon the Disclosing Credit Party's discovery thereof, and in any event, prior to the Title Claims Date. To be effective, such Credit Notice provided by a Disclosing Credit Party shall be in writing, delivered to the other Party prior to the Title Claims Date, and shall include (i) a reasonably detailed description of the alleged Title Credit, including the Disclosing Credit Party's detailed findings and reasons for concluding that such alleged Title Credit exists, (ii) the Allocated Value of the Tract or affected by the Title Credit, (iii) the amount by which the Disclosing Credit Party reasonably believes that the Allocated Value of the affected Tract is increased by the alleged Title Credit and the computations and information upon which the Disclosing Credit Party's belief is based, and (iv) all supporting information and documents in the Disclosing Credit Party's possession or reasonable control relating to such asserted Title Credit necessary for the other Party to verify such Title Credit and the value thereof; provided, however, that an alleged failure to comply with subsections (i) through (iv) above shall not cause any such Credit Notice to be invalid or any Title Credit to be waived so long as the Credit Notice is timely delivered and it provides reasonably sufficient notice and supporting documents to Sellers of the existence of and reasonable details regarding the nature of the alleged Title Credit. Sellers shall be deemed for any and all purposes to have waived, and Buyer shall have no liability for, all Title Credits (A) of which Buyer has not received a Credit Notice from Sellers, or (B) for which Buyer has failed to provide a Credit Notice to Seller, on or before the Title Claims Date meeting the requirements of this Section 3.2(b). Prior to the Title Claims Date, the Party receiving a Credit Notice shall notify the Disclosing Credit Party if any written alleged Title Credits received from the Disclosing Credit Party on or before the Title Claims Date do not comply with the provisions of this Section 3.2(b). Upon a Party's receipt on or prior to the Title Claims Date of any Credit Notice, such Party shall review such Credit Notice in good faith and determine whether it believes such Credit Notice meets the requirements of this Section 3.2(b). If a Party determines that any Credit Notice

fails to meet the requirements of this <u>Section 3.2(b)</u>, such Party shall provide written notice to the Disclosing Credit Party of such determination (and reasonable detail of the basis for such determination) (a "Credit Deficiency Notice") as soon as reasonably practicable following such Party's receipt of the Credit Notice. Notwithstanding anything to the contrary, with respect to each Credit Notice received by Buyer on or prior to the Title Claim Date, Buyer may not dispute any alleged Title Credit set forth in any such Credit Notice on the basis that it did not meet the requirements of this <u>Section 3.2(b)</u> unless it provides prompt written notice to Sellers (no later than five (5) Business Days following the Title Claim Date) of the basis upon which Buyer (acting in good faith) has determined that the Credit Notice failed to meet the requirements of this Section 3.2(b).

- (c) <u>Title Defect Amounts.</u> The value of an uncured or unwaived Title Defect (each, a "**Title Defect Amount**"), and any adjustments to the Purchase Price for the same, shall be calculated as follows, without duplication:
  - (i) if Buyer and Sellers agree in writing on the Title Defect Amount, that amount shall be the Title Defect Amount:
  - (ii) if a Title Defect is an Encumbrance upon an Asset which is liquidated in amount, then the adjustment shall be the sum necessary to be paid to the obligee to remove the Title Defect from the affected Asset;
  - (iii) if a Title Defect is in respect of a discrepancy between (A) Sellers' Actual NRA for a Tract and (B) the aggregate Net Royalty Acres set forth on Exhibit A-1 for such Tract (the "Stated NRA" for such Tract), then the Title Defect Amount shall be the product of (1) the Allocated Value for such Tract, multiplied by (2) a fraction, the numerator of which is the absolute value of difference between the Stated NRA and the Actual NRA, and the denominator of which is the Stated NRA;
  - (iv) if the Title Defect represents an obligation or Encumbrance upon or other defect in title to the affected Tract or Well of a type not described in Section 3.2(c)(i) through Section 3.2(c)(iii), above, then the Title Defect Amount shall be determined by taking into account the Allocated Value of the affected Tract, the legal effect of the Title Defect, the potential economic effect of the Title Defect over the life of the affected Tract or Well, the values placed upon the Title Defect by Buyer and Sellers and such other reasonable factors as are necessary to make a proper evaluation; and
  - (v) in no case shall the value of all Title Defects asserted with regard to the Assets included in a Tract exceed the Allocated Value of the affected Tract. The Title Defect Amount with respect to a Title Defect shall be determined without any duplication of any costs included in any other Title Defect Amount hereunder, or for which Buyer otherwise receives credit in the calculation of the Purchase Price.

#### (d) <u>Certain Limitations on Adjustments.</u>

(i) Buyer shall have no remedy for any individual Title Defect, and Sellers shall have no remedy for any individual Title Credit, unless the Title Defect Amount for such individual Title Defect, or Title Credit Amount for such individual Title Credit, exceeds Five Thousand Dollars (\$5,000.00) (the "**Title Threshold**"), and no Title Defect or Title Credit shall be taken into consideration for the purpose of calculating any Purchase Price reduction or increase under this <u>Section 3.2</u> unless the Title Defect Amount for such

Title Defect, or Title Credit Amount for such Title Credit, exceeds the Title Threshold (which shall be treated as a threshold and not a deductible).

- Buyer shall have no remedy for any Title Defects unless and until the sum of (A) the aggregate Title Defect Amounts for Title Defects with Title Defect Amounts exceeding the Title Threshold minus (B) the aggregate Title Credit Amounts for Title Credits with Title Credit Amounts exceeding the Title Threshold, exceeds Fifty Thousand Dollars (\$50,000.00) ("Title Deductible"). Sellers shall not be entitled to an upward Purchase Price adjustment for any Title Credits unless and until the sum of (1) the aggregate Title Credit Amounts for Title Credits with Title Credit Amounts exceeding the Title Threshold, minus (2) the aggregate Title Defect Amounts for Title Defects with Title Defect Amounts exceeding the Title Threshold, exceeds the Title Deductible. If, and only if, the Purchase Price reduction or increase which would result from all Title Defect Amounts or Title Credit Amounts, as the case may be, that exceed the Title Threshold exceeds the Title Deductible, then the Purchase Price shall be reduced by the aggregate of all Title Defect Amounts or increased by the aggregate of all Title Credit Amounts. For the avoidance of doubt, Buyer or Sellers, as the case may be, shall be entitled to adjust the Purchase Price under this Article 3 from the first dollar of any Title Defects or Title Credits, as the case may be, from and after such time as the aggregate amount of all Title Defects or Title Credits (as the case may be) under this Article 3 equals or exceeds the Title Deductible.
- (iii) The Title Threshold and Title Deductible shall not apply to Title Defects arising by, through or under any Seller or any of its Affiliates or any breach of the Special Warranty.
- (e) <u>Title Credit Amounts</u>. The value of a Title Credit (a "**Title Credit Amount**") shall be calculated as follows:
  - (i) if Buyer and Sellers agree in writing on the Title Credit Amount, that amount shall be the Title Credit Amount;
  - (ii) if a Title Credit results from the Actual NRA for the Tract that contains the affected Asset exceeding the Stated NRA for such Tract, then the Title Credit Amount shall be the product of (A) the Allocated Value for such Tract, <u>multiplied</u> by (B) a fraction, the numerator of which is the absolute value of difference between the Stated NRA and the Actual NRA, and the denominator of which is the Stated NRA;
  - (iii) if the Title Credit is of a type not described in Section 3.2(e)(i) through Section 3.2(e)(ii), then the Title Credit Amount will be determined by taking into account the Allocated Value of the affected Tract, the legal effect of the Title Credit, the potential economic effect of the Title Credit over the life of the affected Tract, the values placed upon the Title Credit by Buyer and Sellers and such other reasonable factors as are necessary to make a proper evaluation.
  - (iv) The Title Credit Amount with respect to a Title Credit shall be determined without any duplication of any credit included in any other Title Credit Amount hereunder, or for which any Seller otherwise receives credit in the calculation of the Purchase Price.
- (f) <u>Cure</u>. Sellers may, but shall not be obligated to, cure Title Defects prior to the Closing or after the Closing subject to the provisions of this <u>Section 3.2(f)</u> and <u>Section 3.2(g)</u>. In the event

Sellers elect, in their sole discretion, to cure one or more Title Defects after Closing, Sellers shall deliver a written notice to Buyer prior to Closing identifying the Title Defects that Sellers have elected to cure (the "Cure Notice"). If Sellers timely deliver a Cure Notice, the Tract or Tracts affected by the Title Defects described in the Cure Notice shall be retained by Sellers at Closing, and subject to Section 3.2(d), the Purchase Price shall be reduced by the Allocated Value of the Tract or Tracts so retained by Sellers. Sellers shall be permitted, but shall not be obligated, to attempt to cure such Title Defects for a period of ninety (90) days after Closing (such period, the "Cure Period"), during which time, Buyer shall cooperate with Sellers in connection with any such attempts by Sellers to so cure. In the event that Sellers cure any such Title Defect during the Cure Period, subject to Section 3.2(g), (i) Sellers shall promptly notify Buyer that the Title Defect has been cured, (ii) Buyer shall pay to Sellers the amount by which the Purchase Price was reduced at Closing with respect to such excluded Assets (or portion thereof), on the first business day following the expiration of the Cure Period (the "Delayed Closing Date"), in accordance with the terms of Agreement, and (iii) Sellers shall convey the Assets so excluded at Closing on the Delayed Closing Date pursuant to an instrument effective as of the Effective Time substantially in the same form as the Conveyance.

#### (g) <u>Title Adjustments; Dispute Resolution</u>.

- (i) Sellers and Buyer shall attempt to agree in writing on all alleged Title Defects and alleged Title Credits, the appropriate adjustment amounts for such alleged Title Defects and alleged Title Credits and the sufficiency of any cure of any such alleged Title Defects (any disagreement regarding the foregoing, a "**Title Dispute**") prior to Closing.
- (ii) If Sellers and Buyer agree in writing prior to Closing on the existence of any Title Defect or Title Credit and the Title Defect Amount or Title Credit Amount in respect thereof, then subject to Section 3.2(d), the Purchase Price shall be reduced at Closing by the aggregate sum of all agreed Title Defect Amounts and the Purchase Price shall be increased at Closing by the aggregate sum of all agreed Title Credit Amounts.
- (iii) If Sellers and Buyer are unable to agree in writing prior to Closing on any Title Dispute, and if Sellers do not elect to cure under Section 3.2(f), then (A) the Assets included in the Tract or Tracts affected by the unresolved Title Defect shall nevertheless be conveyed to Buyer at the Closing, (B) subject to Section 3.2(d), the Purchase Price shall be reduced by the Title Defect Amount for such Title Defect, as alleged by Buyer, and (C) the Title Dispute shall be exclusively and finally resolved under Section 3.2(g)(iv). Promptly following the final determination of any disputed Title Defect or Title Defect Amount by the Parties or by the Title Consultant in accordance with the procedures set forth in Section 3.2(g)(iv), the Parties shall remit any amounts so determined to be owed to either Party with respect to such disputed matter.
- (iv) If the Parties are unable to agree in writing as to any matters that constitute Title Disputes within thirty (30) days following the expiration of the Cure Period, then the Title Disputes shall be exclusively and finally resolved by arbitration to be conducted in Dallas, Texas, by an independent title attorney who has at least fifteen (15) years' experience in oil and gas title involving properties in the state in which the Assets affected by the applicable Title Dispute are located (the "Title Consultant"). If Title Disputes exist with respect to Assets located in multiple states, then separate Proceedings shall be conducted with respect to the Title Disputes relating to Assets in each state. The Title Consultant shall be selected by: (A) mutual written agreement of Buyer and Sellers; or (B) absent such agreement, by the Dallas office of the American Arbitration Association (the "AAA"); provided, however, that such Title Consultant must (1) have not been employed by any Party (or its Affiliate) in the past

ten (10) years, and (2) have no ethical conflict in serving as the Title Consultant. If there is more than one disputed title matter, all such disputes shall be consolidated into the same arbitration process pursuant to this Section 3.2(g). Sellers and Buyer shall each present to the Title Consultant, with a simultaneous copy to the other Party, a single written statement of its position on the defect or benefit in question, together with a copy of this Agreement and any supporting material that such Party desires to furnish, not later than the tenth (10th) Business Day after appointment of the Title Consultant. The Title Consultant, once appointed, shall have no ex parte communications with any of the Parties concerning the determination required under this Section 3.2(g). All communications between any Party or its Affiliates and the Title Consultant shall be conducted in writing, with copies sent simultaneously to the other Parties in the same manner, or at a meeting or conference call to which each of the Parties and their respective Representatives have been invited and of which such Parties have been provided at least five (5) days' notice. In no event shall the Title Consultant select an amount (a) for any upward adjustment to the Purchase Price that is greater than Sellers' proposed amount or less than Buyer's proposed amount or (b) for any downward adjustment to the Purchase Price that is greater than Buyer's proposed amount or less than Sellers' proposed amount. Within thirty (30) days after receipt of such materials and after receipt of any additional information required by the Title Consultant, the Title Consultant shall make its determination, which shall be in accordance with the terms of this Agreement and shall be final and binding upon the Parties, without right of appeal, absent Fraud or manifest error. The Title Consultant may not award damages, interest or penalties to either Buyer or Sellers with respect to any matter. Sellers and Buyer shall each bear their own respective legal fees and other costs of presenting its case. Sellers shall bear fifty percent (50%) and Buyer shall bear fifty percent (50%) of the costs and expenses of the Title Consultant. If the Title Consultant determines that no Title Defect or Title Credit, as applicable, exists on an Asset, then after the Title Consultant delivers written notice to Sellers and Buyer of his or her award with respect to such Title Defect or Title Credit, as applicable, the matter shall be deemed resolved and no adjustment shall be made on account of such asserted Title Defect or Title Credit, as applicable. If the Title Consultant determines that a Title Defect or Title Credit, as applicable, exists on an Asset, then the Title Consultant will deliver written notice to Sellers and Buyer of his or her award with respect to such Title Defect or Title Credit, as applicable, and (i) if the Final Settlement Statement has not been finally resolved, such awards shall be reflected in the Final Settlement Statement, or (ii) if the Final Settlement Statement has been finally resolved, Sellers shall deliver to Buyer, or Buyer shall deliver to Sellers, as applicable, within five (5) Business Days after receipt of the Title Consultant's award, the amount owed to Buyer or Sellers, as applicable, after the determination of such award.

## Article 4 SELLERS' REPRESENTATIONS AND WARRANTIES

As of the Execution Date and the Closing Date, each Seller, on a joint and several basis, represents and warrants to Buyer that, except as set forth on the Disclosure Schedules:

- 4.1 **Organization.** Such Seller is duly formed, validly existing and in good standing under the laws of the jurisdiction of its formation, with full limited liability company power and authority to enter into this Agreement and perform its obligations under this Agreement.
- 4.2 **Qualification.** Such Seller is duly qualified to do business and is in good standing in each jurisdiction in which the nature of its business as now conducted makes such qualification necessary.

- 4.3 **Power; No Conflict.** Such Seller has all requisite power and authority to carry on its business as presently conducted, to enter into this Agreement, each Ancillary Agreement to which it is a party and to consummate the transactions contemplated by this Agreement and each Ancillary Agreement and perform its obligations under this Agreement and each Ancillary Agreement. The execution and the delivery of this Agreement does not, the consummation of the transactions contemplated by this Agreement by Sellers does not, and the fulfillment of and compliance with the terms and conditions of this Agreement will not: (a) conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, (b) give rise to a right of termination, modification, revocation, cancellation or acceleration of any obligation or to the loss of a benefit under or (c) result in the creation of any claim upon any of the Assets or any Party under, in each case, any provision of (i) the organizational documents of such Seller, (ii) any applicable Law or (iii) any contract or instrument (including any judgment, decree or order) to which such Seller is directly a party or by which it is bound or that otherwise relates to any of the Assets, except any matters described in clauses (ii) and (iii) which would not reasonably be expected to have a Material Adverse Effect.
- 4.4 **Authorization and Enforceability.** Such Seller's execution, delivery and performance of or under this Agreement and the Ancillary Agreements to which it is a party have been duly and validly authorized and approved by all necessary limited liability company actions. This Agreement constitutes, and each Ancillary Agreement to which such Seller is a party constitutes or will constitute at Closing, such Seller's legal, valid and binding obligation, enforceable in accordance with its terms, subject, however, to the effects of bankruptcy, insolvency, reorganization, moratorium, fraudulent transfer and other applicable Laws for the protection of creditors, as well as to general principles of equity, regardless whether such enforceability is considered in a Proceeding in equity or at Law.
- 4.5 **Liability for Brokers' Fees.** Neither such Seller nor its Affiliates (or other related Persons) have incurred any liability, contingent or otherwise, for investment bankers', brokers' or finders' fees relating to the transactions contemplated by this Agreement for which Buyer or its Affiliates shall have any responsibility whatsoever.
- 4.6 **Foreign Person.** Such Seller or its regarded owner (if such Seller is a disregarded entity for federal income Tax purposes) is not a "foreign person" within the meaning of Section 1445 of the Internal Revenue Code of 1986, as amended (the "**Code**") and the regulations promulgated thereunder.
- 4.7 **No Bankruptcy.** There are no bankruptcy Proceedings pending, being contemplated by or threatened against such Seller.
- 4.8 **Litigation.** Except as set forth in Section 4.8 of the Disclosure Schedules, (a) there are no Proceedings relating to the Assets, or that challenge or impair such Seller's ability to enter into this Agreement or the Ancillary Agreements, or to consummate the transactions contemplated hereby and thereby that are pending in any court or by or before any Governmental Authority in which such Seller is a party; and (b) no Proceeding relating to this transaction or the Assets, or that challenges or impairs such Seller's ability to enter into this Agreement or the Ancillary Agreements or to consummate the transactions contemplated hereby and thereby, has been threatened against such Seller (or any of its Affiliates). Such Seller is not in default under any order, writ, injunction or decree of any Governmental Authority applicable to them or any of the Assets, and there are no unsatisfied judgments or injunctions issued by a Governmental Authority outstanding against such Seller related to the Assets.
  - 4.9 **Taxes.** Except as set forth on Section 4.9 of the Disclosure Schedules,

13

- (a) all Tax Returns required to be filed by such Seller with respect to Asset Taxes due and payable during such Seller's ownership of the Assets have been timely filed and all Asset Taxes reported on such Tax Returns have been duly paid;
- (b) there are no audits, investigations, litigation or other Proceedings with respect to Asset Taxes pending, or threatened in writing, against such Seller;
- (c) there are no Encumbrances on any of the Assets attributable to Taxes other than statutory liens for Taxes that are not yet due and payable or that are being contested in good faith by appropriate actions; and
- (d) none of the Assets owned by such Seller is subject to any Tax partnership agreement or is otherwise treated, or required to be treated, as held in an arrangement requiring a partnership income Tax Return (other than partnership income Tax Returns required to be filed by such Seller) to be filed under Subchapter K of Chapter I of Subtitle A of the Code.
- 4.10 **Consents; Preferential Rights; Non-Competes.** There are no consents that are required to be obtained in connection with the transfer and conveyance of the Assets to Buyer which expressly provide that the failure to obtain such consent would render the assignment void, or would trigger an obligation of such Seller to pay liquidated damages or cause termination or loss of the affected Asset. There are no rights of first refusal, preferential purchase rights, options or other similar provisions to which the Assets are subject that are triggered in connection with the execution of this Agreement by such Seller. No Asset is subject to any non-compete or area of mutual interest agreements.
- 4.11 **Contracts and Leases.** (a) All Material Contracts are listed in Section 4.11 of the Disclosure Schedules, and full copies of each Material Contract in Seller's or its Affiliates' possession have been made available to Buyer prior to the Execution Date, (b) all Material Contracts and Leases are in full force and effect, and each Material Contract and Lease constitutes the legal, valid and binding obligation of any Seller, on the one hand, and to Seller's Knowledge, the counterparties thereto, on the other hand, and is enforceable in accordance with its terms, (c) no Seller is in default or otherwise in breach with respect to any of such Seller's obligations under any of such Material Contracts or Leases, (d) to Sellers' Knowledge, no other Person is in default or, to Sellers' Knowledge, otherwise in breach with respect to such Person's obligations under such Material Contracts or Leases, and (e) no event has occurred that with notice or lapse of time or both would constitute any default under any such Material Contract or Lease by any Seller or, to Sellers' Knowledge, by any other Person who is a party to such Material Contract or Lease. No Seller has received or given any unresolved written notice of default or termination with respect to any Material Contract or Lease.
- 4.12 **Payment of Royalties; No Suspense Accounts; No Deferred Payments.** To Seller's Knowledge, there are no monies being held in suspense that are otherwise payable to Seller in respect of (a) any royalty payments, lease rentals, delay rentals, shut-in royalties or other payments under any of the Leases, or (b) the overriding royalty payments relating to the Overriding Royalty Interests. Seller owes no bonus, deferred purchase price or other acquisition costs, whether contingent or liquidated, on any of the Assets.
- 4.13 **Environmental Matters.** Except as set forth on Section 4.13 of the Disclosure Schedules, (a) to Seller's Knowledge, such Seller is in compliance with all Environmental Laws to the extent pertaining to the ownership of the Assets, (b) there are no civil, criminal, or administrative actions or notices pending or, to Seller's Knowledge, threatened in writing against such Seller under any Environmental Law, in each case, of which such Seller has received notice, that are related to the Assets or the operations on the Leases, (c) such Seller has not entered into any agreements, consents, orders, decrees or judgments with any

Governmental Authorities based on any prior violations of Environmental Laws, or any material Losses thereunder, that relate to the future use of the Assets and that require any future remediation, (d) such Seller has not received notice from any Governmental Authority that any of the Assets are the subject of any remediation, removal, clean-up, response action, enforcement action or order regarding any actual or alleged presence or release of Hazardous Substances that has not been finally resolved, and (e) there are no pending or, to Seller's Knowledge, threatened claims, demands, written notices of violation or liability or Proceedings under any Environmental Laws that individually or in the aggregate would reasonably be expected to be material to the value of any Asset.

- 4.14 **No Cost-Bearing Interests.** Except as set forth in <u>Section 4.14</u> of the Disclosure Schedules, the Assets do not include any interests which obligate such Seller owning such Assets to bear a share of drilling, operating or any other costs of production or development.
- 4.15 **Condemnation.** There is no actual or, or to Sellers' Knowledge, threatened taking (whether permanent, temporary, whole or partial) of any part of the Assets by reason of condemnation or the threat of condemnation.

# Article 5 BUYER'S REPRESENTATIONS AND WARRANTIES

As of the Execution Date and the Closing Date, Buyer represents and warrants to Sellers that:

- 5.1 **Organization.** Buyer is duly formed, validly existing and in good standing under the laws of the jurisdiction of its formation, with full power and authority to enter into this Agreement and perform its obligations under this Agreement.
- 5.2 **Qualification.** Buyer is duly qualified to do business and is in good standing in each jurisdiction in which the nature of its business as now conducted makes such qualification necessary, except where the failure to be so qualified or in good standing would not materially hinder or impede the consummation by Buyer of the transactions contemplated by this Agreement.
- **Power.** Buyer has all requisite power and authority to carry on its business as presently 5.3 conducted, to enter into this Agreement, each Ancillary Agreement to which it is a party and to consummate the transactions contemplated by this Agreement and the Ancillary Agreements and perform its obligations under this Agreement and the Ancillary Agreements. The execution and delivery of this Agreement does not, and the fulfillment of and compliance with the terms and conditions of this Agreement will not: (a) conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, (b) give rise to a right of termination, modification, revocation, cancellation or acceleration of any obligation or to the loss of a benefit under or (c) result in the creation of any claim upon any of the Assets or any Party under, in each case, any provision of (i) Buyer's governing documents, (ii) any applicable Law or (iii) any contract or instrument (including any judgment, decree, order, statute, rule or regulation applicable to Buyer) to which Buyer or any of its subsidiaries is a party or by which any properties or assets of Buyer or its subsidiaries is bound. No authorization, consent or approval by, or filing with, any Governmental Authority is required for, or in connection with, the authorization, execution, delivery and performance by the Buyer of this Agreement or the consummation of the transactions contemplated by this Agreement and by the Ancillary Agreements; except any matters described in clauses (ii) and (iii) which would not reasonably be expected to have a Material Adverse Effect.
- 5.4 **Authorization and Enforceability.** Buyer's execution, delivery and performance of or under this Agreement and the Ancillary Agreements to which it is a party have been duly and validly authorized and approved by all necessary corporate actions. This Agreement constitutes, and each Ancillary

Agreement to which Buyer is a party constitutes or will constitute at Closing, Buyer's legal, valid and binding obligation, enforceable in accordance with its terms, subject, however, to the effects of bankruptcy, insolvency, reorganization, moratorium and other applicable Laws for the protection of creditors, as well as to general principles of equity, regardless whether such enforceability is considered in a Proceeding in equity or at Law.

- 5.5 **Liability for Brokers' Fees.** Buyer has not incurred and will not incur any liability, contingent or otherwise, for investment bankers', brokers' or finders' fees relating to the transactions contemplated by this Agreement for which any Seller shall have any responsibility whatsoever.
- 5.6 **Litigation.** There are no Proceedings pending, or to Buyer's Knowledge, threatened against Buyer that would challenge or impair the ability of Buyer to enter into this Agreement or the Ancillary Agreements or that would have a Material Adverse Effect upon the ability of Buyer to consummate the transactions contemplated hereby.
- 5.7 **Knowledgeable Investor.** Buyer is an experienced and knowledgeable investor in the oil and gas business and is sophisticated in the evaluation, purchase, ownership, and operation of oil and gas properties and related facilities. Buyer is not acquiring the Assets in connection with a distribution or resale thereof in violation of federal or state securities Laws and the rules and regulations thereunder.
- 5.8 **No Bankruptcy.** There are no bankruptcy Proceedings pending, being contemplated by or threatened against Buyer.

# Article 6 COVENANTS

- 6.1 **Pre-Closing Covenants.** From and after the Execution Date and until the earlier to occur of the Closing and the termination of this Agreement in accordance with its terms, and subject to the terms of the Contracts and the Leases, each Seller shall administer its Assets in a good and workmanlike manner consistent with its past practices (including, for the avoidance of doubt, the granting of leases and lease renewals in the ordinary course of business) and applicable Law, and shall carry on its business with respect to its interest in the Assets in substantially the same manner as before the Execution Date. Notwithstanding the foregoing, from and after the Execution Date and until Closing, no Seller shall, without Buyer's prior written consent, (a) transfer, sell, mortgage, pledge, abandon or dispose of any of the Assets other than the granting of leases and lease renewals in the ordinary course of business; (b) purchase or otherwise acquire any interests that are similar in type to any of the Interests; (c) amend, modify or terminate any Material Contract or enter into any Contract that would have been a Material Contract if it had been in effect on the Execution Date; or (d) agree, whether in writing or otherwise, to do any of the foregoing; provided that nothing in this Section 6.1 shall be construed to prevent or limit any Seller from complying with applicable Laws or the terms of any existing Material Contract or Lease.
- Reasonable Efforts. Each Party agrees that it will not voluntarily undertake any course of action inconsistent with the provisions or intent of this Agreement and will use its Reasonable Efforts to take, or cause to be taken, all action and to do, or cause to be done, all things reasonably necessary, proper, or advisable under applicable Laws to consummate the transactions contemplated by this Agreement, including (a) cooperating in determining whether any consents, approvals, orders, authorizations, waivers, declarations, filings, or registrations of or with any Governmental Authority or third party are required in connection with the consummation of the transactions contemplated by this Agreement; (b) obtaining any consents, approvals, and registrations; (c) causing to be lifted or rescinded any injunction or restraining order or other order adversely affecting the ability of the Parties to consummate the transactions contemplated by this Agreement; (d) defending, and cooperation in defending, all Proceedings challenging

16

this Agreement or the consummation of the transactions contemplated by this Agreement; and (e) executing any additional instruments necessary to consummate the transactions contemplated by this Agreement.

- Amendment of Disclosure Schedules. From time to time up to the earlier of the Closing 6.3 Date or termination of this Agreement in accordance with the terms hereof, with respect to the representations and warranties of a Party made in this Agreement, the Parties shall have the right, by written notice to other Party, to promptly supplement or amend the Disclosure Schedules to reflect any event or circumstance that first came into existence after the Execution Date and would have been required to be set forth or described on one or more existing or new Disclosure Schedules if such event or circumstance was in existence on the Execution Date of this Agreement (for the avoidance of doubt, no revision or supplement of the Disclosure Schedules may be effected on account of obtaining knowledge after the Execution Date of an event or circumstance that first came into existence before the Execution Date); provided that any such supplement, amendment or addition shall not be deemed to have (a) cured any inaccuracy in or breach of any representation, warranty or covenant for purposes of determining whether or not the conditions set forth in Sections 8.1 or 8.2 (as applicable) have been met or (b) except with respect to a Subject Schedule Supplement, cured any inaccuracy in or breach of any representation, warranty or covenant for purposes of determining Sellers' indemnification obligations pursuant to Section 10.2 (and, for the avoidance of doubt, such liabilities will count against the Deductible). If any matter disclosed pursuant to any schedule supplement by Sellers in accordance with this Section 6.3 gives rise to the right of Buyer to terminate this Agreement (such a "Subject Schedule Supplement") and Buyer nevertheless elects to proceed with the Closing, then such matters giving rise to the right of Buyer to terminate that are disclosed pursuant to any such Subject Schedule Supplement shall be deemed waived by Buyer and Buyer shall not be entitled to make a claim pursuant to Section 10.2(a) with respect to such matters.
- Limitations on Representations and Warranties. ALL ORAL INFORMATION OR 6.4 MATERIALS, DOCUMENTS AND OTHER INFORMATION MADE AVAILABLE TO BUYER BY SELLERS OR ANY OF THEIR REPRESENTATIVES AT ANY TIME IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY HAVE BEEN MADE AVAILABLE AS AN ACCOMMODATION AND HAVE BEEN PROVIDED ON AN "AS IS" BASIS. EXCEPT FOR THE EXPRESS AND SPECIFIC REPRESENTATIONS AND WARRANTIES OF EACH SELLER IN ARTICLE 4, THE SELLER OFFICER CERTIFICATES, THE ANCILLARY AGREEMENTS OF EACH SELLER DELIVERED AT THE CLOSING AND THE SPECIAL WARRANTY, AS APPLICABLE, BUYER ACKNOWLEDGES THAT SELLERS HAVE NOT MADE, AND SELLERS HEREBY EXPRESSLY DISCLAIM AND NEGATE ANY OTHER REPRESENTATION OR WARRANTY (EXPRESS, IMPLIED, AT COMMON LAW, BY STATUTE OR OTHERWISE), INCLUDING THE WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE, AND FURTHER INCLUDING, WITHOUT LIMITATION, ANY REPRESENTATION AND WARRANTIES RELATING TO (A) PRODUCTION RATES, RECOMPLETION OPPORTUNITIES, GEOLOGY, ENGINEERING, DECLINE RATES OR THE QUALITY, QUANTITY OR VOLUME OF THE RESERVES OF OIL AND GAS OR OTHER MINERALS, IF ANY, ATTRIBUTABLE TO THE ASSETS, (B) THE ACCURACY, COMPLETENESS OR MATERIALITY OF ANY INFORMATION, DATA OR OTHER MATERIALS (WRITTEN OR ORAL) NOW, HERETOFORE OR HEREAFTER FURNISHED TO BUYER BY OR ON BEHALF OF SELLER, AND (C) THE ENVIRONMENTAL CONDITION OF THE ASSETS. SELLERS AND BUYER AGREE THAT, TO THE EXTENT REQUIRED BY APPLICABLE LAW TO BE EFFECTIVE, THE DISCLAIMERS OF CERTAIN WARRANTIES CONTAINED IN THIS SECTION ARE "CONSPICUOUS" DISCLAIMERS FOR THE PURPOSES OF ANY APPLICABLE LAW, RULE OR ORDER.
- 6.5 **Post-Closing Obligations.** If Closing occurs, Sellers and Buyer have the following post-Closing obligations:

17

- (a) Property Records. Within thirty (30) days after Closing, Sellers shall deliver (at Sellers' cost and expense) the Records (which may be in the form of original records or electronic copies) to Buyer at the address of Buyer. With respect to any original Records delivered to Buyer, (i) Sellers shall be entitled to retain copies of such Records, and (ii) Buyer shall preserve and retain any such original Records for at least seven (7) years beyond the Closing Date. During such seven (7) year period, Seller shall be entitled to obtain access to such Records, at reasonable business hours and upon prior notice to Buyer, so that Sellers may make copies of such original Records, at Sellers' expense (and Sellers shall maintain the strict confidentiality of any Records of which it retains any copies), including as may be reasonable or necessary in connection with any Proceeding or threatened Proceeding against Seller.
- (b) Recording and Filing. Buyer, within thirty (30) days after Closing, shall use Reasonable Efforts to: (i) record the Conveyance and all other instruments that must be recorded to effectuate the transfer of the Assets; and (ii) file for approval with the applicable federal, state or local agencies the Conveyance (to the extent required by applicable Law) and all other federal, state or local transfer documents required to effectuate transfer of the Assets. Buyer shall provide Sellers a recorded copy of the Conveyance and other recorded instruments, and approved copies of the Conveyance and other federal, state or local transfer documents, promptly after they are available.
- (c) Special Warranty. Subject to Section 10.3, the Conveyance delivered at Closing shall provide that each Seller, on a joint and several basis, will warrant and defend Defensible Title to the Assets against whomsoever lawfully claims the same or any part thereof solely to the extent a claim is brought by, through or under such Seller or any of its Affiliates but not otherwise (the "Special Warranty"). To the extent Seller may legally assign such rights and grant such subrogation, Buyer shall have all rights of legal subrogation to third party claims or Seller claims asserted or held by that third party or by Seller arising from or related to the Special Warranty.
- 6.6 **Exclusivity.** From and after the Execution Date and until the earlier to occur of the Closing and the termination of this Agreement in accordance with its terms: (a) each Seller shall not, directly or indirectly through any of its Affiliates or any of its or their respective Representatives, (i) encourage, solicit, initiate, facilitate or continue inquiries regarding bids, offers, inquiries or proposals from any Person (other than Buyer and its designees, agents and all their respective Affiliates) with respect to the sale, transfer or disposition of all or any portion of the Assets ("**Acquisition Proposals**"), (ii) enter into discussions or negotiations with, or provide any information to, any Person concerning a possible Acquisition Proposal or (iii) enter into any agreements or other instruments (whether or not binding) regarding an Acquisition Proposal; (b) each Seller shall immediately cease and cause to be terminated, and shall cause its Affiliates and all of its and their Representatives to immediately cease and cause to be terminated, all existing discussions or negotiations with any Persons with respect to, or that could lead to, an Acquisition Proposal; and (c) each Seller shall promptly provide Buyer with notice of any unsolicited Acquisition Proposals received by such Seller.

# Article 7 TAX MATTERS

7.1 **Transfer Taxes.** All required documentary, filing and recording fees and expenses incurred in connection with the filing and recording of the assignments, conveyances or other instruments required to convey title to the Assets to Buyer shall be borne by Buyer. If any Transfer Taxes are imposed on the transfer of the Assets to Buyer, Buyer shall bear and pay any such Transfer Taxes. Sellers and Buyer shall reasonably cooperate in good faith to minimize, to the extent permissible under applicable Law, the amount of any such Transfer Taxes.

#### 7.2 Asset Taxes.

- (a) The applicable Seller shall be allocated all Asset Taxes attributable to (i) any Tax period ending prior to the Effective Time and (ii) the portion of any Straddle Period ending immediately prior to the Effective Time. Buyer shall be allocated and bear all Asset Taxes attributable to (A) any Tax period beginning after the Effective Time and (B) the portion of any Straddle Period beginning at the Effective Time. Buyer shall be responsible for payment to the applicable Governmental Authorities of all Asset Taxes that become due and payable, and the filing of all Tax Returns for Asset Taxes that become due, on or after the Closing Date.
- For purposes of determining the allocations described in Section 7.2(a), (i) Asset Taxes that are based upon income, sales, revenue or similar items (other than Asset Taxes described in clause (ii) or (iii) below) shall be allocated to the period in which the transaction giving rise to such Asset Taxes occurred, (ii) Asset Taxes that are ad valorem, property or similar Asset Taxes imposed on a periodic basis pertaining to a Straddle Period shall be allocated between the portion of such Straddle Period ending immediately prior to the Effective Time and the portion of such Straddle Period beginning at the Effective Time based on the number of days in the applicable Straddle Period that occur before the date on which the Effective Time occurs, on the one hand, and the number of days in such Straddle Period that occur on or after the date on which the Effective Time occurs, on the other hand, and (iii) Asset Taxes that are attributable to the severance or production of Hydrocarbons (other than Asset Taxes described in clause (ii) above) shall be allocated to the period in which the severance or production giving rise to such Asset Taxes occurred. For purposes of (A) the preceding sentence, any exemption, deduction, credit or other item that is calculated on an annual basis shall be allocated pro rata per day between the portion of the Straddle Period ending immediately prior to the Effective Time and the portion of the Straddle Period beginning at the Effective Time; and (B) clause (ii) of the preceding sentence, the period for such Asset Taxes shall begin on the date on which ownership of the applicable Assets gives rise to liability for the particular Asset Tax and shall end on the day before the next such date.
- (c) As provided in Article 2, the Purchase Price shall be adjusted for Asset Taxes allocable to each Party under Section 7.2(a) and Section 7.2(b). To the extent the actual amount of an Asset Tax is not known at the Final Settlement Date, Buyer and Sellers shall utilize the most recent information available in estimating the amount of such Asset Tax for purposes of adjusting the Purchase Price under this Section 7.2(c). To the extent the actual amount of an Asset Tax (or the amount thereof paid or economically borne by a Party) is ultimately determined to be different than the amount (if any) that was taken into account for purposes of adjusting the Purchase Price under this Section 7.2(c), Sellers shall pay Buyer, or Buyer shall pay Sellers, as the case may be, within a commercially reasonable amount of time following written notice of such determination, any amount necessary to cause each of Sellers and Buyer to bear its appropriate share of such Asset Tax as determined under this Section 7.2.
- 7.3 **Tax Allocation.** The Parties agree that the Purchase Price any other amounts treated as consideration for U.S. federal income Tax purposes shall be allocated among the Assets for U.S. federal and applicable state income Tax purposes in accordance with all applicable Treasury Regulations promulgated under Section 1060 of the Code, and to the extent allowed by applicable Law, in a manner consistent with the Allocated Values. The initial draft of such allocation shall be prepared by Buyer and shall be provided to Seller no later than thirty (30) days after the Final Settlement Date. The Parties shall then cooperate to prepare a final allocation, which also shall be materially consistent with the Allocated Values to the extent allowed by applicable Law. In the event that Sellers and Buyer are not able to resolve all of the items in the allocation, all such unresolved items shall be determined by the Accounting Consultant in accordance with the procedures of Section 2.4(c), mutatis mutandis. The final allocation shall be updated to reflect any adjustments to the Purchase Price and any other amounts treated as consideration for U.S. federal income Tax purposes. The final allocation shall be used by Sellers and Buyer as the basis for

reporting asset values and other items, including preparing Internal Revenue Service Form 8594 (Asset Acquisition Statement under Section 1060), if required. Each Party agrees not to take any position inconsistent with such final allocation unless (a) required by Law or as required by final adjustment or settlement with any Governmental Authority or agency, in which case written notice of such change shall be provided to the other Party, or (b) with the consent of the other Party.

- 7.4 **Refunds of Taxes.** Sellers shall be entitled to any and all refunds of Taxes allocated to Sellers under Section 7.2, and Buyer shall be entitled to any and all refunds of Taxes allocated to Buyer under Section 7.2. If a Party receives a refund of Taxes to which the other Party is entitled under this Section 7.4, the first Party shall promptly pay such amount to the other Party, net of any reasonable costs or expenses incurred by the first Party in procuring such refund.
- 7.5 **Cooperation.** Sellers and Buyer each agree to furnish or cause to be furnished to the other, upon request, as promptly as practicable, such information and assistance relating to the Assets, including access to books and records, as is reasonably necessary for the filing of all Tax Returns by Buyer or Sellers, the making of any election relating to Taxes, the preparation for any audit by any taxing authority and the prosecution or defense of any claim, suit or Proceeding relating to any Tax attributable to the Assets.

## Article 8 CONDITIONS PRECEDENT AND TERMINATION

- 8.1 **Conditions to Obligations of Sellers.** The obligations of Sellers to consummate the transactions provided for in this Agreement are subject, at the option of Sellers, to the fulfillment on or prior to the Closing of each of the following conditions:
- (a) (i) The Buyer Fundamental Representations shall be true and correct as of the Execution Date and as of the Closing Date as though made on and as of such date (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date) and (ii) the representations and warranties of Buyer contained in this Agreement that are not Buyer Fundamental Representations shall be true and correct as of the Execution Date and as of the Closing Date as though made on and as of such date (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date), except where all such breaches of the representations and warranties described in clause (ii) taken collectively would not have a Buyer Material Adverse Effect;
- (b) Buyer shall have performed or complied in all material respects with all obligations, covenants and agreements contained in this Agreement as to which performance or compliance by Buyer is required prior to the Closing;
- (c) Buyer shall deliver to the Sellers a certificate signed by an officer of Buyer in his or her capacity as an officer of Buyer (and not individually), dated as of the Closing Date, certifying that the conditions specified in <u>Section 8.1(a)</u> and <u>Section 8.1(b)</u> have been fulfilled (the "**Buyer Officer Certificate**");
- (d) No material suit, action, litigation or other Proceeding instituted by any third party shall be pending before any Governmental Authority seeking to restrain, prohibit, enjoin or declare illegal, or seeking substantial damages in connection with, the transactions contemplated by this Agreement, and no statute, rule, regulation, executive order, decree, temporary restraining order, preliminary or permanent injunction, judgment or other order shall have been enacted, entered, promulgated, enforced or issued by any Governmental Authority preventing the consummation of the transactions contemplated by this Agreement;

- (e) Buyer shall be ready, willing and able to deliver to Sellers, and shall have delivered to Sellers, the closing deliverables to be delivered by Buyer in accordance with Section 9.2; and
- (f) The closing of the transactions contemplated in the Purchase and Sale Agreement dated as of December 6, 2021 between Buyer and Palmetto Investment Partners II, LLC (the "Concurrent PSA") shall have occurred concurrently with or prior to the Closing.
- 8.2 **Conditions to Obligations of Buyer.** The obligations of Buyer to consummate the transactions provided for in this Agreement are subject, at the option of Buyer, to the fulfillment on or prior to the Closing of each of the following conditions:
- (a) (i) The Seller Fundamental Representations shall be true and correct on and as of the Execution Date and the Closing Date as though made on and as of such dates (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date) and (ii) the representations and warranties of each Seller contained in this Agreement that are not Seller Fundamental Representations shall be true and correct on and as of the Execution Date and the Closing Date as though made on and as of such dates (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date), except where all such breaches of the representations and warranties described in clause (ii) taken collectively would not have a Seller Material Adverse Effect.
- (b) Sellers shall have performed or complied in all material respects all with all obligations, covenants and agreements contained in this Agreement as to which performance or compliance by Sellers is required prior to the Closing.
- (c) Each Seller shall deliver to Buyer a certificate signed by an officer of such Seller in his or her capacity as an officer of such Seller (and not individually), dated as of the Closing Date, certifying that the conditions specified in <u>Section 8.2(a)</u> and <u>Section 8.2(b)</u> have been fulfilled (each a "Seller Officer Certificate" and, together with the Buyer Officer Certificate, the "Officer Certificates").
- (d) No material suit, action, litigation or other Proceeding instituted by any third party shall be pending before any Governmental Authority seeking to restrain, prohibit, enjoin or declare illegal, or seeking substantial damages in connection with, the transactions contemplated by this Agreement, and no statute, rule, regulation, executive order, decree, temporary restraining order, preliminary or permanent injunction, judgment or other order shall have been enacted, entered, promulgated, enforced or issued by any Governmental Authority preventing the consummation of the transactions contemplated by this Agreement;
- (e) Sellers shall be ready, willing and able to deliver to Buyer, and shall have delivered to Buyer, the closing deliverables to be delivered by Sellers in accordance with <u>Section 9.2</u>; and
- (f) The closing of the transactions contemplated in the Concurrent PSA shall have occurred concurrently with or prior to the Closing.
- 8.3 **Frustration of Closing Conditions.** Neither Buyer nor Sellers may rely, either as a basis for not consummating the transactions contemplated in this Agreement or terminating this Agreement in accordance with this Agreement and abandoning the transactions contemplated in this Agreement, on the failure of any condition set forth in Section 8.1(a) Section 8.1(b) or Section 8.1(f) or Section 8.2(a), Section 8.2(b) or Section 8.2(f), as the case may be, to be satisfied if such failure was caused by such Party's breach of any provision of this Agreement (or, as it pertains to the conditions in Section 8.1(f) and Section 8.2(f), the breach by such Party or its Affiliate of the Concurrent PSA) or failure of such Party or its Affiliate to

21

act in good faith. For clarity, for purposes of this <u>Section 8.3</u>, Sellers and the "Seller" under the Concurrent PSA are Affiliates of one another.

#### 8.4 **Termination.** This Agreement may be terminated prior to Closing:

- (a) by the mutual written consent of the Parties;
- (b) by Sellers, at their option, if any of the conditions set forth in <u>Section 8.1</u> have not been satisfied on or before the Scheduled Closing Date and, following written notice thereof from Sellers to Buyer specifying the reason such condition is unsatisfied (including any breach by Buyer of this Agreement), such condition remains unsatisfied for a period of ten (10) Business Days after Buyer's receipt of written notice thereof from Sellers;
- (c) by Buyer, at its option, if any of the conditions set forth in <u>Section 8.2</u> have not been satisfied on or before the Scheduled Closing Date and, following written notice thereof from Buyer to Sellers specifying the reason such condition is unsatisfied (including any breach by any Seller of this Agreement), such condition remains unsatisfied for a period of ten (10) Business Days after Sellers' receipt of written notice thereof from Buyer;
- (d) by Seller or Buyer, if the sum of all Title Defect Amounts for Title Defects asserted by Buyer in good faith on or before the Title Claims Date that exceed the Title Threshold exceeds thirty percent (30%) of the unadjusted Purchase Price;
- (e) by Sellers or Buyer if Closing shall not have occurred on or before the Scheduled Closing Date; or
- (f) automatically, unless the Parties agree otherwise in writing, upon the termination of the Concurrent PSA;

provided, however, that no Party shall have the right to terminate this Agreement under clauses (b), (c) or (e) above if (i) such Party is, at such time, in material breach of any provision of this Agreement, or (ii) the other Party has filed in good faith (and then solely for so long as such Party continues to diligently pursue) an action before a Governmental Authority seeking specific performance of such first Party's obligation under this Agreement to consummate the Closing as permitted by Section 11.16.

#### 8.5 Effect of Termination and Remedies.

- (a) <u>Sellers' Remedies</u>. Subject to <u>Section 8.5(c)</u>, prior to Closing, Sellers' sole and exclusive remedy for any breach by Buyer of this Agreement is (i) to seek specific performance of Buyer's obligations under this Agreement in accordance with <u>Section 11.17</u> or (ii) to terminate this Agreement in accordance with <u>Section 8.4</u>. Upon such termination, Sellers shall be free immediately to enjoy all rights of ownership of the Assets and to sell, transfer, encumber or otherwise dispose of the Assets to any Person without any restriction under this Agreement. Subject to <u>Section 8.5(c)</u>, prior to Closing, the remedies set forth in this <u>Section 8.5(a)</u> shall be Sellers' sole and exclusive remedies for Buyer's default or Sellers' termination of this Agreement, and Sellers hereby expressly waive and release all other remedies.
- (b) <u>Buyer's Remedies.</u> Subject to <u>Section 8.5(c)</u>, prior to Closing, Buyer's sole and exclusive remedy for any breach by Sellers of this Agreement is (i) to seek specific performance of Sellers' obligations under this Agreement in accordance with <u>Section 11.17</u> or (ii) to terminate this Agreement in accordance with <u>Section 8.4</u> and Buyer will be entitled to recover its actual damages against Sellers. Subject to <u>Section 8.5(c)</u>, prior to Closing, the remedies set forth in this <u>Section 8.5(b)</u> shall be Buyer's sole and

exclusive remedies for Sellers' default or Buyer's termination of this Agreement, and Buyer hereby expressly waives and releases all other remedies.

- (c) <u>Exceptions</u>. Notwithstanding anything in this Agreement to the contrary, termination of this Agreement shall not prejudice or impair Sellers' or Buyer's rights and obligations under the provisions of this <u>Section 8.5</u>, and all of <u>Article 11</u> (other than <u>Section 11.4</u>), which shall survive the termination of this Agreement.
- (d) No Other Remedies. Subject to (and except as otherwise provided in) the foregoing provisions in this Section 8.5, following any termination of this Agreement prior to Closing, no Party shall have any liability or obligation under this Agreement and Sellers shall be free immediately to enjoy all rights of ownership of the Assets and to sell, transfer, encumber or otherwise dispose of the Assets to any Person without any restriction under this Agreement; provided, however, in no event shall Sellers dispose any of the Assets if (i) there are any actions or claims pending or litigation threatened as a result of any alleged breaches of the Agreement by Sellers or (ii) Buyer is seeking specific performance pursuant to the terms of the Agreement.

# Article 9 CLOSING

#### 9.1 **Date and Place of Closing.**

- <u>Date of Closing</u>. Subject to the satisfaction or waiver of the applicable conditions to Closing, the consummation of the transactions contemplated by this Agreement (the "Closing") shall be held virtually and electronically on December 15, 2021 (such date, the "Scheduled Closing Date"): provided, that, if all conditions to Closing in Article 8 (other than those conditions that are only capable of being satisfied at Closing, but subject to all such conditions being satisfied or waived at Closing) have not been yet waived in writing or satisfied on or before the Scheduled Closing Date, then, subject to Section 8.4 (and without limiting Section 8.4(e)), the Closing shall occur within five (5) Business Days following the date on which all such conditions have been waived in writing or satisfied (other than those conditions that are only capable of being satisfied at Closing, but subject to all such conditions being satisfied or waived at Closing), or on such earlier or later date or at such other place or in such other manner as the Parties agree in writing. The date on which the Closing actually occurs is hereinafter referred to as the "Closing Date." All events of Closing shall each be deemed to have occurred simultaneously with the other, regardless of when actually occurring and each will be a condition precedent to the other. If the Closing occurs, all conditions of Closing shall be deemed to have been satisfied or waived (but Sellers' and Buyer's warranties, representations, covenants and indemnities are not waived and will survive the Closing to the extent provided in Section 10.3).
- (b) <u>Place of Closing</u>. Closing will take place remotely by electronic exchange of documents and signatures, or at such other place as mutually agreed upon by the Parties.

#### 9.2 **Closing Obligations.** At Closing:

- (a) Sellers and Buyer shall deliver to one another the Officer Certificates, as applicable;
- (b) Sellers and Buyer shall execute, acknowledge and deliver, with respect to all of the Assets, including the Mineral Interests, Royalty Interests, Overriding Royalty Interests, a deed and assignment effective as of the Effective Time substantially in the form of Exhibit B (the "Conveyance");

- (c) Sellers and Buyer shall execute and deliver an acknowledgement of the Preliminary Settlement Statement;
- (d) Buyer shall cause the Purchase Price, as adjusted in accordance with the Preliminary Settlement Statement, to be paid by wire transfer of immediately available funds to the account(s) designated by the Sellers in writing;
- (e) each Seller shall deliver to Buyer an executed certificate of non-foreign status that meets the requirements set forth in Treasury Regulations § 1.1445-2(b)(2), substantially in the form attached hereto as Exhibit C (the "FIRPTA Certificate"); and
- (f) Sellers and Buyer shall execute and deliver any other agreements, instruments or documents which are expressly required by the other terms of this Agreement to be executed and delivered at the Closing.

# Article 10 ASSUMPTION AND RETENTION OF OBLIGATIONS AND INDEMNIFICATION; DISCLAIMERS

Buyer's Assumption of Liabilities and Obligations. From and after Closing, subject to Sellers' obligations in Section 10.2(a), (a) Buyer shall assume and pay, perform, fulfill and discharge (or cause to be paid, performed, fulfilled and discharged) all claims, costs, expenses, liabilities and obligations (collectively, "Obligations") of Sellers, known or unknown, arising from, based upon, relating to or associated with the Assets, or to the ownership or management of the Assets, arising on or after the Effective Time, in each case, other than the Retained Obligations (the "Assumed Liabilities"); provided that the Assumed Liabilities shall not include, and Sellers shall retain all (and Buyer shall have no obligation with respect to any) Obligations arising from or related to the Retained Obligations. For the avoidance of doubt, any calculation or determination of any Obligations under this Agreement shall not include any measure of damages or theories of liability expressly waived by the Parties pursuant to Section 10.7.

#### 10.2 **Indemnification.**

- (a) <u>Sellers' Indemnification of Buyer</u>. From and after Closing, subject to the limitations set forth in this <u>Article 10</u> and <u>Section 11.11</u> below, Sellers shall jointly and severally be responsible for, pay on a current basis, and **DEFEND**, **INDEMNIFY AND SAVE AND HOLD HARMLESS** Buyer and its Affiliates, and its and their respective equityholders, partners, members, directors, officers, managers, employees, attorneys, consultants, agents and representatives (collectively, the "**Buyer Indemnitees**"), from and against all Losses and Obligations, whether or not incurred in the investigation or defense of any of the same or in asserting, preserving or enforcing any of their respective rights hereunder, arising directly or indirectly from, based upon, in connection with, related to, or that are associated with:
  - (i) Any breach by any Seller of any of its representations and warranties contained in this Agreement;
  - (ii) Any breach by any Seller of any of its covenants or agreements under this Agreement;
    - (iii) Seller Taxes; and
    - (iv) the Retained Obligations.

- (b) <u>Buyer's Indemnification of Sellers</u>. From and after the Closing, subject to the limitations set forth in this <u>Article 10</u> and <u>Section 11.11</u> below, Buyer shall be responsible for, pay on a current basis, and **DEFEND, INDEMNIFY AND SAVE AND HOLD HARMLESS** each Seller and its Affiliates, their respective equityholders, partners, members, directors, officers, managers, employees, attorneys, consultants, agents and representatives (collectively, the "**Seller Indemnitees**"), from and against all Losses and Obligations, whether or not incurred in the investigation or defense of any of the same or in asserting, preserving or enforcing any of their respective rights hereunder, arising directly or indirectly from, based upon, in connection with, related to, or that are otherwise associated with:
  - (i) any breach by Buyer of any of its representations or warranties contained in this Agreement;
  - (ii) any breach by Buyer of any of its covenants or agreements under this Agreement; and
    - (iii) the Assumed Liabilities.

#### (c) <u>Limitations on Indemnity</u>.

- (i) Subject to and without limiting <u>Section 11.16</u>, following the Closing, except with respect to Special Warranty and Fraud, indemnification under this <u>Section 10.2</u> shall be the sole and exclusive remedy available to each Party (and in lieu of all other remedies) against the other Party for any claims arising out of or based upon the matters set forth in this Agreement and the transactions contemplated hereby, and (except as provided in <u>Section 11.16</u>) no Party shall seek relief against the other Party other than through the indemnification provided in this <u>Section 10.2</u>.
- (ii) Sellers shall not have any liability under <u>Section 10.2(a)(i)</u> for any individual Losses arising from breaches of representations and warranties (other than the Seller Fundamental Representations) unless such individual Loss exceeds an amount equal to Five Thousand Dollars (\$5,000.00) (the "**Individual Claim Threshold**"), and then only to the extent that the aggregate amount of all such Losses which meet the Individual Claim Threshold, collectively, exceed Fifty Thousand Dollars (\$50,000.00) (the "**Deductible**"), in which event Sellers shall be liable for the aggregate amount of all such Losses.
- (iii) No indemnification may be sought by any Party under this Agreement in respect of any Loss to the extent such Loss is reflected in the calculations of the Preliminary Settlement Statement or as finally determined in connection with the Final Settlement Statement. No indemnified Party shall be entitled to recover more than once for the same Loss.
- (iv) Notwithstanding anything to the contrary contained elsewhere in this Agreement, except in cases where a Party has acted with Fraud, or except for breaches of the Seller Fundamental Representations or Buyer Fundamental Representations, as applicable, in no event shall Sellers have any obligation to indemnify Buyer or Buyer Indemnitees against, or reimburse Buyer or any Buyer Indemnitees for, any Obligations or Losses arising under Section 10.2(a)(i) in excess of an amount equal to thirty percent (30%) of the unadjusted Purchase Price. In no event shall either Party's aggregate liability under this Agreement for any and all indemnification obligations exceed an amount equal to the Purchase Price.

- (v) Any claim for indemnity to which a Seller Indemnitee or Buyer Indemnitee is entitled must be asserted by and through Sellers or Buyer, as applicable. The amount of any indemnification provided under Section 10.2(a) and Section 10.2(b) shall be reduced by any amounts actually recovered by the indemnified Party under insurance policies or otherwise from third parties, net of any collection costs (including Taxes), and excluding any insurance underwritten by the indemnified Parties.
- 10.3 **Survival of Warranties, Representations and Covenants.** All representations, warranties, covenants, and performance obligations set forth in this Agreement shall survive the Closing until fully performed, subject to applicable statutes of limitations, except:
- (a) the representations and warranties of the Seller in <u>Article 4</u> (other than the Seller Fundamental Representations) shall survive Closing for a period of twelve (12) months;
- (b) the representations and warranties of Seller set forth in Section 4.1 through Section 4.7 (inclusive) and Section 4.9 (such representations, the "Seller Fundamental Representations") shall survive the Closing without time limit, except that the representations in Section 4.9 shall survive for the applicable statute of limitations period plus sixty (60) days;
- (c) the representations and warranties of Buyer set forth in <u>Article 5</u> (other than the Buyer Fundamental Representations) shall survive Closing for a period of twelve (12) months;
- (d) the representations and warranties of Buyer set forth in <u>Section 5.1</u> through <u>Section 5.5</u> (inclusive) and <u>Section 5.8</u> (such representations, the "**Buyer Fundamental Representations**") shall survive the Closing without time limit;
  - (e) the Special Warranty shall survive Closing for a period of thirty-six (36) months;
- (f) any covenants of the Parties to be performed prior to or at Closing shall expire upon and as of Closing;
- (g) the indemnities in <u>Section 10.2(b)(iii)</u> shall survive the Closing for a period of thirty-six (36) months; and
- (h) the indemnities in <u>Section 10.2(a)(iv)</u> shall survive the Closing without time limit; and
- (i) the indemnity in Section  $\underline{10.2(a)(iii)}$  shall survive until sixty (60) days after the expiration of the applicable statute of limitations (each such period described in <u>clauses (a)</u> through  $\underline{(h)}$ , as applicable, the "Survival Period").

All indemnities herein other than those in Section 10.2(b)(iii), Section 10.2(a)(iv) and Section 10.2(a)(iii) shall terminate as of the termination date of each respective representation, warranty, covenant or performance obligation that is subject to the indemnification thereto (as specified herein), in each case, except to the extent a claim notice with respect to any breach of the foregoing has been properly delivered before the expiration of the Survival Period applicable thereto alleging a right to indemnification or defense for Obligations or Losses arising out of, relating to, or otherwise attributable to the breach of such representation, warranty, covenant or performance obligation, then, solely to the extent directly related to the matters expressly set forth in such claim notice, such representation, warranty, covenant or performance obligation shall continue to survive until the claims asserted in such claim notice

have been fully and finally resolved under the terms of this Agreement or by written agreement (including a settlement agreement) of the Parties.

- 10.4 **Reservation as to Non-Parties.** Nothing in this Agreement is intended to limit or otherwise waive any recourse Buyer or Sellers may have against any non-Party for any Obligations or Losses that may be incurred with respect to the Assets.
- 10.5 Acceptance of Title Condition. EXCEPT AS OTHERWISE SET FORTH IN THIS AGREEMENT (INCLUDING, WITHOUT LIMITATION, ANY REPRESENTATIONS, WARRANTIES AND COVENANTS OF SELLERS UNDER THIS AGREEMENT) AND EXCEPT AS TO THE SPECIAL WARRANTY OF DEFENSIBLE TITLE UNDER THIS AGREEMENT OR THE CONVEYANCE DELIVERED AT CLOSING, BUYER WILL ACCEPT THE ASSETS AT CLOSING IN THEIR PRESENT CONDITION, "AS IS AND WHERE IS AND WITH ALL FAULTS." BUYER ACKNOWLEDGES AND AGREES THAT, EXCEPT AS OTHERWISE SET FORTH IN THIS AGREEMENT AND IN THE CONVEYANCE DELIVERED AT CLOSING, SELLERS HAVE MADE NO REPRESENTATIONS OR WARRANTIES OF ANY KIND, EXPRESS OR IMPLIED, WRITTEN, ORAL, OR OTHERWISE, AS TO THE ACCURACY OR COMPLETENESS OF THE BACKGROUND MATERIALS OR ANY OTHER INFORMATION RELATING TO THE ASSETS FURNISHED BY OR ON BEHALF OF SELLERS OR TO BE FURNISHED TO BUYER OR ITS REPRESENTATIVES, INCLUDING SELLERS' INTERNAL APPRAISALS AND INTERPRETIVE DATA.
- 10.6 Express Negligence Rule. THE DEFENSE, INDEMNIFICATION, HOLD HARMLESS, RELEASE AND ASSUMED LIABILITIES PROVISIONS PROVIDED FOR IN THIS AGREEMENT SHALL BE APPLICABLE AND ENFORCEABLE WHETHER OR NOT THE LIABILITIES, LOSSES, COSTS, EXPENSES, OBLIGATIONS AND DAMAGES IN QUESTION AROSE OR RESULTED SOLELY OR IN PART FROM THE ACTIVE, PASSIVE, CONCURRENT OR COMPARATIVE NEGLIGENCE, STRICT LIABILITY OR OTHER FAULT OR VIOLATION OF LAW OF OR BY ANY INDEMNIFIED PARTY. BUYER AND SELLERS ACKNOWLEDGE THAT THIS STATEMENT COMPLIES WITH THE EXPRESS NEGLIGENCE RULE AND IS CONSPICUOUS.
- Waiver of Certain Damages. NOTWITHSTANDING ANYTHING IN THIS AGREEMENT TO THE CONTRARY, NO PARTY SHALL BE LIABLE TO THE OTHER PARTY FOR SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES SUFFERED BY SUCH PARTY RESULTING FROM OR ARISING OUT OF THIS AGREEMENT OR THE BREACH THEREOF OR UNDER ANY OTHER THEORY OF LIABILITY, WHETHER TORT, NEGLIGENCE, STRICT LIABILITY, BREACH OF CONTRACT, WARRANTY, INDEMNITY OR OTHERWISE, INCLUDING LOSS OF USE, INCREASED COST OF OPERATIONS, LOSS OF PROFIT OR REVENUE, DIMINUTION IN VALUE, LOST BUSINESS OPPORTUNITY OR BUSINESS INTERRUPTIONS. IN FURTHERANCE OF THE FOREGOING, EACH PARTY RELEASES THE OTHER PARTY AND WAIVES ANY RIGHT OF RECOVERY FOR SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES (INCLUDING ANY CALCULATIONS OF LOSSES USING ANY "MULTIPLES OF PROFITS", "MULTIPLES OF CASH-FLOW" OR OTHER SIMILAR METHODOLOGY) SUFFERED BY SUCH PARTY REGARDLESS OF WHETHER ANY SUCH DAMAGES ARE CAUSED BY THE OTHER PARTY'S NEGLIGENCE (AND REGARDLESS OF WHETHER SUCH NEGLIGENCE IS SOLE, JOINT, CONCURRENT, ACTIVE, PASSIVE OR GROSS NEGLIGENCE), FAULT, OR LIABILITY WITHOUT FAULT: PROVIDED, HOWEVER, THE FOREGOING SHALL NOT BE CONSTRUED AS LIMITING AN OBLIGATION OF A PARTY TO INDEMNIFY, RELEASE, DEFEND AND HOLD HARMLESS THE OTHER PARTY AGAINST CLAIMS ASSERTED BY THIRD PARTIES, INCLUDING, THIRD PARTY CLAIMS FOR SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES.

27

- 10.8 **Redhibition Waiver.** BUYER: (A) WAIVES ALL RIGHTS IN REDHIBITION PURSUANT TO LOUISIANA CIVIL CODE ARTICLE 2475 AND ARTICLES 2520 THROUGH 2548; (B) ACKNOWLEDGES THAT THIS EXPRESS WAIVER SHALL BE CONSIDERED A MATERIAL AND INTEGRAL PART OF THIS SALE AND THE CONSIDERATION THEREOF; AND (C) ACKNOWLEDGES THAT THIS WAIVER HAS BEEN BROUGHT TO THE ATTENTION OF BUYER, HAS BEEN EXPLAINED IN DETAIL AND THAT BUYER HAS VOLUNTARILY AND KNOWINGLY CONSENTED TO THIS WAIVER OF WARRANTY OF FITNESS AND WARRANTY AGAINST REDHIBITORY VICES AND DEFECTS WITH RESPECT TO THE ASSETS.
- 10.9 UTPCPL Waiver. TO THE EXTENT APPLICABLE TO THE ASSETS OR ANY PORTION THEREOF, BUYER HEREBY WAIVES THE PROVISIONS OF THE LOUISIANA UNFAIR TRADE PRACTICES AND CONSUMER PROTECTION LAW (LA. R.S. 51.1401, ET SEQ.). BUYER WARRANTS AND REPRESENTS THAT IT: (A) IS EXPERIENCED AND KNOWLEDGEABLE WITH RESPECT TO THE OIL AND GAS INDUSTRY GENERALLY AND WITH TRANSACTIONS OF THIS TYPE SPECIFICALLY; (B) POSSESSES AMPLE KNOWLEDGE, EXPERIENCE AND EXPERTISE TO EVALUATE INDEPENDENTLY THE MERITS AND RISKS OF THE CONTEMPLATED TRANSACTIONS; AND (C) IS NOT IN A SIGNIFICANTLY DISPARATE BARGAINING POSITION.
- 10.10 **Determination of Losses.** With respect to any Party's indemnification obligations under this <u>Article 10</u>, solely for purposes of determining the amount of Losses resulting from a breach or inaccuracy of a representation or warranty of the Sellers in <u>Article 4</u> or of the Buyer in <u>Article 5</u> (including any bringdown of any such representation or warranty in any certificate delivered pursuant to this Agreement), such representations and warranties shall be read without giving effect to any materiality, Material Adverse Effect or similar qualification therein.
- Party in writing (including the full particulars of such claim) of any potential Obligations or Losses of which it becomes aware and for which it is entitled to indemnification from such other Party under this Agreement; <a href="mailto:provided">provided</a> that the failure to so notify such other Party shall not affect the other Party's indemnification obligations under this Agreement with respect to such claim except to the extent that such other Party is materially prejudiced by such failure. The indemnifying Party is obligated to defend at the indemnifying Party's sole expense any litigation or other administrative or adversarial proceeding against the indemnified Party relating to any claim for which the indemnifying Party has agreed to indemnify and hold the indemnified Party harmless under this Agreement; <a href="provided">provided</a>, <a href="however">however</a>, the indemnified Party shall have the right to participate with the indemnifying Party in the defense of any such claim at its own expense.

#### Article 11 MISCELLANEOUS

- 11.1 **Expenses.** Except as otherwise specifically provided in this Agreement, all fees, costs and expenses incurred by Buyer, on the one hand, or Sellers, on the other, in negotiating this Agreement and the Ancillary Agreements or in consummating the transactions contemplated by this Agreement shall be paid by the Party incurring the same.
- 11.2 **Notices.** All notices and communications required or permitted to be given under this Agreement, shall be sufficient in all respects if given in writing and delivered personally, sent by electronic mail (so long as such electronic mail is acknowledged by the recipient as having been received; provided that such recipient is affirmatively obligated to promptly acknowledge receipt), sent by bonded overnight courier or mailed by U.S. Express Mail, Federal Express or United Parcel Service Express Delivery or by certified or registered United States Mail with all postage fully prepaid, addressed to the appropriate person

#### at the address for such Person shown below:

#### If to Sellers:

c/o SRP Management Services, LLC 3811 Turtle Creek Blvd, Suite 1100 Dallas, Texas 75219 Attn: Matt Veazey, Director

Email: matt.veazey@srp-ok.com Telephone: 214-699-4954

with a copy (which will not constitute notice) to:

Kirkland & Ellis LLP 609 Main Street Houston, Texas 77002 Attn: Rahul D. Vashi, P.C. Lindsey M. Jaquillard

Email: rahul.vashi@kirkland.com

lindsey.jaquillard@kirkland.com

Telephone: 713-836-3600

#### If to Buyer:

PHX Minerals Inc. 1601 NW Expressway Valliance Bank Building, Suite 1100 Oklahoma City, Oklahoma 73118

Attn: Chad Stephens Email: chad@phxmin.com Telephone: (405) 945-6105

with a copy (which will not constitute notice) to:

Jackson Walker LLP 1401 McKinney St., Suite 1900 Houston, Texas 77010 Attn: Kirk Tucker

Email: ktucker@jw.com Telephone: (713) 752.4389

Any notice given in accordance with this Agreement shall be deemed to have been given when delivered to the addressee in person, by electronic mail (provided such electronic mail is acknowledged by the recipient as described above) or by courier during normal business hours, or upon actual receipt by the addressee after such notice has either been delivered to an overnight courier or deposited in the United States Mail or with Federal Express or United Parcel Service, as the case may be. Any Party may change its contact information for notice by giving written notice to the other Party in the manner provided in this Section 11.2. If the date specified in this Agreement for giving any notice or taking any action is not a Business Day (or if the period during which any notice is required to be given or any action taken expires

on a date which is not a Business Day), then the date for giving such notice or taking such action (and the expiration date of such period during which notice is required to be given or action taken) shall be the next day which is a Business Day.

#### 11.3 Governing Law; Jurisdiction; Venue: Jury Waiver; Binding Arbitration.

- (a) This Agreement, the Ancillary Agreements and any other documents delivered pursuant hereto and the legal relations between the Parties shall be governed and construed in accordance with the laws of the State of Texas, without giving effect to principles of conflicts of laws that would result in the application of the laws of another jurisdiction; <u>provided</u> that any dispute pertaining to real property shall be governed by the laws of the state in which such real property is located.
- Subject to and without limiting Section 2.4(c), the Parties agree that any and all disputes or claims by any Party arising from or related to this Agreement that cannot be amicably settled, will be determined solely and exclusively by arbitration in accordance with the Federal Arbitration Act and using the rules of the American Arbitration Association or any successor thereof when not in conflict with such act; provided, however, that nothing contained in this Section 11.3 shall limit any Party's right to bring (i) post-arbitration actions seeking to enforce an arbitration award or (ii) actions seeking injunctive or other similar relief under Section 11.16 in the event of a breach or threatened breach of any of the provisions of this Agreement (or any other agreement contemplated hereby). Arbitration will take place at an appointed time and place in Dallas, Texas, at a mutually agreeable and neutral venue. The Parties to the dispute will attempt in good faith to agree on a single impartial arbitrator. If such Parties cannot agree on a single impartial arbitrator, each Party to the dispute will select one impartial arbitrator, and the two so designated will select a third (3rd) impartial arbitrator. If either Party to the dispute should fail to designate an arbitrator within fourteen (14) days after arbitration is requested, or if the two (2) arbitrators should fail to select a third arbitrator within thirty (30) days after arbitration is requested, then an arbitrator will be selected in accordance with the Commercial Arbitration Rules of the American Arbitration Association. Each arbitrator will have at least ten (10) years' experience in the oil and gas industry in the State of Texas (or, if the dispute pertains to real property, the state where such real property is located). Discovery will be made pursuant to the Federal Rules of Civil Procedure. Final hearing on the matter will be held within nine (9) months after the date the Parties to the dispute agree on a single arbitrator or the selection of the third arbitrator, as applicable (such date, the "Arbitration Commencement Date") and a final decision with a written opinion stating the reasons therefor will be rendered within fourteen (14) days after the conclusion of the testimony at the final hearing, which written opinion shall be final and binding; provided, however, that Sellers will have the exclusive right, in their discretion, to require that (A) discovery be completed by a specific date, which date must be at least forty-five (45) days after the Arbitration Commencement Date, or (B) the final hearing be commenced by a date earlier than nine (9) months after the Arbitration Commencement Date, which date must be at least sixty (60) days after the Arbitration Commencement Date. The prevailing Party in the arbitration proceeding shall recover all reasonable and necessary attorneys' fees and costs, including, but not limited to, all attorneys' fees, expert witness fees, court reporter fees, support staff fees, arbitration venue fees, and arbitrator fees. A Party need not be awarded any damages to be considered the prevailing Party, and a Party need not succeed on all or any affirmative claims to be considered the prevailing Party. The Parties irrevocably waive their right to any form of appeal, review or recourse to any state court or other judicial authority, insofar as such waiver may be validly made. Judgment upon an award of the majority of the arbitrators will be binding. Judgment on the award may be entered and enforced by any court of competent jurisdiction. In no event, may the arbitrators award damages that have been waived under this Agreement. The arbitration process will be kept confidential and such conduct, statements, promises, offers, views and opinions will not be discoverable or admissible in any legal proceeding for any purpose, except to the extent reasonably necessary to enforce the final decision of the arbitrators.

- FOR PURPOSES OF (I) ANY POST-ARBITRATION ACTIONS SEEKING TO ENFORCE AN ARBITRATION AWARD UNDER THIS AGREEMENT OR (II) ACTIONS SEEKING INJUNCTIVE OR OTHER SIMILAR RELIEF UNDER SECTION 11.16, THE VENUE FOR ANY SUCH ACTION BROUGHT UNDER THIS AGREEMENT SHALL BE DALLAS COUNTY, TEXAS. EACH PARTY CONSENTS TO PERSONAL JURISDICTION IN ANY ACTION BROUGHT IN THE UNITED STATES FEDERAL COURTS LOCATED WITHIN DALLAS COUNTY, TEXAS (OR, IF JURISDICTION IS NOT AVAILABLE IN THE UNITED STATES FEDERAL COURTS, TO PERSONAL JURISDICTION IN ANY ACTION BROUGHT IN THE STATE COURTS LOCATED IN DALLAS COUNTY, TEXAS) WITH RESPECT TO ANY DISPUTE, CLAIM OR CONTROVERSY ARISING OUT OF OR IN RELATION TO OR IN CONNECTION WITH THIS AGREEMENT, AND EACH OF THE PARTIES AGREES THAT ANY ACTION INSTITUTED BY IT AGAINST THE OTHER WITH RESPECT TO ANY SUCH DISPUTE, CONTROVERSY OR CLAIM WILL BE INSTITUTED EXCLUSIVELY IN THE UNITED STATES DISTRICT COURT FOR THE NORTHERN DISTRICT OF TEXAS. THE PARTIES HEREBY WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM BROUGHT BY ANY PARTY AGAINST ANOTHER IN ANY MATTER WHATSOEVER ARISING OUT OF OR IN RELATION TO OR IN CONNECTION WITH THIS AGREEMENT. IN ADDITION, EACH PARTY IRREVOCABLY WAIVES ANY OBJECTION, INCLUDING ANY OBJECTION TO THE LAYING OF VENUE OR BASED ON THE GROUNDS OF FORUM NON CONVENIENS, WHICH IT MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY SUCH ACTION IN THE RESPECTIVE JURISDICTIONS REFERENCED IN THIS SECTION.
- 11.4 **Further Assurances.** From time to time after Closing, Sellers and Buyer shall each execute, acknowledge and deliver to the other such further instruments and take such other action as may be reasonably requested in order to accomplish more effectively the purposes of the transactions contemplated by this Agreement.
- 11.5 **Amendments: Waivers.** This Agreement may not be amended, nor any rights hereunder waived except by an instrument in writing signed by the Party to be charged with such amendment or waiver and delivered by such Party to the Party claiming the benefit of such amendment or waiver. No waiver of any breach or violation or, default under or inaccuracy in any representation, warranty or covenant set forth in this Agreement will be deemed to extend to any prior or subsequent breach, violation, default, inaccuracy or affect in any way any rights arising by virtue of any prior or subsequent occurrence. No delay or omission on the part of any Party in exercising any right, power or remedy under this Agreement will operate as a waiver thereof.
- Agreement or any of its rights under this Agreement without the prior written consent of the other Parties (which consent may be granted or withheld in each such other Party's respective sole discretion); provided, however, that Buyer shall have the right, upon prior written notice to Sellers, to assign, to one or more of its Affiliates, Buyer's rights under this Agreement to acquire the Assets at Closing; provided further, that without the prior written consent of the other Parties, no assignment pursuant to this Section 11.6 shall relieve any Party from its obligations under this Agreement.
- 11.7 **Counterparts/Fax Signatures.** This Agreement may be executed and delivered in one or more counterparts, each of which when executed and delivered shall be an original, and all of which when executed shall constitute one and the same instrument. The exchange of copies of this Agreement and of signature pages by facsimile or by electronic image scan transmission in .pdf format shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes.

31

#### 11.8 Construction.

- (a) Capitalized terms used but not otherwise defined in this Agreement have the meanings given to such terms in <u>Annex I</u>.
- (b) The headings of the Articles and Sections of this Agreement are for guidance and convenience of reference only and shall not limit or otherwise affect any of the terms or provisions of this Agreement. The Disclosure Schedules are hereby incorporated in this Agreement by reference and constitute a part of this Agreement. The words "this Agreement," "herein," "hereby," "hereunder" and "hereof," and words of similar import, refer to this Agreement as a whole and not to any particular Article, Section, subsection or other subdivision unless expressly so limited. All references to "\$" or "dollars" shall be deemed references to United States dollars. References made in this Agreement, including use of a pronoun, shall be deemed to include where applicable, masculine, feminine, singular or plural, individuals or entities. The word "or" when used in a list shall not indicate that the listed items are exclusive of each other. The words "include," "includes," and "including" are deemed to be followed by "without limitation" whether or not they are in fact followed by such words or words of similar import. References to any Law, Contract, agreement or other instrument as it may be amended from time to time.
- 11.9 **Entire Agreement.** This Agreement and the Ancillary Agreements constitute the entire understanding among the Parties, their respective partners, members, trustees, shareholders, officers, directors and employees with respect to the subject matter hereof, superseding all negotiations, prior discussions and prior agreements and understandings relating to such subject matter, including any letter of intent entered into between the Parties.
- 11.10 Successors and Permitted Assigns. This Agreement shall be binding upon, and shall inure to the benefit of, the Parties, and their respective successors and permitted assigns.
- 11.11 **Parties in Interest.** Nothing in this Agreement shall entitle any Person, other than the Parties, or the Parties' respective related indemnified parties hereunder, to any claim, cause of action, remedy or right of any kind; <u>provided</u>, that only a Party will have the right to enforce the provisions of this Agreement on its own behalf or on behalf of its related indemnified parties (but shall not be obligated to do so).
- 11.12 **Severability.** In the event any term of this Agreement is found by any court to be void or otherwise unenforceable, the remainder of this Agreement shall remain valid and enforceable, and, to the fullest extent permitted by Law, such offending term or terms shall be replaced with an enforceable term or enforceable terms that as nearly as possible effect the Parties' intent.
- 11.13 **Joint Preparation.** The Parties acknowledge and agree that this Agreement was jointly prepared by the Parties, and that each Party was afforded the opportunity to consult with counsel in the preparation and negotiation of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties and no presumption or burden of proof shall arise favoring or disfavoring either Party by virtue of the authorship of any of the provisions of this Agreement.
- 11.14 **Publicity.** Sellers and Buyer shall consult with each other with regards to all publicity and other releases concerning this Agreement and the transactions contemplated hereby and, except as required by applicable Law or the applicable rules or regulations of any Governmental Authority or stock exchange, neither Party shall issue any such publicity or other release without the prior written consent of the other Party hereto (such consent not to be unreasonably withheld, conditioned or delayed). Without the prior

written consent of the other Party, neither Buyer nor Sellers shall disclose or issue any press release or public announcement that includes the name of the non-disclosing Party; <u>provided</u>, <u>however</u>, the foregoing shall not restrict disclosures in order to comply with applicable securities or other laws, to comply with demands or requests from regulatory authorities, or to comply with existing loan or other agreements binding upon such Party.

- 11.15 **Disclosure Schedules.** The matters set forth on the disclosure schedules hereto (collectively, the "**Disclosure Schedules**") are not necessarily matters that Sellers are required to disclose or matters that would constitute a breach of any representation or warranty had such matters not been disclosed. No disclosure by any Seller in the Exhibits or the Disclosure Schedules relating to any possible breach or violation of any contract or applicable Law shall be construed as an admission or indication that such breach or violation exists or has actually occurred.
- 11.16 **Specific Performance.** Subject to and without limiting Section 8.5, from and after Closing, if either Sellers or Buyer violates or fails or refuses to perform any covenant or agreement made by Sellers or Buyer (as applicable) in this Agreement, the non-breaching Party (in the case of Buyer) or Parties (in the case of Sellers), subject to the terms of this Agreement and in addition to any remedy at Law for damages or other relief permitted under this Agreement, may institute and prosecute an action in any court of competent jurisdiction to enforce specific performance of such covenant or agreement or seek any other equitable relief, without the necessity of proving actual damages or posting of a bond. Each of Sellers and Buyer hereby acknowledges and agrees that the rights of the other to consummate the transactions contemplated by this Agreement are special, unique and of extraordinary character and that, if any Seller or Buyer violates or fails or refuses to perform any covenant or agreement made by it in this Agreement, the non-breaching Party (in the case of Buyer) or Parties (in the case of Sellers) may be without an adequate remedy at Law.
- 11.17 **Confidentiality.** Subject to Section 11.14, for a period of eighteen (18) months after the Closing Date, each Seller shall, and shall cause its Affiliates to, not use or make disclosures to third parties of any confidential or proprietary information relating to Assets except with the prior written consent of Buyer or as required by applicable Law, except to the extent that such information (a) is generally available to the public through no breach by such Seller or its Affiliates of this Section 11.17, (b) is lawfully acquired by such Seller or its Affiliates after the Closing from sources which are not actually known to such Seller to be prohibited from disclosing such information due to confidentiality obligations owed to Buyer, (c) is required for purposes of compliance by such Seller or its Affiliates with Tax or regulatory reporting requirements, (d) is needed in connection with the exercise, enforcement or performance of Sellers' rights and obligations hereunder, the Ancillary Agreements or the enforcement, performance or defense of the same, or (e) is disclosed (i) in the course of any trial or other legal Proceeding involving Sellers or their Affiliates, (ii) as required by any applicable securities Law or legal process or other Law (including any subpoena, interrogatory, or other similar requirement for such information to be disclosed) or rules of any applicable national stock exchange, or (iii) as is necessary to allow Sellers or their Affiliates to manage or operate the Excluded Assets or Seller's or its Affiliates' other assets or Retained Obligations; provided, however, that (A) nothing shall prohibit such Seller or its Affiliates from using such information in the conduct of their respective businesses following the Closing so long as such use is consistent in all material respects with the manner in which such Seller or its Affiliates used such information prior to the Closing and (B) such Seller and its Affiliates may discuss the underlying investment with respect to the Assets and the acquisition or disposition of the Assets with its Representatives and current and potential partners or investors or in connection with legitimate fundraising activities or fund performance reporting with current or prospective investors, lenders or partners.
- 11.18 **No Recourse.** Except as provided in this <u>Section 11.18</u>, nothing express or implied in this Agreement or any document, agreement or instrument delivered in relation to this Agreement is intended

or shall be construed to confer upon or give any Person, other than the Parties, any right or remedies under or by reason of this Agreement. All claims, obligations, liabilities or causes of action (whether in contract or in tort, in Law or in equity or granted by statute) that may be based upon, in respect of, arise under, out or by reason of, be connected with or relate in any manner to this Agreement, or the negotiation, execution or performance of this Agreement (including any representation or warranty made in, in connection with or as an inducement to, this Agreement) and the transactions contemplated hereby, may be made only against (and such representations and warranties are those solely of) the Parties (the "Contracting Parties"). No Person who is not a Contracting Party, including any past, present or future director, officer, employee, incorporator, member, partner, manager, equityholder, Affiliate, agent, attorney, representative or assignee of, and any financial advisor or lender to, any Contracting Party, or any past, present or future director, officer, employee, incorporator, member, partner, manager, equityholder, Affiliate, agent, attorney, representative or assignee of, and any financial advisor or lender to, any of the foregoing (collectively, the "Nonparty Affiliates"), shall have any liability (whether in contract or in tort, in law or in equity, or granted by statute) for any claims, causes of action, obligations or liabilities arising under, out of, in connection with or related in any manner to this Agreement or the transactions contemplated hereby or based on, in respect of or by reason of this Agreement or its negotiation, execution, performance or breach of this Agreement and the transactions contemplated hereby, and, to the maximum extent permitted by Law, each Contracting Party hereby waives and releases all such liabilities, claims, causes of action and obligations against any such Nonparty Affiliates. Without limiting the foregoing, to the maximum extent permitted by Law, each Contracting Party disclaims any reliance upon any Nonparty Affiliates with respect to the performance of this Agreement or any representation or warranty made in connection with, or as an inducement to, this Agreement. This Section 11.18 shall not be effective as to any Seller to the extent such Seller makes any distribution to its members, including of any portion of the Purchase Price, that would violate any provision of Subchapter VIII of the Delaware Limited Liability Company Act or similar Law.

[Signature Pages to Follow]

IN WITNESS WHEREOF, the Parties have duly executed this Agreement as of the Execution Date.

#### **SELLERS:**

# **Merrimac Properties Partners, LLC**

By: /s/ Ryan Turner
Name:Ryan Turner
Title: Managing Person

By: /s/ William T. Fennebresque
Name: William T. Fennebresque

Title: Managing Person

# **Quarter Horse Energy Partners, LLC**

By: /s/ Ryan Turner
Name:Ryan Turner
Title: Managing Person

By: /s/ William T. Fennebresque
Name: William T. Fennebresque

Title: Managing Person

# **BUYER:**

# **PHX Minerals Inc.**

By: /s/ Chad L. Stephens III
Name: Chad L. Stephens III
Title: President and Chief Executive Officer

#### ANNEX I

#### **DEFINED TERMS**

- "AAA" has the meaning set forth in Section 3.2(g)(iv).
- "Accounting Consultant" has the meaning set forth in Section 2.4(c).
- "Acquisition Proposals" has the meaning set forth in Section 6.6.
- "Actual NRA" means the actual Net Royalty Acres determined by the Parties or the Title Consultant, as the case may be, to be owned by Sellers in each Tract pursuant to Article 3.
- "Affiliate" means, with respect to any Person, any other Person that directly or indirectly controls, is controlled by or is under common control with, such first Person. For the purposes of this definition, "control" (including, with correlative meanings, the terms "controlling," "controlled by" and "under common control with"), as applied to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of that Person, whether through the ownership of voting securities, by contract or otherwise.
  - "Agreement" has the meaning set forth in the introductory paragraph of this Agreement.
  - "Allocated Value" has the meaning set forth in Section 2.2.
  - "Ancillary Agreements" means the Conveyance and the FIRPTA Certificate.
  - "Arbitration Commencement Date" has the meaning set forth in Section 11.3(b).
- "Asset Taxes" means ad valorem, property, excise, severance, production and similar Taxes based upon the operation or ownership of the Assets or the production of Hydrocarbons or the receipt of proceeds therefrom, but excluding, for the avoidance of doubt, Income Taxes and Transfer Taxes.
- "Assets" means, save, except for and excluding the Excluded Assets, all of Sellers' collective right, title and interest in and to the following assets and properties from and after the Effective Time:
- (a) all fee mineral interests and lessor royalty interests in all oil, natural gas, coalbed methane and other liquid or gaseous Hydrocarbons, as well as their respective constituent products (including condensate, casinghead gas, distillate and natural gas liquids), and any other minerals, similar or dissimilar, customarily produced in conjunction therewith (all such substances are defined for purposes of this Agreement as a "Mineral" or the "Minerals"), in, on and under or produced and saved from the lands, tracts and properties described in Exhibit A-1 attached hereto (such lands, tracts, and properties described in Exhibit A-1, the "Lands"), and from all wells located on the Lands (collectively, the "Wells"), together with any royalty interests attributable thereto and any units, lands, tracts or other properties pooled with any of the Lands (collectively, the "Mineral Interests");
- (b) any overriding royalty interests held by Sellers burdening oil, gas or other minerals produced, saved or sold from the Lands, including those described in <u>Exhibit A-1</u> attached hereto (collectively, the "Overriding Royalty Interests");

- (c) the non-participating royalty interests burdening the Lands, including those described in <u>Exhibit A-1</u> attached hereto (collectively, the "**Royalty Interests**", together with the Mineral Interests and Overriding Royalty Interests, the "**Interests**");
- (d) any oil, gas and mineral leases that relate to the Interests, including all reversionary rights thereunder, including those described in <u>Exhibit D</u>, which <u>Exhibit D</u> shall be provided by Seller to Buyer, based on information reasonably available in Sellers' files, within five (5) calendar days from the Execution Date (collectively, the "**Leases**");
- (e) all division and transfer orders, and, except for the Leases, all written contracts, contractual rights, interests and other written agreements to which any Seller is a party (including as assignee) and under which any of the Interests are bound (collectively, the "Contracts");
  - (f) all Records, except for the Excluded Records;
- (g) (i) all rights of ingress and egress, at all times for the purpose of exploring, drilling, operating for and producing from the Lands for Minerals and removing the same therefrom, and of laying pipelines, storing oil, building tanks, processing and treating plants and facilities, power stations, telephone lines, roads and other structures necessary to produce, save, care for, treat, store, compress, process, and transport said products, attributable to the conveyed interests, and (ii) all rights with respect to the use and occupation of the surface of the Lands and the subsurface depths under the Lands for the sole purpose of development and maintenance of the Interests; and (iii) the right of ingress and egress across any lands owned or controlled by Sellers to the extent such ingress and egress across such lands is reasonable to obtain access for the development and maintenance of the Interests;
- (h) all proceeds attributable to any of the Interests or the Leases and attributable to the time period after the Effective Time;
- (i) to the extent transferable (provided that Sellers will not be required to pay any fees or incur any liabilities in order to transfer such data), all proprietary and non-proprietary geologic, geophysical and seismic data that relate to the Interests (excluding any interpretive data);
- (j) all permits to the extent primarily relating to or applicable to any of the Interests or Leases and (i) required for the ownership of the Interests or Leases and (ii) transferable pursuant to applicable Law;
- (k) to the extent relating to the Assets and to the extent assignable, all rights and interests of either Seller (i) under any agreement or policy of indemnity or insurance (including any rights, claims or causes of action of either Seller against third parties under any indemnity or hold harmless agreements) and any indemnities received in connection with Sellers' prior acquisition of any of the Assets, but in each case only to the extent that such rights or interests arise from obligations or liabilities for which Buyer is responsible under this Agreement, or (ii) relating to claims and causes of action that may be asserted against a third party to the extent such rights and claims arise from obligations or liabilities assumed by Buyer hereunder; and
- (1) except to the extent pertaining to any Retained Obligations, and only to the extent assignable, all rights, claims and causes of action to the extent, and only to the extent, that such rights, claims or causes of action are associated with any of the other Assets and relate to the period from and after the Effective Time.

"Assumed Liabilities" has the meaning set forth in Section 10.1.

- "Business Day" means any weekday other than a weekday on which commercial banks in the city of Oklahoma City, Oklahoma and Dallas, Texas are authorized or required by applicable Law to be closed.
  - "Buyer" has the meaning set forth in the introductory paragraph of this Agreement.
  - "Buyer Fundamental Representations" has the meaning set forth in Section 10.3.
  - "Buyer Indemnitees" has the meaning set forth in Section 10.2(a).
  - "Buyer Material Adverse Effect" means a Material Adverse Effect with respect to Buyer.
  - "Buyer Officer Certificate" has the meaning set forth in Section 8.1(c).
  - "Closing" has the meaning set forth in <u>Section 9.1</u>.
  - "Closing Date" has the meaning set forth in Section 9.1.
  - "Code" has the meaning set forth in Section 4.6.
  - "Concurrent PSA" has the meaning set forth in Section 8.1(f).
  - "Contracting Party" has the meaning set forth in <u>Section 11.18</u>.
  - "Contracts" has the meaning set forth in the definition of "Assets."
  - "Conveyance" has the meaning set forth in Section 9.2(b).
  - "Credit Deficiency Notice" has the meaning set forth in Section 3.2(b).
  - "Credit Notice" has the meaning set forth in Section 3.2(b).
  - "Cure Notice" has the meaning set forth in Section 3.2(f).
  - "Cure Period" has the meaning set forth in Section 3.2(f).
  - "Deductible" has the meaning set forth in Section 10.2(c)(ii).
  - "**Defect Allegation**" has the meaning set forth in Section 3.2(a).
- "Defensible Title" means such title of Sellers to the Assets that, as of the Effective Time and as of the Closing Date, is deducible of record or provable by documentation that would be successfully defended if challenged and that, subject to the Permitted Encumbrances, (a) as to each Tract, entitles Sellers to not less than the Stated NRA for such Tract and (b) is free and clear of all Encumbrances.
  - "**Deficiency Notice**" has the meaning set forth in Section 3.2(a).
  - "Delayed Closing Date" has the meaning set forth in Section 3.2(f).
  - "Disclosing Credit Party" has the meaning set forth in Section 3.2(b).
  - "Disclosure Schedules" has the meaning set forth in Section 11.15.

"Dispute Notice" has the meaning set forth in Section 2.4(a).

"Effective Time" means, as to each Asset subject to this Agreement, the same date and time that was deemed the "Effective Time" (or such similar definition) in the applicable acquisition agreement or other instrument of conveyance through which the applicable Seller acquired its interest in, and derives its title to, such Asset (each, a "Title Instrument"), as specified in the applicable Title Instrument. "Encumbrance" means any lien, judgment, mortgage, deed of trust, security instrument, pledge or option that is binding upon the Assets, or other preferential arrangement or other similar encumbrance.

"Environmental Laws" means the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. § 9601 et seq.; the Resource Conservation and Recovery Act, 42 U.S.C. § 6901 et seq.; the Federal Water Pollution Control Act, 33 U.S.C. § 1251 et seq.; the Clean Air Act, 42 U.S.C. § 7401 et seq.; the Hazardous Materials Transportation Act, 49 U.S.C. § 5101 et seq.; the Toxic Substances Control Act, 15 U.S.C. §§ 2601 through 2629; the Oil Pollution Act, 33 U.S.C. § 2701 et seq.; the Emergency Planning and Community Right to Know Act, 42 U.S.C. § 11001 et seq.; and the Safe Drinking Water Act, 42 U.S.C. §§ 300f through 300j, and all other applicable Laws of any Governmental Authority having jurisdiction over the property in question addressing (a) pollution or pollution control; (b) protection of natural resources, the environment or biological resources; or (c) the disposal or release or threat of release of hazardous substances, in each case as enacted and in effect on or prior to the Closing Date.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

#### "Excluded Assets" means the following assets and properties:

- (a) all proceeds attributable to any of the Assets to the extent attributable to the time period before the Effective Time, including any lease bonuses, royalties and other proceeds and benefits to the extent (i) attributable to the time period before the Effective Time and (ii) received no later than one (1) year following the Closing Date;
- (b) all claims of Seller or any of its Affiliates for refunds of, rights to receive funds from any Governmental Authority, or loss carry forwards or credits with respect to Seller Taxes;
- (c) any refunds of costs or other expenses borne by any Seller or its predecessors in title, insofar as such refunds relate to time periods prior to the Effective Time or to any Retained Obligations;
  - (d) the Excluded Records;
- (e) all claims and causes of action of Sellers relating to (i) any Retained Obligation or (ii) production that occurred prior to the Effective Time, and all audit rights arising under any Contracts with respect to the time periods prior to the Effective Time or any Retained Obligations; provided, however, if any third party makes a claim regarding time periods prior to the Effective Time that is or would be an Assumed Liability, then the claims and causes of action against such third party, to the extent relating to such Assumed Liability, that would otherwise be Excluded Assets pursuant to this clause (e), shall become Assets;
- (f) all computer or communications software owned, licensed or used by Sellers, other than the Records;

- (g) any logo, service mark, copyright, trade name or trademark of or associated with Sellers or any Affiliate of Sellers or any business of Sellers or of any Affiliate of Sellers;
  - (h) motor vehicles and other rolling stock owned by Sellers; and
- (i) all oil, gas and Hydrocarbons produced, saved or sold from the Assets with respect to all periods prior to the Effective Time.

"Excluded Records" means (a) any files, records, or data (other than title opinions or memoranda relating to the Assets), which Sellers believe in good faith are protected by attorney-client privilege or confidentiality agreements, provided that Sellers have used Reasonable Efforts to obtain waivers of such confidentiality agreements, (b) all of Sellers' corporate minute books, corporate financial records and Income Tax records that relate to Sellers' business generally, (c) all files, records or data relating to the marketing process for the Assets including any and all bids received in connection therewith, and (d) (i) Seller's interpretation of any geologic, geophysical and seismic data, and (ii) any non-proprietary geologic, geophysical and seismic data that relate to the Interests, to the extent that (A) such non-proprietary geologic, geophysical and seismic data are not transferable without consideration or without a third party's consent, and (B) such consent was not obtained from such third party or Buyer has not agreed in writing to pay such consideration prior to the time of Closing.

"Execution Date" has the meaning set forth in the introductory paragraph of this Agreement.

"Final Price" has the meaning set forth in Section 2.4(a).

"Final Settlement Date" has the meaning set forth in Section 2.4(a)

"Final Settlement Statement" has the meaning set forth in Section 2.4(a).

"FIRPTA Certificate" has the meaning set forth in Section 9.2(e).

"Fraud" means an intentional and knowing misrepresentation of fact with the intent to deceive solely with respect to the making of the representations and warranties in this Agreement or any instrument or other agreement entered into in connection herewith which constitutes common law fraud under Delaware law.

"GAAP" means United States generally accepted accounting principles as in effect from time to time, consistently applied.

"Governmental Authority" means any federal, state, local, municipal, tribal or other government; any governmental, regulatory or administrative agency, commission, body or other authority exercising or entitled to exercise any administrative, executive, judicial, legislative, regulatory or taxing authority or power, and any court or governmental tribunal, including any tribal authority having or asserting jurisdiction.

"Hazardous Substances" means any pollutants, contaminants, toxins or hazardous or extremely hazardous substances, materials, wastes, constituents, compounds or chemicals that are regulated by, or will, given extent levels, form the basis of remediation liability under, any Environmental Laws (including NORM and other substances) in effect as of the Execution Date and each Closing Date.

"Hydrocarbons" means any oil, gas, casinghead gas, condensate, distillate or other liquid or gaseous hydrocarbon of every kind or description.

"Income Taxes" means any U.S. federal, state or local or non-U.S. income Tax or Tax based on profits, net profits, margin revenues or similar Taxes (including franchise Taxes and any capital gains and net worth Taxes).

"Individual Claim Threshold" has the meaning set forth in Section 10.2(c)(ii).

"Interests" has the meaning set forth in the definition of "Assets."

"**Knowledge**" means the actual knowledge, without any obligation of inquiry, of (x) in the case of Sellers, of Matt Veazey or (y) in the case of Buyer, of Ralph D'Amico.

"Lands" has the meaning set forth in the definition of "Assets."

"Law" means any applicable statute, law, rule, regulation, ordinance, order, code, ruling, writ, injunction, decree or other official act of or by any Governmental Authority.

"Leases" has the meaning set forth in the definition of "Assets."

"Losses" means any and all losses, costs, expenses (including court costs, reasonable fees and expenses of attorneys, technical experts and expert witnesses and the cost of investigation), liabilities, damages, diminution of value of any Asset, demands, suits, claims, causes of action and sanctions of every kind and character (including civil fines); provided, however, for the avoidance of doubt, any calculation or determination of Losses shall not include any measure of damages or theories of liability expressly waived by the Parties pursuant to Section 10.7.

"Material Adverse Effect" means, with respect to a Party, any circumstance, change or effect that has had or would be reasonably likely to (a) have, individually or in the aggregate with any other circumstance, change or effect, a material and adverse effect on the ownership or financial condition of the Assets in excess of an amount equal to fifteen percent (15%) of the Purchase Price (with respect to a Seller Material Adverse Effect) or the assets and properties of Buyer (with respect to a Buyer Material Adverse Effect), taken as a whole or (b) materially impair, prevent or delay such Party's timely consummation of the transactions contemplated hereby, but shall exclude any circumstance, change or effect resulting or arising from: (i) any change in general conditions in the industries or markets in which such Party operates; (ii) seasonal reductions in revenues or earnings of such Party in the ordinary course of its business; (iii) any adverse change, event or effect on the global, national or regional energy industry as a whole, including any such change to energy prices or the value of oil and gas assets and properties or other commodities, goods or services, or the availability or costs of hedges; (c) national or international political conditions, including any engagement in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack; (d) changes in GAAP or the interpretation thereof; (e) the entry into or announcement of this Agreement, actions contemplated by this Agreement or the consummation of the transactions contemplated hereby; (f) changes or developments in financial or securities markets or the economy in general; (g) the outbreak or continuation of any disease or epidemic (including COVID-19); (h) natural declines in well performance or reclassification or recalculation of reserves in the ordinary course of business consistent with ordinary, prudent and customary practices in the oil and natural gas exploration and production industry; (i) acts or failures to act of any Governmental Authorities and changes in Law or the interpretation thereof from and after the Execution Date; (j) effects of weather, meteorological events, natural disasters or other acts of God; or (k) any title defects, title losses or encumbrances on title; provided that the exclusions set forth in subsections (b), (d), (f), (g) and (j) shall be effective only insofar as such exclusions do not disproportionately affect the assets and properties of the affected Party as compared to similarly situated oil and gas assets or similarly situated owners of oil and gas assets.

- "Material Contracts" means any of the following Contracts to which any Seller is a party:
- (a) Contracts restricting in any material respect the owner of the Assets from freely engaging in any business or competing anywhere;
- (b) each Contract that burdens any of the Assets and is an indenture, mortgage, loan, credit agreement, sale-leaseback, guaranty of any obligation, bond, letter of credit, or similar financial Contract (other than Permitted Encumbrances) that will not be terminated at or before the Closing;
- (c) any Contract that burdens any of the Assets, covering any futures, hedges, swap, collar, put, call, floor, cap, option or the like that is intended to reduce or eliminate the fluctuations in the prices of commodities, currency, exchange rates or interest rates that will not be terminated at or before the Closing;
  - (d) any Contract the primary purpose of which is to indemnify another Person;
  - (e) Contracts to sell, exchange or otherwise dispose of all or any part of the Assets;
- (f) Contracts involving obligations of, or payments to or from, the owner of the Assets that, in each case, can reasonably be expected to result in aggregate payments by Sellers in excess of Fifty Thousand Dollars (\$50,000.00) in the current or any subsequent fiscal year; and
- (g) Contracts between any Seller and any Affiliate of such Seller that will be binding on the Assets after the Closing.
- "Merrimac" has the meaning set forth in the introductory paragraph of this Agreement.
- "Mineral Interests" has the meaning set forth in the definition of "Assets."
- "Minerals" has the meaning set forth in the definition of "Assets."
- "Net Mineral Acres" means with regard to an Interest, (a) the number of gross surface acres included in such Interest, <u>multiplied</u> by (b) Sellers' percentage ownership interest in the oil and gas fee mineral interests beneath such gross surface acres; <u>provided</u>, <u>however</u>, that if Sellers' Net Mineral Acres vary for different formations or areas within a Tract, a separate calculation shall be performed with respect to each such formation or area for the purposes of calculating Net Mineral Acres under this Agreement.
- "Net Royalty Acres" means, with regard to the Interest, the number equal to (a) the number of Net Mineral Acres attributable to such Interest <u>multiplied</u> by (b) the applicable Royalty Rate for such Interest <u>multiplied</u> by (c) eight (8); <u>provided, however</u>, that if an Interest varies as to different target formations or areas within a Tract, a separate calculation shall be performed with respect to each such formation or area for the purposes of calculating Net Royalty Acres under this Agreement. For any unleased tracts, a twelve and one-half percent (12.5%) Royalty Rate shall be used as the applicable Royalty Rate.
  - "Nonparty Affiliates" has the meaning set forth in Section 11.18.
  - "**Obligations**" has the meaning set forth in <u>Section 10.1</u>.
  - "Officer Certificates" has the meaning set forth in Section 8.2(c).
  - "Overriding Royalty Interests" has the meaning set forth in the definition of "Assets."

"Parties" has the meaning set forth in the introductory paragraph of this Agreement.

"Party" has the meaning set forth in the introductory paragraph of this Agreement.

#### "Permitted Encumbrances" means:

- (a) the terms and provisions of any Contract or Lease of record to the extent that such terms and provisions do not, individually and in the aggregate, operate to reduce Sellers' Net Royalty Acres in any Tract below the Stated NRA for such Tract;
- (b) lessors' royalties, non-participating royalties and similar burdens of record that do not, individually or in the aggregate, operate to reduce Sellers' Net Royalty Acres in any Tract below the Stated NRA for such Tract:
- (c) preferential rights to purchase and required third party consents to assignment of any Asset that are not applicable to the transactions contemplated by this Agreement;
- (d) rights to consent by, required notices to, filings with or other actions by Governmental Authorities or any other Person in connection with the sale or conveyance of any of the Assets if the same are customarily obtained in the ordinary course of business subsequent to such sale or conveyance;
- (e) any materialman's, mechanics', repairman's, employees', contractors', operators' liens or other similar liens or charges for liquidated amounts arising in the ordinary course of business for amounts not yet delinquent or, if delinquent, that are being contested in good faith, or which will be released prior to Closing;
- (f) any liens for Taxes and assessments not yet delinquent or, if delinquent, that are being contested in good faith in the ordinary course of business and for which Sellers have agreed to pay under the terms of this Agreement or which have been prorated under the terms of this Agreement;
- (g) all rights reserved to or vested in any Governmental Authority to control or regulate the Assets in any manner;
  - (h) the lack of executive rights in any of the Lands;
- (i) defects or irregularities of title as to which the relevant statute of limitation or prescription would bar any attack or claim against Sellers' title;
- (j) defects in the chain of title arising from the failure to recite marital status, omissions of successors or heirship, or the lack of probate proceedings and defects arising out of lack of corporate or other entity authorization, absent reasonable evidence that the defect results in a third party's actual and valid claim of title to the affected Interest:
- (k) any defects based solely on (i) lack of information, including lack of information in Sellers' files, the lack of third party records or unavailability of information from applicable Governmental Authorities, absent reasonable evidence that the defect results in a third party's actual and valid claim of title to the affected Interest or (ii) the absence of certain documents in Sellers' files that are referenced by other documents that are located in Sellers' files;

- (l) division orders and sales contracts terminable without penalty upon no more than ninety (90) days' notice to the purchaser;
- (m) defects or irregularities arising out of lack of corporate authorization or a variation in corporate name, absent affirmative evidence that such corporate action was not authorized and results in another Person's superior claim of title to the relevant Assets;
  - (n) any matter waived in writing by Buyer;
- (o) defects arising out of a scrivener's error that would not reasonably be expected to adversely affect title to the Assets;
- (p) defects based on a gap in the chain of title of the Asset if such gap exists prior to January 1, 1990, unless such gap is shown to exist in the county or parish records by an abstract of title, title opinion or landman's title chain or runsheet;
- (q) defects resulting from lack of survey, unless a survey is expressly required by the underlying lease or applicable Law; and
- (r) defects based on failure to record releases of liens, production payments or mortgages that have expired by their own terms or the enforcement of which are barred by applicable statutes of limitation; in each case, unless Buyer provided evidence that such defects or irregularities may result in another Person's actual and superior claim of title.
- "Person" means any natural person, corporation, partnership, trust, limited liability company, court, agency, government, board, commission, estate or other entity or authority.
  - "Preliminary Settlement Statement" has the meaning set forth in Section 2.3(a).
- "**Proceeding**" means any proceeding, action, arbitration, litigation, subpoena, or suit (whether civil, criminal, administrative, investigative, or informal) commenced, brought, conducted, or heard by or before, or otherwise involving, any Governmental Authority or arbitrator.
  - "Purchase Price" has the meaning set forth in Section 2.1.
  - "OHEP" has the meaning set forth in the introductory paragraph of this Agreement.
- "Reasonable Efforts" means a Party's commercially reasonable efforts in accordance with reasonable commercial practice.
- "Records" means copies of all files, records, and data in any Seller's possession or control to the extent relating to any of the Assets, including deed, mineral and Lease records, assignments, division order records, title records (including abstracts of title, title opinions and memoranda and title curative documents), Contracts, electronic data files (if any), maps, production records, decline curves and graphical production curves and financial, accounting and Asset Tax records, but excluding all Excluded Records.
- "Representatives" means with respect to any Person, its officers, directors, managers, employees, consultants, agents, financial advisors, attorneys, accountants and other representatives.
- "Retained Obligations" means all Obligations arising from or related to the following: (a) the ownership of the Assets prior to the Effective Time, (b) the amount of all Taxes for which Sellers are

responsible pursuant to <u>Article 7</u>, (c) the Excluded Assets, and (d) general and administrative expenses (as such term is understood in accordance with GAAP) of the Sellers.

- "Review Period" has the meaning set forth in Section 2.4(a).
- "Royalty Interests" has the meaning set forth in the definition of "Assets."
- "Royalty Rate" means the applicable royalty interest (a) set forth in the applicable Lease, with respect to any Mineral Interest, minus the proportionate share, if any, of any non-participating royalty interests that burden such Mineral Interest or (b) carved out of the lessee's working interest derived from such Lease, with respect to any Overriding Royalty Interest.
  - "Scheduled Closing Date" has the meaning set forth in Section 9.1.
  - "Securities Act" means the Securities Act of 1933, as amended.
  - "Seller" has the meaning set forth in the introductory paragraph of this Agreement.
  - "Seller Fundamental Representations" has the meaning set forth in <u>Section 10.3</u>.
  - "Seller Indemnitees" has the meaning set forth in Section 10.2(b).
  - "Seller Material Adverse Effect" means a Material Adverse Effect with respect to Sellers.
  - "Seller Officer Certificate" has the meaning set forth in Section 8.2(c).
- "Seller Taxes" shall mean (a) Income Taxes imposed by applicable Law on Sellers and (b) Asset Taxes allocable to Sellers pursuant to <u>Article 7</u> (taking into account, and without duplication of, such Asset Taxes effectively borne by Sellers as a result of (i) the adjustments to the Purchase Price made pursuant to <u>Section 2.3</u>, and (ii) any payments made from one Party to the other in respect of Asset Taxes pursuant to <u>Article 7</u>).
  - "Sellers" has the meaning set forth in the introductory paragraph of this Agreement.
  - "Special Warranty" has the meaning set forth in Section 6.5(c).
  - "Stated NRA" has the meaning set forth in Section 3.2(c)(iii).
  - "Straddle Period" means any taxable period beginning before and ending after the Effective Time.
  - "Subject Schedule Supplement" has the meaning set forth in Section 6.3.
  - "Survival Period" has the meaning set forth in Section 10.3.
- "Tax Returns" means any return, declaration, report, claim for refund, or information return or statement relating to Taxes filed or required to be filed with any Governmental Authority, including any schedules or attachment thereto, and including any amendment thereof.
- "Taxes" means (a) all taxes, assessments, fees and other charges in the nature of a tax imposed by any Governmental Authority, including any federal, state, local or foreign income tax, surtax, remittance tax, presumptive tax, net worth tax, special contribution tax, production tax, severance tax, value added tax, withholding tax, gross receipts tax, profits tax, ad valorem tax, personal property tax, real property tax,

sales tax, goods and services tax, transfer tax, use tax, excise tax, franchise tax, occupation tax, payroll tax, employment tax, unemployment tax, disability tax, alternative or add-on minimum tax and estimated tax, and (b) any interest, fine, penalty or additions to tax imposed by a Governmental Authority in connection with any item described in clause (a).

"Title Claims Date" has the meaning set forth in Section 3.1.

"Title Consultant" has the meaning set forth in Section 3.2(g)(iv).

"**Title Credit**" means any right, circumstance or condition that operates to increase Sellers' Actual NRA in any Tract above the Stated NRA set forth on <u>Exhibit A-1</u> for such Tract.

"Title Credit Amount" has the meaning set forth in Section 3.2(e).

"Title Deductible" has the meaning set forth in Section 3.2(d)(ii).

"Title Defect" means (a) any Encumbrance, defect or other matter that causes (or if not cured, could reasonably be expected to cause) Sellers not to have Defensible Title to any of the Assets; or (b) any default by Sellers under a Lease or Contract or other agreement that (i) has an adverse effect on the operation, value or use of any Asset, (ii) prevents Sellers from receiving the proceeds of production attributable to Sellers' interest in any Asset, or (iii) results in cancellation or impairment of Sellers' interest in any Asset.

"Title Defect Amount" has the meaning set forth in Section 3.2(c).

"Title Dispute" has the meaning set forth in Section 3.2(g)(i).

"Title Instrument" has the meaning set forth in the definition of "Effective Time".

"Title Threshold" has the meaning set forth in Section 3.2(d)(i).

"Tract" has the meaning set forth in Section 2.2.

"Transfer Taxes" means, collectively, transfer, sales, use, excise, documentary, stamp, gross receipts, goods and services, registration or other similar Taxes relating to the transactions contemplated by this Agreement.

"Wells" has the meaning set forth in the definition of "Assets."

# EXHIBIT A-1 - TRACTS AND ALLOCATED VALUES

The following Assets in Caddo Parish, LA:

Tract	Seller	Section	Township-Range	NMA	Royalty	NRA
T-0085559	Merrimac	8	17N-15W	3.52	0.20	5.63
T-0086334	Merrimac	9	17N-15W	1.30	0.20	2.08
T-0085561	Merrimac	8	17N-15W	5.86	0.20	9.38
T-0086361	Merrimac	9	17N-15W	2.16	0.20	3.46
T-0085517	Merrimac	8	17N-15W	3.52	0.20	5.63
T-0086320	Merrimac	9	17N-15W	1.30	0.20	2.08
T-0085365	Merrimac	8	17N-15W	5.86	0.20	9.38
T-0086249	Merrimac	9	17N-15W	2.16	0.20	3.46
T-0082217	Merrimac	4	17N-15W	0.84	0.22	1.47
T-0082198	Merrimac	4	17N-15W	0.84	0.22	1.47
T-0082074	Merrimac	4	17N-15W	0.84	0.22	1.48
T-0082009	Merrimac	4	17N-15W	0.84	0.22	1.48
T-0086357	Merrimac	9	17N-15W	2.83	0.20	4.53
T-0082278	Merrimac	4	17N-15W	22.42	0.22	39.47
T-0082277	Merrimac	4	17N-15W	19.71	0.22	34.69
T-0082275	Merrimac	4	17N-15W	4.93	0.22	8.68
T-0082274	Merrimac	4	17N-15W	4.93	0.22	8.67
T-0082276	Merrimac	4	17N-15W	0.16	0.22	0.28
T-0082273	Merrimac	4	17N-15W	4.93	0.22	8.67
T-0082272	Merrimac	4	17N-15W	4.75	0.22	8.36
T-0082271	Merrimac	4	17N-15W	9.85	0.22	17.34
T-0082270	Merrimac	4	17N-15W	4.93	0.22	8.67
T-0082269	Merrimac	4	17N-15W	4.92	0.22	8.67
T-0086237	Merrimac	9	17N-15W	6.53	0.20	10.45
T-0086236	Merrimac	9	17N-15W	1.42	0.20	2.26
T-0082068	Merrimac	4	17N-15W	4.92	0.22	8.67
T-0082017	Merrimac	4	17N-15W	2.46	0.22	4.33
T-0082050	Merrimac	4	17N-15W	2.46	0.22	4.33
T-0082239	QHEP	4	17N-15W	4.92	0.22	8.67
T-0082233	QHEP	4	17N-15W	4.93	0.22	8.67
T-0082237	QHEP	4	17N-15W	4.93	0.22	8.68
T-0082232	QHEP	4	17N-15W	4.75	0.22	8.36
T-0082231	QHEP	4	17N-15W	19.71	0.22	34.69
T-0082230	QHEP	4	17N-15W	9.85	0.22	17.34
T-0082236	QHEP	4	17N-15W	0.16	0.22	0.28
T-0082238	QHEP	4	17N-15W	4.93	0.22	8.67
T-0082235	QHEP	4	17N-15W	22.42	0.22	39.47
T-0082234	QHEP	4	17N-15W	4.93	0.22	8.67
T-0086398	QHEP	9	17N-15W	6.53	0.20	10.45
T-0086396	QHEP	9	17N-15W	1.42	0.20	2.26
Total				220.68		381.28

API	Wells	Operator	Status
1701734912	MAYFIELD 9 H; HA RA SUY #1	ENSIGHT IV ENERGY MGMT LLC	PDP
1701736213	PINEHILLS 9-4 HC; HA RA SUY #1-ALT	BLUE DOME OPERATING LLC	PDP
1701736434	PINEHILLS 9-4 HC; HA RA SUY #2-ALT	BLUE DOME OPERATING LLC	DUC
1701736435	PINEHILLS 9-4 HC; HA RA SUY #3-ALT	BLUE DOME OPERATING LLC	DUC
1701736436	PINEHILLS 9-4 HC; HA RA SUY #4-ALT	BLUE DOME OPERATING LLC	DUC
1701736437	PINEHILLS 9-4 HC; HA RA SUY #5-ALT	BLUE DOME OPERATING LLC	DUC
1701736329	ROBINSON 8-5 HC; HA RA SU116 #2 ALT	BLUE DOME OPERATING LLC	PDP
1701736330	ROBINSON 8-5 HC; HA RA SU116 #1	BLUE DOME OPERATING LLC	PDP

#### **EXHIBIT B**

#### FORM OF CONVEYANCE

# <u>DEED, ASSIGNMENT AND BILL OF SALE</u> (Special Warranty Deed)

STATE OF LOUISIANA
PARISH OF CADDO

This Deed, Assignment and Bill of Sale (this "*Deed*") is dated as of the acknowledgments below but is effective for all purposes as to each Tract (as defined below) as of the Effective Date of each respective instrument referenced on <a href="Exhibit A-1">Exhibit A-1</a> attached hereto (the "*Effective Time*"), from [SELLER], a Delaware limited liability company ("*Grantor*"), having its principal address at 3811 Turtle Creek Blvd., Suite 1100, Dallas, Texas 75219, to PHX Minerals Inc., an Oklahoma corporation ("*Grantee*") having its principal address at 1601 NW Expressway, Valliance Bank Tower, Suite 1100, Oklahoma City, Oklahoma 73118. Grantor and Grantee are sometimes referred to herein individually as a "*Party*" and collectively as the "*Parties*".

Grantor, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and subject to the terms set forth herein, does by these presents grant, bargain, sell, convey, assign, transfer, set over, and deliver unto Grantee all of Grantor's rights, title, and interest in, to, and under the following (collectively, the "Assets"), except to the extent constituting Excluded Assets, as hereinafter defined:

- (a) all fee mineral interests and lessor royalty interests in all oil, natural gas, coalbed methane and other liquid or gaseous Hydrocarbons, as well as their respective constituent products (including condensate, casinghead gas, distillate and natural gas liquids), and any other minerals, similar or dissimilar, customarily produced in conjunction therewith (all such substances are defined for purposes of this Deed as a "*Mineral*" or the "*Minerals*"), in, on and under or produced and saved from each of the respective tracts of land and properties conveyed to Grantor by the respective instruments (each, individually a "*Tract*") listed and described in <u>Exhibit A-1</u> attached hereto (such lands, Tracts, and properties described in <u>Exhibit A-1</u>, the "*Lands*"), together with any royalty interests attributable thereto and any units, lands, tracts, mineral servitudes, or other properties pooled with any of the Lands (collectively, the "*Mineral Interests*");
- (b) any overriding royalty interests held by Grantor burdening oil, gas or other minerals produced, saved or sold from the Lands, including those described in <u>Exhibit A-1</u> attached hereto (collectively, the "*Overriding Royalty Interests*");
- (c) the non-participating royalty interests burdening the Lands, including those described in <u>Exhibit A-1</u> attached hereto (collectively, the "*Royalty Interests*", together with the Mineral Interests and Overriding Royalty Interests, the "*Interests*");

- (d) any oil, gas and mineral leases that relate to the Interests, including all reversionary rights thereunder, including those described in <u>Exhibit A-1</u> attached hereto (collectively, the "*Leases*");
- (e) all division and transfer orders, and, except for the Leases, all written contracts, contractual rights, interests and other written agreements to which Grantor is a party (including as assignee) and under which any of the Interests are bound (collectively, the "Contracts");
  - (f) all Records, except for the Excluded Records;
- (g) (i) all rights of ingress and egress, at all times for the purpose of exploring, drilling, operating for and producing from the Lands for Minerals and removing the same therefrom, and of laying pipelines, storing oil, building tanks, processing and treating plants and facilities, power stations, telephone lines, roads and other structures necessary to produce, save, care for, treat, store, compress, process, and transport said products, attributable to the conveyed interests, and (ii) all rights with respect to the use and occupation of the surface of the Lands and the subsurface depths under the Lands for the sole purpose of development and maintenance of the Interests; and (iii) the right of ingress and egress across any lands owned or controlled by Grantor to the extent such ingress and egress across such lands is reasonable to obtain access for the development and maintenance of the Interests;
- (h) all proceeds attributable to any of the Interests or the Leases and attributable to the time period after the Effective Time;
- (i) to the extent transferable (*provided* that Grantor will not be required to pay any fees or incur any liabilities in order to transfer such data), all proprietary and non-proprietary geologic, geophysical and seismic data that relate to the Interests (excluding any interpretive data);
- (j) all permits to the extent primarily relating to or applicable to any of the Interests or Leases and (i) required for the ownership of the Interests or Leases and (ii) transferable pursuant to applicable Law;
- (k) to the extent relating to the Assets and to the extent assignable, all rights and interests of either Grantor (i) under any agreement or policy of indemnity or insurance (including any rights, claims or causes of action of either Grantor against third parties under any indemnity or hold harmless agreements) and any indemnities received in connection with Grantor's prior acquisition of any of the Assets, but in each case only to the extent that such rights or interests arise from obligations or liabilities for which Grantee is responsible under this Deed, or (ii) relating to claims and causes of action that may be asserted against a third party to the extent such rights and claims arise from obligations or liabilities assumed by Grantee hereunder; and
- (l) except to the extent pertaining to any Retained Obligations, and only to the extent assignable, all rights, claims and causes of action to the extent, and only to the extent, that such rights, claims or causes of action are associated with any of the other Assets and relate to the period from and after the Effective Time.

SAVING, EXCEPTING AND RESERVING to Grantor, however, all of the following assets (the "*Excluded Assets*"):

- (a) all proceeds attributable to any of the Assets to the extent attributable to the time period before the Effective Time, including any lease bonuses, royalties and other proceeds and benefits to the extent (i) attributable to the time period before the Effective Time and (ii) received no later than one year following the Closing Date;
- (b) all claims of Grantor or any of their Affiliates for refunds of, rights to receive funds from any Governmental Authority, or loss carry forwards or credits with respect to Seller Taxes;
- (c) any refunds of costs or other expenses borne by Grantor or its predecessors in title, insofar as such refunds relate to time periods prior to the Effective Time or to any Retained Obligations;
  - (d) the Excluded Records;
- (e) all claims and causes of action of Grantor relating to (i) any Retained Obligation or (ii) production that occurred prior to the Effective Time, and all audit rights arising under any Contracts with respect to the time periods prior to the Effective Time or any Retained Obligations; *provided, however*, if any third party makes a claim regarding time periods prior to the Effective Time that is or would be an Assumed Liability, then the claims and causes of action against such third party, to the extent relating to such Assumed Liability, that would otherwise be Excluded Assets pursuant to this clause (e), shall become Assets;
- (f) all computer or communications software owned, licensed or used by Grantor, other than the Records;
- (g) any logo, service mark, copyright, trade name or trademark of or associated with Grantor or any Affiliate of Grantor or any business of Grantor or of any Affiliate of Grantor;
  - (h) motor vehicles and other rolling stock owned by Grantor; and
- (i) all oil, gas and Hydrocarbons produced, saved or sold from the Assets with respect to all periods prior to the Effective Time.
- TO HAVE AND TO HOLD all and singular the Assets, together with all rights, titles, interests, estates, remedies, powers and privileges thereto appertaining unto Grantee and its successors, legal representatives, and assigns forever, subject to the following:
- 1. This Deed is subject to and delivered pursuant to that certain Purchase and Sale Agreement dated [EXECUTION DATE], by and among Grantor and Grantee (as may be amended from time to time, the "*Purchase and Sale Agreement*"), and nothing in this Deed shall operate to limit, release, or impair any of Grantor's or Grantee's respective rights, obligations, remedies, or indemnities in the Purchase and Sale Agreement. Capitalized terms used in this Deed shall have the meanings prescribed in this Deed where such capitalized terms are defined; *provided, however*, that capitalized terms used in this Deed and not otherwise defined shall have the meanings given

to such terms in the Purchase and Sale Agreement. Each defined term shall be equally applicable both to the singular and the plural forms of the term so defined. To the extent the terms and provisions of this Deed are in conflict, or inconsistent, with the terms and provisions of the Purchase and Sale Agreement, the terms and provisions of the Purchase and Sale Agreement shall control.

- 2. This Deed shall extend to, be binding upon, and inure to the benefit of the Parties and their respective successors and assigns.
- 3. Subject to the terms of the Purchase and Sale Agreement, from and after the date hereof, Grantee assumes and hereby agrees to pay, perform, fulfill and discharge (or cause to be paid, performed, fulfilled and discharged) the Assumed Liabilities; provided that the Assumed Liabilities shall not include, and Grantor shall retain (and Grantee shall have no obligation with respect to any) the Retained Obligations.
- 4. Grantor, for a period of thirty-six (36) months after the date hereof, hereby warrants and defends Defensible Title to the Assets unto Grantee, its successors and assigns, against whomsoever lawfully claims the same or any part thereof, solely to the extent such claim arises by, through or under such Grantor but not otherwise. Grantee is hereby specifically assigned, and subrogated to, all warranties of title which Grantor may have from predecessors in interest (other than such Grantor or any Affiliate of such Grantor) to the extent applicable with respect to the Assets and to the extent such Grantor or such Affiliates may legally assign such rights and grant such subrogation.
- 5. EXCEPT AS OTHERWISE SET FORTH IN SECTION 3.2 AND ARTICLE 4 OF THE PURCHASE AND SALE AGREEMENT AND EXCEPT AS TO THE SPECIAL WARRANTY OF DEFENSIBLE TITLE UNDER THIS DEED, GRANTEE WILL ACCEPT THE ASSETS AT CLOSING IN THEIR PRESENT CONDITION, "AS IS AND WHERE IS AND WITH ALL FAULTS." GRANTEE ACKNOWLEDGES AND AGREES THAT, EXCEPT AS OTHERWISE SET FORTH IN THIS DEED AND ARTICLE 4 OF THE PURCHASE AND SALE AGREEMENT, GRANTOR HAS MADE NO REPRESENTATIONS OR WARRANTIES OF ANY KIND, EXPRESS OR IMPLIED, WRITTEN, ORAL, OR OTHERWISE, AS TO THE ACCURACY OR COMPLETENESS OF THE BACKGROUND MATERIALS OR ANY OTHER INFORMATION RELATING TO THE ASSETS FURNISHED BY OR ON BEHALF OF GRANTOR OR TO BE FURNISHED TO GRANTEE OR ITS REPRESENTATIVES, INCLUDING GRANTOR'S INTERNAL APPRAISALS AND INTERPRETIVE DATA.
- 6. THIS DEED AND THE LEGAL RELATIONS BETWEEN THE PARTIES SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS, WITHOUT GIVING EFFECT TO PRINCIPLES OF CONFLICTS OF LAWS THAT WOULD RESULT IN THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION, *PROVIDED THAT*, ANY DISPUTE PERTAINING TO REAL PROPERTY SHALL BE GOVERNED BY THE LAWS OF THE STATE OF LOUISIANA.

- SUBJECT TO AND WITHOUT LIMITING SECTION 2.4(C) OF THE PURCHASE AND SALE AGREEMENT, THE PARTIES AGREE THAT ANY AND ALL DISPUTES OR CLAIMS BY ANY PARTY ARISING FROM OR RELATED TO THIS DEED THAT CANNOT BE AMICABLY SETTLED, WILL BE DETERMINED SOLELY AND EXCLUSIVELY BY ARBITRATION IN ACCORDANCE WITH THE FEDERAL ARBITRATION ACT AND USING THE RULES OF THE AMERICAN ARBITRATION ASSOCIATION OR ANY SUCCESSOR THEREOF WHEN NOT IN CONFLICT WITH SUCH ACT; PROVIDED, HOWEVER, THAT NOTHING CONTAINED IN THIS SECTION 7 SHALL LIMIT ANY PARTY'S RIGHT TO BRING (I) POST-ARBITRATION ACTIONS SEEKING TO ENFORCE AN ARBITRATION AWARD OR (II) ACTIONS SEEKING INJUNCTIVE OR OTHER SIMILAR RELIEF UNDER SECTION 11.16 OF THE PURCHASE AND SALE AGREEMENT IN THE EVENT OF A BREACH OR THREATENED BREACH OF ANY OF THE PROVISIONS OF THIS DEED (OR ANY OTHER AGREEMENT CONTEMPLATED HEREBY). IN FURTHERANCE OF THE FOREGOING, THE ARBITRATION PROCEDURES SET FORTH IN SECTIONS 11.3(B)-11.3(C) OF THE PURCHASE AND SALE AGREEMENT ARE INCORPORATED HEREIN BY REFERENCE, MUTATIS MUTANDIS, AS THOUGH SUCH PROCEDURES WERE SET FORTH HEREIN IN FULL.
- TO THE EXTENT APPLICABLE TO THE ASSETS OR ANY PORTION THEREOF, GRANTEE HEREBY WAIVES THE PROVISIONS OF THE LOUISIANA UNFAIR TRADE PRACTICES AND CONSUMER PROTECTION LAW (LA. R.S. 51.1401, ET SEQ.). GRANTEE WARRANTS AND REPRESENTS THAT IT: (A) IS EXPERIENCED AND KNOWLEDGEABLE WITH RESPECT TO THE OIL AND GAS **INDUSTRY** GENERALLY AND WITH TRANSACTIONS OF THE **TYPE** CONTEMPLATED HEREIN AND IN THE PURCHASE AND SALE AGREEMENT SPECIFICALLY; (B) POSSESSES AMPLE KNOWLEDGE, EXPERIENCE AND EXPERTISE TO EVALUATE INDEPENDENTLY THE MERITS AND RISKS OF THE CONTEMPLATED TRANSACTIONS; AND (C) IS NOT IN A SIGNIFICANTLY DISPARATE BARGAINING POSITION.
- 9. The Exhibits to this Deed are hereby incorporated by reference and constitute a part of this Deed.
- 10. The Parties agree to execute, acknowledge and deliver to the other such further instruments and take such other action as may be reasonably requested in order to accomplish more effectively the purposes of the transactions contemplated by this Deed.
- 11. In the event any term of this Deed is found by any court to be void or otherwise unenforceable, the remainder of this Deed shall remain valid and enforceable, and, to the fullest extent permitted by Law, such offending term or terms shall be replaced with an enforceable term or enforceable terms that as nearly as possible effect the Parties' intent.
- 12. This Deed may be executed and delivered in one or more counterparts, each of which when executed and delivered shall be an original, and all of which when executed shall constitute one and the same instrument.

[Signature and Acknowledgment Pages Follow]

IN WITNESS WHEREOF, the Parties have executed this Deed on the date first set forth above, but this Deed shall be effective for all purposes as of the Effective Time.

	<b>GRANTOR:</b>
	[SELLER]
	By:
	Name: Ryan Turner
	Title: Managing Person
	Witness:
	Witness:
STATE OF	<u>ACKNOWLEDGMENT</u>
COUNTY OF	
by Ryan Turner, known to r liability company, who affir	cknowledged before me this day of, 2021, ne to be the Managing Person of [SELLER], a Delaware limited med that the foregoing instrument was signed on behalf of such on of this instrument was the free act and deed of such company.
	Notary Public in and for the State of

	GRANTEE:	
	PHX MINERALS INC.	
	By: Name: Kenna Clapp Title: Director of Land	
	<u>ACKNOWLEDGMENT</u>	
STATE OF	 \{\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\	
oy Kenna Clapp, known corporation, who affirme	was acknowledged before me this day of to me to be the Director of Land of PHX Minerals Inc., and that the foregoing instrument was signed on behalf of statistics instrument was the free act and deed of such company.	an Oklahoma uch company
	Notary Public in and for the State of	<del></del>

# **EXHIBIT A-1**

# **LANDS**

The following described Tracts, insofar and only insofar as the interest acquired by Grantor by virtue of the respective instrument listed below:

[To be updated pursuant to Article 3]

	Effective Date:
	Recorded:
	Legal Description:
2.	Grantor:
	Grantee:
	Effective Date:
	Recorded:
	Legal Description:
3.	Grantor:
	Grantee:
	Effective Date:
	Recorded:
	Legal Description:
4.	Grantor:
	Grantee:
	Effective Date:
	Recorded:
	Legal Description:
5.	Grantor:
	Grantee:
	Effective Date:
	Recorded:
	Legal Description:

1. Grantor: Grantee:

Including without limitation Grantor's interest in the following described wells, insofar and only insofar as attributable to the Tracts:

[To be updated pursuant to Article 3]

API	Well Name	Operator
1701734912	MAYFIELD 9 H; HA RA SUY #1	ENSIGHT IV ENERGY MGMT LLC
1701736213	PINEHILLS 9-4 HC; HA RA SUY #1-ALT	BLUE DOME OPERATING LLC
1701736434	PINEHILLS 9-4 HC; HA RA SUY #2-ALT	BLUE DOME OPERATING LLC
1701736435	PINEHILLS 9-4 HC; HA RA SUY #3-ALT	BLUE DOME OPERATING LLC
1701736436	PINEHILLS 9-4 HC; HA RA SUY #4-ALT	BLUE DOME OPERATING LLC
1701736437	PINEHILLS 9-4 HC; HA RA SUY #5-ALT	BLUE DOME OPERATING LLC
1701736329	ROBINSON 8-5 HC; HA RA SU116 #2 ALT	BLUE DOME OPERATING LLC
1701736330	ROBINSON 8-5 HC; HA RA SU116 #1	BLUE DOME OPERATING LLC

# PURCHASE AND SALE AGREEMENT BY AND BETWEEN PALMETTO INVESTMENT PARTNERS II, LLC,

(as Seller)

**AND** 

PHX MINERALS INC.

(as Buyer)

Dated as of: December 6, 2021

# **TABLE OF CONTENTS**

Article	1 PURCHASE AND SALE	
1.1	Purchase and Sale	4
1.2	Effective Time.	4
1.3	Excluded Assets.	4
Article	2 PURCHASE PRICE; REVENUES AND EXPENSES	4
2.1	Purchase Price.	4
2.2	Allocation of Purchase Price.	4
2.3	Adjustments to the Purchase Price at Closing.	4
2.4	Adjustments to the Purchase Price Following Closing.	6
2.5	Payment Method.	7
2.6	Principles of Accounting.	7
Article	3 SCOPE OF DUE DILIGENCE AND TITLE DEFECTS	7
3.1	Scope of Due Diligence Review and Access.	7
3.2	Title Defects and Purchase Price Adjustment.	7
Article	4 SELLER'S REPRESENTATIONS AND WARRANTIES	13
4.1	Organization	13
4.2	Qualification	13
4.3	Power; No Conflict.	13
4.4	Authorization and Enforceability.	13
4.5	Liability for Brokers' Fees.	13
4.6	Foreign Person.	13
4.7	No Bankruptcy.	13
4.8	Litigation.	13
4.9	Taxes.	14
4.10	Consents; Preferential Rights; Non-Competes.	14
4.11	Contracts and Leases.	14
4.12	Payment of Royalties; No Suspense Accounts; No Deferred Payments	14
4.13	Environmental Matters.	15
4.14	No Cost-Bearing Interests.	15
4.15	Condemnation.	15
Article	5 BUYER'S REPRESENTATIONS AND WARRANTIES	15
5.1	Organization.	15
5.2	Qualification	15
5.3	Power.	15
5.4	Authorization and Enforceability.	16

5.5	Liability for Brokers' Fees.	16
5.6	Litigation.	16
5.7	Knowledgeable Investor	16
5.8	No Bankruptcy.	16
Article	6 COVENANTS	16
6.1	Pre-Closing Covenants.	16
6.2	Reasonable Efforts.	17
6.3	Amendment of Disclosure Schedules.	17
6.4	Limitations on Representations and Warranties.	17
6.5	Post-Closing Obligations.	18
6.6	Exclusivity	18
Article	7 TAX MATTERS	19
7.1	Transfer Taxes	19
7.2	Asset Taxes.	19
7.3	Tax Allocation.	20
7.4	Refunds of Taxes	20
7.5	Cooperation.	20
Article	8 CONDITIONS PRECEDENT AND TERMINATION	20
8.1	Conditions to Obligations of Seller.	20
8.2	Conditions to Obligations of Buyer.	21
8.3	Frustration of Closing Conditions.	22
8.4	Termination.	22
8.5	Effect of Termination and Remedies.	23
Article	9 CLOSING	23
9.1	Date and Place of Closing.	23
9.2	Closing Obligations.	24
	10 ASSUMPTION AND RETENTION OF OBLIGATIONS AND MNIFICATION; DISCLAIMERS	24
	Buyer's Assumption of Liabilities and Obligations.	
	Indemnification.	
10.3	Survival of Warranties, Representations and Covenants.	26
	Reservation as to Non-Parties.	
10.5	Acceptance of Title Condition.	27
10.6	Express Negligence Rule.	27
	Waiver of Certain Damages.	
	Redhibition Waiver	28

10.9 UTPC	PL Waiver.	28
10.10 Deterr	nination of Losses.	28
10.11 Notice	and Defense.	28
Article 11 MIS	SCELLANEOUS	29
11.1 Expen	ses	29
11.2 Notice	es	29
11.3 Gover	ning Law; Jurisdiction; Venue: Jury Waiver; Binding Arbitration	30
11.4 Furthe	er Assurances.	31
11.5 Amen	dments: Waivers.	32
11.6 Assign	nment	32
11.7 Count	erparts/Fax Signatures.	32
11.8 Constr	ruction	32
11.9 Entire	Agreement.	32
11.10 Succes	ssors and Permitted Assigns.	33
11.11 Parties	s in Interest.	33
11.12 Severa	ability.	33
11.13 Joint F	Preparation	33
11.14 Public	ity	33
11.15 Disclo	sure Schedules.	33
11.16 Specif	ic Performance.	34
11.17 Confid	lentiality	34
11.18 No Re	course	34
ANNEY DE	ENIED TERMO	
ANNEX – DE	FINED TERMS	A
List of Exhibits	<u>s</u>	
Exhibit A-1	Tracts and Allocated Values	
Exhibit B	Form of Conveyance	
Exhibit C Exhibit D	Form of Certificate of Non-Foreign Status Leases	
List of Disclos	ure Schedules	
Section 4.8	Litigation	
Section 4.9 Section 4.11	Taxes Contracts	
Section 4.11 Section 4.13	Environmental Matters	
Section 4.14	No Cost-Bearing Interests	

#### PURCHASE AND SALE AGREEMENT

This Purchase and Sale Agreement (the "Agreement") is dated as of this 6th day of December, 2021 (the "Execution Date"), and is by and between Palmetto Investment Partners II, LLC, a Delaware limited liability company ("Seller"), and PHX Minerals Inc., an Oklahoma corporation ("Buyer"). Seller and Buyer are each individually referred to in this Agreement as a "Party" and collectively as the "Parties."

#### RECITALS

WHEREAS, Seller owns certain rights, title and interests in and to certain mineral interests, royalty interests and overriding royalty interests in the oil, gas, and other minerals underlying certain lands located in Caddo Parish, Louisiana; and

WHEREAS, Seller desires to sell such rights, title and interests to Buyer, and Buyer desires to purchase such rights, titles and interests from Seller on the terms and subject to the conditions set forth in this Agreement.

NOW THEREFORE, in consideration of the mutual promises, representations, warranties, covenants and agreements contained in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Buyer and Seller hereby agree as follows:

# Article 1 PURCHASE AND SALE

- 1.1 Purchase and Sale. Subject to the terms and conditions of this Agreement, at the Closing, Buyer agrees to purchase from Seller, and Seller agrees to sell, assign, transfer, convey and deliver to Buyer, all of such Seller's right, title and interest in and to the Assets for the Purchase Price specified in Article 2.
- **1.2 Effective Time.** If the Closing occurs, the purchase and sale of the Assets shall be effective as of the date Seller was vested with its interest in the respective Assets (the "Effective Time").
- 1.3 Excluded Assets. Notwithstanding any other provision of this Agreement to the contrary, the Assets to be conveyed and assigned under this Agreement do not include the Excluded Assets, all of which are reserved by Seller.

# Article 2 PURCHASE PRICE; REVENUES AND EXPENSES

- **2.1 Purchase Price.** Subject to the adjustments set forth below and the other terms and conditions of this Agreement, the total consideration for the purchase, sale and conveyance of the Assets to be paid by Buyer to Seller is Six Hundred One Thousand Seven Hundred Ninety-Six Dollars and Seventy-Four Cents (\$601,796.74) (the "**Purchase Price**").
- 2.2 Allocation of Purchase Price. The unadjusted Purchase Price shall be allocated among each tract of Land identified on Exhibit A-1 (each, a "Tract"), with each such Tract having the "Allocated Value" set forth for such Tract on Exhibit A-1.

#### 2.3 Adjustments to the Purchase Price at Closing.

(a) At Closing, without limiting Section 2.4, the Purchase Price will be adjusted in accordance with this Section 2.3. No later than three (3) Business Days prior to the Closing Date, Seller will in good faith prepare (as a reasonable and prudent Person) and provide to Buyer a preliminary settlement

statement (the "Preliminary Settlement Statement") identifying all adjustments to the Purchase Price, without duplication, known by Seller to be made at Closing, in accordance with this Agreement and as of the time of the preparation of the Preliminary Settlement Statement (including reasonable estimates of any such amounts where actual amounts are not available). After the delivery of the Preliminary Settlement Statement to Buyer and prior to the Closing, Seller shall consult in good faith with Buyer regarding any reasonable changes to the Preliminary Settlement Statement proposed by Buyer prior to the Closing. Without limiting Section 2.4, the Preliminary Settlement Statement, as agreed upon in writing by Seller and Buyer, will be used to adjust the Purchase Price at Closing; provided, however, that if Seller and Buyer cannot agree in writing on the Preliminary Settlement Statement prior to Closing, then the Preliminary Settlement Statement proposed by Seller, taking into account any adjustments thereto agreed to by the Parties, will be used to adjust the Purchase Price at Closing. Seller and Buyer acknowledge that some items in the Preliminary Settlement Statement may be estimates or otherwise subject to change in the Final Settlement Statement.

# (b) The Purchase Price will be increased (without duplication) by the following:

- (i) the amount of any proceeds (including any lease bonuses, royalties and other proceeds) received by Buyer (and not otherwise paid or remitted to Seller) to the extent attributable to (A) any of the Assets and (B) the time period before the Effective Time:
- (ii) subject to <u>Section 3.2</u>, the Title Credit Amount of any Title Credit (if applicable);
- (iii) the amount of all Asset Taxes allocable to Buyer pursuant to <u>Article 7</u> but paid or otherwise economically borne by Seller; and
  - (iv) as otherwise agreed in writing by the Parties.

### (c) The Purchase Price will be decreased (without duplication) by the following:

- (i) the amount of any proceeds (including any lease bonuses, royalties and other proceeds) received by Seller (and not otherwise paid or remitted to Buyer) to the extent attributable to (A) any of the Assets and (B) the period beginning on and following the Effective Time;
- (ii) subject to <u>Section 3.2</u>, the Title Defect Amount of any uncured Title Defect that is not waived in writing by Buyer (if applicable);
- (iii) the Allocated Value of the Assets excluded from the transactions contemplated hereby pursuant to <u>Section 3.2(f)</u>;
- (iv) the amount of all Asset Taxes allocable to Seller pursuant to <u>Article 7</u> but paid or otherwise economically borne by Buyer; and
  - (v) as otherwise agreed in writing by the Parties.

# 2.4 Adjustments to the Purchase Price Following Closing.

(a) <u>Final Settlement Statement</u>. Within ninety (90) days after the Closing Date (such 90th day, the "**Final Settlement Date**"), Seller will prepare a proposed final settlement statement (the

"Final Settlement Statement"), which shall take into account all final adjustments made to the Purchase Price and shows the resulting final adjusted Purchase Price ("Final Price"). The Final Settlement Statement shall set forth the actual adjustments required under Section 2.3. As soon as practicable, and in any event within thirty (30) days, after receipt of the Final Settlement Statement (the "Review Period"), Buyer shall deliver to Seller a written report containing any proposed changes to the Final Settlement Statement and an explanation of any such changes and the reasons therefor (the "Dispute Notice"). If Buyer fails to timely deliver a Dispute Notice to Seller containing changes Buyer proposes to be made to the Final Settlement Statement, the Final Settlement Statement as delivered by Seller will be deemed to be correct and mutually agreed upon by the Parties, and will, without limiting Section 7.2(c), be final and binding on the Parties and not subject to further audit or arbitration. If the Final Price set forth in the Final Settlement Statement is mutually agreed upon by Seller and Buyer, the Final Settlement Statement and the Final Price shall, without limiting Section 7.2(c), be final and binding on the Parties hereto. Any disputed items included in a Dispute Notice regarding the Final Settlement Statement shall be resolved as provided in Section 2.4(c).

- (b) Payment of Post-Closing Adjustments. Any adjustments to the Purchase Price as reflected in the Final Settlement Statement will be offset against each other so that only one payment is required between the Parties. The Party owing payment will pay the other Party in cash the net post-Closing adjustment to the Purchase Price set forth in the Final Settlement Statement within five (5) Business Days following the date upon which all matters in the proposed Final Settlement Statement (excluding any Title Disputes) are either (i) agreed upon by the Seller and Buyer in writing, (ii) deemed accepted by Buyer at the end of the Review Period pursuant to Section 2.4(a), or (iii) finally determined by the Accounting Consultant in accordance with Section 2.4(c), as applicable.
- Resolution of Disputed Items. After the completion and delivery of the proposed Final Settlement Statement, the Parties shall negotiate in good faith to attempt to reach an agreement in writing on the amount due with respect to any disputed items included in the Dispute Notice. If the Parties agree in writing on the amount due with respect to any disputed items in the proposed Final Settlement Statement, and any payment adjustment is required, the Party owing payment will pay the other Party in the time period provided in Section 2.4(b). If the Parties are unable to agree in writing on the amount due with respect to any disputed items in the proposed Final Settlement Statement within thirty (30) days after Seller receives Buyer's written Dispute Notice, then Buyer and Seller shall engage an independent national accounting firm mutually agreed upon by the Parties (the "Accounting Consultant") to exclusively and finally resolve any such disputed matters set forth in the Dispute Notice in accordance with the terms of this Agreement. Each of Buyer and Seller shall within fifteen (15) Business Days after either Party's written request for arbitration summarize its position with regard to such dispute in a written document and submit such summaries to the Accounting Consultant, together with a copy of the Dispute Notice, the Final Settlement Statement, this Agreement and any other documentation such Party may desire to submit. If there is more than one or more Dispute Notice or disputed matter, all such disputes shall be consolidated into the same arbitration process pursuant to this Section 2.4(c). The Parties shall cooperate diligently with any reasonable request of the Accounting Consultant in an effort to resolve the matters submitted to the Accounting Consultant as soon as reasonably possible after the Accounting Consultant is engaged. The Accounting Consultant, once appointed, shall have no ex parte communications with any of the Parties concerning the determination required under this Section 2.4(c). All communications between any Party or its Affiliates and the Accounting Consultant shall be conducted in writing, with copies sent simultaneously to the other Parties in the same manner, or at a meeting or conference call to which each of the Parties and their respective Representatives have been invited and of which such Parties have been provided at least five (5) days' written notice. In no event shall the Accounting Consultant select an amount (i) for any upward adjustment to the Purchase Price that is greater than Seller's proposed amount or less than Buyer's proposed amount or (ii) for any downward adjustment to the Purchase Price that is greater than Buyer's proposed amount or less than Seller's proposed amount. Within thirty (30) days after receipt of such materials from the Parties and after receipt of any additional information required by the Accounting

Consultant, the Accounting Consultant shall make its determination, which shall be in accordance with the terms of this Agreement and shall be final and binding upon the Parties, without right of appeal, absent Fraud or manifest error. The Accounting Consultant may not award damages, interest or penalties to either Buyer or Seller with respect to any matter. Seller and Buyer shall each bear their own respective legal fees and other costs of presenting its case. Seller shall bear fifty percent (50%) and Buyer shall bear fifty percent (50%) of the costs and expenses of the Accounting Consultant.

- (d) <u>Further Revenues and Expenses</u>. From and after the Closing Date, if a Party or Parties receive revenues that belong to another Party under this Agreement, the Party or Parties receiving the revenues agrees to remit those revenues to the applicable Party promptly (but no later than thirty (30) days) following such first Party's receipt thereof.
- **2.5 Payment Method.** Unless the Parties otherwise agree in writing, all payments made or otherwise required under this Agreement (a) to Seller, will be made in United States dollars by wire transfer in immediately available funds to such account(s) as are designated in writing by the Seller to Buyer and (b) to Buyer, will be made in United States dollars by wire transfer in immediately available funds to such account(s) as are designated in writing by Buyer to Seller.
- **2.6 Principles of Accounting.** The Preliminary Settlement Statement and Final Settlement Statement will be prepared in accordance with GAAP, consistently applied in the petroleum industry, and applicable Laws.

# Article 3 SCOPE OF DUE DILIGENCE AND TITLE DEFECTS

3.1 Scope of Due Diligence Review and Access. In order to allow Buyer to conduct title due diligence with respect to the Assets, from and after the Execution Date until 5:00 P.M. Central Time on December 8, 2021 (the "Title Claims Date"), Seller shall provide Buyer with reasonable access to the Records; provided, however, (a) such access does not unreasonably interfere with the normal operations and business of Seller or its Affiliates, (b) such access shall occur in such a manner as Seller reasonably determine to be appropriate to protect the confidentiality of the transactions contemplated by this Agreement, and (c) nothing herein shall require Seller to provide access, or to disclose any information, to Buyer if such access or disclosure (i) would waive any legal privilege (except with respect to any title opinions covering any Asset), (ii) would be in violation of applicable Laws or regulations of any Governmental Authority, (y) would be in violation of any confidentiality obligations to a third party or (iii) would constitute access to the Excluded Records. Further, Buyer shall have no right (A) to access any of the Assets or (B) to perform or conduct any environmental sampling or other invasive environmental investigation on or about or with respect to any of the Assets. Buyer shall keep confidential all information made available to Buyer by or on behalf of Seller or its Representatives until the Closing.

#### 3.2 Title Defects and Purchase Price Adjustment.

(a "**Defect Notice.** Buyer shall give Seller written notice of alleged Title Defects (a "**Defect Allegation**") promptly upon Buyer's discovery thereof, and in any event, prior to the Title Claims Date. To be effective, such Defect Allegation shall be in writing, delivered to Seller before the Title Claims Date and shall include (i) a reasonably detailed description of the alleged Title Defect, including Buyer's detailed findings and reasons for concluding that such alleged Title Defect exists, (ii) the Allocated Value of the Tract affected by the Title Defect, (iii) the amount by which Buyer reasonably believes that the Allocated Value of the affected Tract is reduced by the alleged Title Defect and the computations and information upon which Buyer's belief is based, and (iv) all supporting information and documents in Buyer's possession or reasonable control relating to such asserted Title Defect necessary for Seller to verify

such Title Defect and the value thereof; provided, however, that an alleged failure to comply with subsections (i) through (iv) above shall not cause any such Defect Allegation to be invalid or any Title Defect to be waived so long as the Defect Allegation is timely delivered and it provides reasonably sufficient notice and supporting documents to Seller of the existence of and reasonable details regarding the nature of the alleged Title Defect. Except for any claim under the Special Warranty, Buyer shall be deemed for any and all purposes to have waived, and Seller shall have no liability for, any Title Defects and other defects of title of which Seller have not received a Defect Allegation from Buyer on or before the Title Claims Date meeting the requirements of this Section 3.2(a). Notwithstanding the foregoing, Buyer shall be entitled to modify or amend any Defect Allegation or other communication until the Title Claims Date, and Buyer shall not be deemed to have waived any Title Defects that are ultimately asserted in a Defect Allegation from Buyer on or before the Title Claims Date meeting the requirements of this Section 3.2(a). Upon Seller's receipt on or prior to the Title Claims Date of any Defect Allegation, Seller shall review such Defect Allegation in good faith and determine whether Seller believes such Defect Allegation meets the requirements of this Section 3.2(a). If Seller determines that any Defect Allegation fails to meet the requirements of this Section 3.2(a), Seller shall provide written notice to Buyer of such determination (and reasonable detail of the basis for such determination) (a "Deficiency Notice") as soon as reasonably practicable following Seller's receipt of such Defect Allegation. Notwithstanding anything to the contrary, with respect to each Defect Allegation received by Seller on or prior to the Title Claim Date, Seller may not dispute any alleged Title Defect set forth in any such Defect Allegation on the basis that it did not meet the requirements of this Section 3.2(a) unless Seller provides prompt written notice to Buyer (no later than five (5) Business Days following the Title Claim Date) of the basis upon which Seller (acting in good faith) has determined that the Defect Allegation failed to meet the requirements of this Section 3.2(a).

Title Credit Notice. Seller or Buyer, as the case may be, as a "Disclosing Credit Party," may (but are not obligated to), give the other Party written notice of alleged Title Credits (a "Credit Notice") promptly upon the Disclosing Credit Party's discovery thereof, and in any event, prior to the Title Claims Date. To be effective, such Credit Notice provided by a Disclosing Credit Party shall be in writing, delivered to the other Party prior to the Title Claims Date, and shall include (i) a reasonably detailed description of the alleged Title Credit, including the Disclosing Credit Party's detailed findings and reasons for concluding that such alleged Title Credit exists, (ii) the Allocated Value of the Tract or affected by the Title Credit, (iii) the amount by which the Disclosing Credit Party reasonably believes that the Allocated Value of the affected Tract is increased by the alleged Title Credit and the computations and information upon which the Disclosing Credit Party's belief is based, and (iv) all supporting information and documents in the Disclosing Credit Party's possession or reasonable control relating to such asserted Title Credit necessary for the other Party to verify such Title Credit and the value thereof; provided, however, that an alleged failure to comply with subsections (i) through (iv) above shall not cause any such Credit Notice to be invalid or any Title Credit to be waived so long as the Credit Notice is timely delivered and it provides reasonably sufficient notice and supporting documents to Seller of the existence of and reasonable details regarding the nature of the alleged Title Credit. Seller shall be deemed for any and all purposes to have waived, and Buyer shall have no liability for, all Title Credits (A) of which Buyer has not received a Credit Notice from Seller, or (B) for which Buyer has failed to provide a Credit Notice to Seller, on or before the Title Claims Date meeting the requirements of this Section 3.2(b). Prior to the Title Claims Date, the Party receiving a Credit Notice shall notify the Disclosing Credit Party if any written alleged Title Credits received from the Disclosing Credit Party on or before the Title Claims Date do not comply with the provisions of this Section 3.2(b). Upon a Party's receipt on or prior to the Title Claims Date of any Credit Notice, such Party shall review such Credit Notice in good faith and determine whether it believes such Credit Notice meets the requirements of this Section 3.2(b). If a Party determines that any Credit Notice fails to meet the requirements of this Section 3.2(b), such Party shall provide written notice to the Disclosing Credit Party of such determination (and reasonable detail of the basis for such determination) (a "Credit Deficiency Notice") as soon as reasonably practicable following such Party's receipt of the Credit Notice. Notwithstanding anything to the contrary, with respect to each Credit Notice received by Buyer on or prior

to the Title Claim Date, Buyer may not dispute any alleged Title Credit set forth in any such Credit Notice on the basis that it did not meet the requirements of this <u>Section 3.2(b)</u> unless it provides prompt written notice to Seller (no later than five (5) Business Days following the Title Claim Date) of the basis upon which Buyer (acting in good faith) has determined that the Credit Notice failed to meet the requirements of this Section 3.2(b).

- (c) <u>Title Defect Amounts.</u> The value of an uncured or unwaived Title Defect (each, a "**Title Defect Amount**"), and any adjustments to the Purchase Price for the same, shall be calculated as follows, without duplication:
  - (i) if Buyer and Seller agree in writing on the Title Defect Amount, that amount shall be the Title Defect Amount:
  - (ii) if a Title Defect is an Encumbrance upon an Asset which is liquidated in amount, then the adjustment shall be the sum necessary to be paid to the obligee to remove the Title Defect from the affected Asset:
  - (iii) if a Title Defect is in respect of a discrepancy between (A) Seller's Actual NRA for a Tract and (B) the aggregate Net Royalty Acres set forth on Exhibit A-1 for such Tract (the "Stated NRA" for such Tract), then the Title Defect Amount shall be the product of (1) the Allocated Value for such Tract, multiplied by (2) a fraction, the numerator of which is the absolute value of difference between the Stated NRA and the Actual NRA, and the denominator of which is the Stated NRA;
  - (iv) if the Title Defect represents an obligation or Encumbrance upon or other defect in title to the affected Tract or Well of a type not described in Section 3.2(c)(i) through Section 3.2(c)(iii), above, then the Title Defect Amount shall be determined by taking into account the Allocated Value of the affected Tract, the legal effect of the Title Defect, the potential economic effect of the Title Defect over the life of the affected Tract or Well, the values placed upon the Title Defect by Buyer and Seller and such other reasonable factors as are necessary to make a proper evaluation; and
  - (v) in no case shall the value of all Title Defects asserted with regard to the Assets included in a Tract exceed the Allocated Value of the affected Tract. The Title Defect Amount with respect to a Title Defect shall be determined without any duplication of any costs included in any other Title Defect Amount hereunder, or for which Buyer otherwise receives credit in the calculation of the Purchase Price.

#### (d) Certain Limitations on Adjustments.

- (i) Buyer shall have no remedy for any individual Title Defect, and Seller shall have no remedy for any individual Title Credit, unless the Title Defect Amount for such individual Title Defect, or Title Credit Amount for such individual Title Credit, exceeds Five Thousand Dollars (\$5,000.00) (the "Title Threshold"), and no Title Defect or Title Credit shall be taken into consideration for the purpose of calculating any Purchase Price reduction or increase under this Section 3.2 unless the Title Defect Amount for such Title Defect, or Title Credit Amount for such Title Credit, exceeds the Title Threshold (which shall be treated as a threshold and not a deductible).
- (ii) Buyer shall have no remedy for any Title Defects unless and until the sum of (A) the aggregate Title Defect Amounts for Title Defects with Title Defect Amounts

exceeding the Title Threshold minus (B) the aggregate Title Credit Amounts for Title Credits with Title Credit Amounts exceeding the Title Threshold, exceeds Fifty Thousand Dollars (\$50,000.00) ("Title Deductible"). Seller shall not be entitled to an upward Purchase Price adjustment for any Title Credits unless and until the sum of (1) the aggregate Title Credit Amounts for Title Credits with Title Credit Amounts exceeding the Title Threshold, minus (2) the aggregate Title Defect Amounts for Title Defects with Title Defect Amounts exceeding the Title Threshold, exceeds the Title Deductible. If, and only if, the Purchase Price reduction or increase which would result from all Title Defect Amounts or Title Credit Amounts, as the case may be, that exceed the Title Threshold exceeds the Title Deductible, then the Purchase Price shall be reduced by the aggregate of all Title Defect Amounts or increased by the aggregate of all Title Credit Amounts. For the avoidance of doubt, Buyer or Seller, as the case may be, shall be entitled to adjust the Purchase Price under this Article 3 from the first dollar of any Title Defects or Title Credits, as the case may be, from and after such time as the aggregate amount of all Title Defects or Title Credits (as the case may be) under this Article 3 equals or exceeds the Title Deductible.

- (iii) The Title Threshold and Title Deductible shall not apply to Title Defects arising by, through or under any Seller or any of its Affiliates or any breach of the Special Warranty.
- (e) <u>Title Credit Amounts</u>. The value of a Title Credit (a "**Title Credit Amount**") shall be calculated as follows:
  - (i) if Buyer and Seller agree in writing on the Title Credit Amount, that amount shall be the Title Credit Amount;
  - (ii) if a Title Credit results from the Actual NRA for the Tract that contains the affected Asset exceeding the Stated NRA for such Tract, then the Title Credit Amount shall be the product of (A) the Allocated Value for such Tract, <u>multiplied</u> by (B) a fraction, the numerator of which is the absolute value of difference between the Stated NRA and the Actual NRA, and the denominator of which is the Stated NRA;
  - (iii) if the Title Credit is of a type not described in Section 3.2(e)(i) through Section 3.2(e)(ii), then the Title Credit Amount will be determined by taking into account the Allocated Value of the affected Tract, the legal effect of the Title Credit, the potential economic effect of the Title Credit over the life of the affected Tract, the values placed upon the Title Credit by Buyer and Seller and such other reasonable factors as are necessary to make a proper evaluation.
  - (iv) The Title Credit Amount with respect to a Title Credit shall be determined without any duplication of any credit included in any other Title Credit Amount hereunder, or for which Seller otherwise receives credit in the calculation of the Purchase Price.
- (f) <u>Cure</u>. Seller may, but shall not be obligated to, cure Title Defects prior to the Closing or after the Closing subject to the provisions of this <u>Section 3.2(f)</u> and <u>Section 3.2(g)</u>. In the event Seller elects, in its sole discretion, to cure one or more Title Defects after Closing, Seller shall deliver a written notice to Buyer prior to Closing identifying the Title Defects that Seller has elected to cure (the "Cure Notice"). If Seller timely delivers a Cure Notice, the Tract or Tracts affected by the Title Defects described in the Cure Notice shall be retained by Seller at Closing, and subject to <u>Section 3.2(d)</u>, the Purchase Price shall be reduced by the Allocated Value of the Tract or Tracts so retained by Seller. Seller

shall be permitted, but shall not be obligated, to attempt to cure such Title Defects for a period of ninety (90) days after Closing (such period, the "Cure Period"), during which time, Buyer shall cooperate with Seller in connection with any such attempts by Seller to so cure. In the event that Seller cures any such Title Defect during the Cure Period, subject to Section 3.2(g), (i) Seller shall promptly notify Buyer that the Title Defect has been cured, (ii) Buyer shall pay to Seller the amount by which the Purchase Price was reduced at Closing with respect to such excluded Assets (or portion thereof), on the first business day following the expiration of the Cure Period (the "Delayed Closing Date"), in accordance with the terms of Agreement, and (iii) Seller shall convey the Assets so excluded at Closing on the Delayed Closing Date pursuant to an instrument effective as of the Effective Time substantially in the same form as the Conveyance.

#### (g) Title Adjustments; Dispute Resolution.

- (i) Seller and Buyer shall attempt to agree in writing on all alleged Title Defects and alleged Title Credits, the appropriate adjustment amounts for such alleged Title Defects and alleged Title Credits and the sufficiency of any cure of any such alleged Title Defects (any disagreement regarding the foregoing, a "**Title Dispute**") prior to Closing.
- (ii) If Seller and Buyer agree in writing prior to Closing on the existence of any Title Defect or Title Credit and the Title Defect Amount or Title Credit Amount in respect thereof, then subject to Section 3.2(d), the Purchase Price shall be reduced at Closing by the aggregate sum of all agreed Title Defect Amounts and the Purchase Price shall be increased at Closing by the aggregate sum of all agreed Title Credit Amounts.
- (iii) If Seller and Buyer are unable to agree in writing prior to Closing on any Title Dispute, and if Seller does not elect to cure under Section 3.2(f), then (A) the Assets included in the Tract or Tracts affected by the unresolved Title Defect shall nevertheless be conveyed to Buyer at the Closing, (B) subject to Section 3.2(d), the Purchase Price shall be reduced by the Title Defect Amount for such Title Defect, as alleged by Buyer, and (C) the Title Dispute shall be exclusively and finally resolved under Section 3.2(g)(iv). Promptly following the final determination of any disputed Title Defect or Title Defect Amount by the Parties or by the Title Consultant in accordance with the procedures set forth in Section 3.2(g)(iv), the Parties shall remit any amounts so determined to be owed to either Party with respect to such disputed matter.
- If the Parties are unable to agree in writing as to any matters that constitute Title Disputes within thirty (30) days following the expiration of the Cure Period, then the Title Disputes shall be exclusively and finally resolved by arbitration to be conducted in Dallas, Texas, by an independent title attorney who has at least fifteen (15) years' experience in oil and gas title involving properties in the state in which the Assets affected by the applicable Title Dispute are located (the "Title Consultant"). If Title Disputes exist with respect to Assets located in multiple states, then separate Proceedings shall be conducted with respect to the Title Disputes relating to Assets in each state. The Title Consultant shall be selected by: (A) mutual written agreement of Buyer and Seller; or (B) absent such agreement, by the Dallas office of the American Arbitration Association (the "AAA"); provided, however, that such Title Consultant must (1) have not been employed by any Party (or its Affiliate) in the past ten (10) years, and (2) have no ethical conflict in serving as the Title Consultant. If there is more than one disputed title matter, all such disputes shall be consolidated into the same arbitration process pursuant to this Section 3.2(g). Seller and Buyer shall each present to the Title Consultant, with a simultaneous copy to the other Party, a single written statement of its position on the defect or benefit in question, together with a copy of this Agreement and any supporting material that such Party desires to furnish, not later than the tenth (10th)

Business Day after appointment of the Title Consultant. The Title Consultant, once appointed, shall have no ex parte communications with any of the Parties concerning the determination required under this Section 3.2(g). All communications between any Party or its Affiliates and the Title Consultant shall be conducted in writing, with copies sent simultaneously to the other Parties in the same manner, or at a meeting or conference call to which each of the Parties and their respective Representatives have been invited and of which such Parties have been provided at least five (5) days' notice. In no event shall the Title Consultant select an amount (a) for any upward adjustment to the Purchase Price that is greater than Seller's proposed amount or less than Buyer's proposed amount or (b) for any downward adjustment to the Purchase Price that is greater than Buyer's proposed amount or less than Seller's proposed amount. Within thirty (30) days after receipt of such materials and after receipt of any additional information required by the Title Consultant, the Title Consultant shall make its determination, which shall be in accordance with the terms of this Agreement and shall be final and binding upon the Parties, without right of appeal, absent Fraud or manifest error. The Title Consultant may not award damages, interest or penalties to either Buyer or Seller with respect to any matter. Seller and Buyer shall each bear their own respective legal fees and other costs of presenting its case. Seller shall bear fifty percent (50%) and Buyer shall bear fifty percent (50%) of the costs and expenses of the Title Consultant. If the Title Consultant determines that no Title Defect or Title Credit, as applicable, exists on an Asset, then after the Title Consultant delivers written notice to Seller and Buyer of his or her award with respect to such Title Defect or Title Credit, as applicable, the matter shall be deemed resolved and no adjustment shall be made on account of such asserted Title Defect or Title Credit, as applicable. If the Title Consultant determines that a Title Defect or Title Credit, as applicable, exists on an Asset, then the Title Consultant will deliver written notice to Seller and Buyer of his or her award with respect to such Title Defect or Title Credit, as applicable, and (i) if the Final Settlement Statement has not been finally resolved, such awards shall be reflected in the Final Settlement Statement, or (ii) if the Final Settlement Statement has been finally resolved, Seller shall deliver to Buyer, or Buyer shall deliver to Seller, as applicable, within five (5) Business Days after receipt of the Title Consultant's award, the amount owed to Buyer or Seller, as applicable, after the determination of such award.

# Article 4 SELLER'S REPRESENTATIONS AND WARRANTIES

As of the Execution Date and the Closing Date, Seller represents and warrants to Buyer that, except as set forth on the Disclosure Schedules:

- **4.1 Organization.** Seller is duly formed, validly existing and in good standing under the laws of the jurisdiction of its formation, with full limited liability company power and authority to enter into this Agreement and perform its obligations under this Agreement.
- **4.2 Qualification.** Seller is duly qualified to do business and is in good standing in each jurisdiction in which the nature of its business as now conducted makes such qualification necessary.
- 4.3 Power; No Conflict. Seller has all requisite power and authority to carry on its business as presently conducted, to enter into this Agreement, each Ancillary Agreement to which it is a party and to consummate the transactions contemplated by this Agreement and each Ancillary Agreement and perform its obligations under this Agreement and each Ancillary Agreement. The execution and the delivery of this Agreement does not, the consummation of the transactions contemplated by this Agreement by Seller does not, and the fulfillment of and compliance with the terms and conditions of this Agreement will not: (a) conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, (b) give rise to a right of termination, modification, revocation, cancellation or acceleration of any obligation or to the loss of a benefit under or (c) result in the creation of any claim upon any of the Assets or any Party under, in each case, any provision of (i) the organizational documents of Seller, (ii) any applicable Law or (iii) any contract or instrument (including any judgment, decree or order) to which Seller is directly a party or by which it is bound or that otherwise relates to any of the Assets, except any matters described in clauses (ii) and (iii) which would not reasonably be expected to have a Material Adverse Effect.
- **4.4 Authorization and Enforceability.**Seller's execution, delivery and performance of or under this Agreement and the Ancillary Agreements to which it is a party have been duly and validly authorized and approved by all necessary limited liability company actions. This Agreement constitutes, and each Ancillary Agreement to which Seller is a party constitutes or will constitute at Closing, Seller's legal, valid and binding obligation, enforceable in accordance with its terms, subject, however, to the effects of bankruptcy, insolvency, reorganization, moratorium, fraudulent transfer and other applicable Laws for the protection of creditors, as well as to general principles of equity, regardless whether such enforceability is considered in a Proceeding in equity or at Law.
- 4.5 Liability for Brokers' Fees. Neither Seller nor its Affiliates (or other related Persons) have incurred any liability, contingent or otherwise, for investment bankers', brokers' or finders' fees relating to the transactions contemplated by this Agreement for which Buyer or its Affiliates shall have any responsibility whatsoever.
- **4.6 Foreign Person.** Seller or its regarded owner (if Seller is a disregarded entity for federal income Tax purposes) is not a "foreign person" within the meaning of Section 1445 of the Internal Revenue Code of 1986, as amended (the "Code") and the regulations promulgated thereunder.
- **4.7 No Bankruptcy.** There are no bankruptcy Proceedings pending, being contemplated by or threatened against Seller.
- **4.8 Litigation.** Except as set forth in <u>Section 4.8</u> of the Disclosure Schedules, (a) there are no Proceedings relating to the Assets, or that challenge or impair Seller's ability to enter into this Agreement or the Ancillary Agreements, or to consummate the transactions contemplated hereby and thereby that are

pending in any court or by or before any Governmental Authority in which Seller is a party; and (b) no Proceeding relating to this transaction or the Assets, or that challenges or impairs Seller's ability to enter into this Agreement or the Ancillary Agreements or to consummate the transactions contemplated hereby and thereby, has been threatened against Seller (or any of its Affiliates). Seller is not in default under any order, writ, injunction or decree of any Governmental Authority applicable to them or any of the Assets, and there are no unsatisfied judgments or injunctions issued by a Governmental Authority outstanding against Seller related to the Assets.

- **4.9** Taxes. Except as set forth on Section 4.9 of the Disclosure Schedules,
- (a) all Tax Returns required to be filed by Seller with respect to Asset Taxes due and payable during Seller's ownership of the Assets have been timely filed and all Asset Taxes reported on such Tax Returns have been duly paid;
- (b) there are no audits, investigations, litigation or other Proceedings with respect to Asset Taxes pending, or threatened in writing, against Seller;
- (c) there are no Encumbrances on any of the Assets attributable to Taxes other than statutory liens for Taxes that are not yet due and payable or that are being contested in good faith by appropriate actions; and
- (d) none of the Assets owned by Seller is subject to any Tax partnership agreement or is otherwise treated, or required to be treated, as held in an arrangement requiring a partnership income Tax Return (other than partnership income Tax Returns required to be filed by Seller) to be filed under Subchapter K of Chapter I of Subtitle A of the Code.
- 4.10 Consents; Preferential Rights; Non-Competes. There are no consents that are required to be obtained in connection with the transfer and conveyance of the Assets to Buyer which expressly provide that the failure to obtain such consent would render the assignment void, or would trigger an obligation of Seller to pay liquidated damages or cause termination or loss of the affected Asset. There are no rights of first refusal, preferential purchase rights, options or other similar provisions to which the Assets are subject that are triggered in connection with the execution of this Agreement by Seller. No Asset is subject to any non-compete or area of mutual interest agreements.
- 4.11 Contracts and Leases. (a) All Material Contracts are listed in Section 4.11 of the Disclosure Schedules, and full copies of each Material Contract in Seller's or its Affiliates' possession have been made available to Buyer prior to the Execution Date, (b) all Material Contracts and Leases are in full force and effect, and each Material Contract and Lease constitutes the legal, valid and binding obligation of Seller, on the one hand, and to Seller's Knowledge, the counterparties thereto, on the other hand, and is enforceable in accordance with its terms, (c) Seller is not in default or otherwise in breach with respect to any of Seller's obligations under any of such Material Contracts or Leases, (d) to Seller's Knowledge, no other Person is in default or, to Seller's Knowledge, otherwise in breach with respect to such Person's obligations under such Material Contracts or Leases, and (e) no event has occurred that with notice or lapse of time or both would constitute any default under any such Material Contract or Lease by any Seller or, to Seller's Knowledge, by any other Person who is a party to such Material Contract or Lease. No Seller has received or given any unresolved written notice of default or termination with respect to any Material Contract or Lease.
- 4.12 Payment of Royalties; No Suspense Accounts; No Deferred Payments. To Seller's Knowledge, there are no monies being held in suspense that are otherwise payable to Seller in respect of (a) any royalty payments, lease rentals, delay rentals, shut-in royalties or other payments under any of the

Leases, or (b) the overriding royalty payments relating to the Overriding Royalty Interests. Seller owes no bonus, deferred purchase price or other acquisition costs, whether contingent or liquidated, on any of the Assets.

- 4.13 Environmental Matters. Except as set forth on Section 4.13 of the Disclosure Schedules, (a) to Seller's Knowledge, Seller is in compliance with all Environmental Laws to the extent pertaining to the ownership of the Assets, (b) there are no civil, criminal, or administrative actions or notices pending or, to Seller's Knowledge, threatened in writing against Seller under any Environmental Law, in each case, of which Seller has received notice, that are related to the Assets or the operations on the Leases, (c) Seller has not entered into any agreements, consents, orders, decrees or judgments with any Governmental Authorities based on any prior violations of Environmental Laws, or any material Losses thereunder, that relate to the future use of the Assets and that require any future remediation, (d) Seller has not received notice from any Governmental Authority that any of the Assets are the subject of any remediation, removal, clean-up, response action, enforcement action or order regarding any actual or alleged presence or release of Hazardous Substances that has not been finally resolved, and (e) there are no pending or, to Seller's Knowledge, threatened claims, demands, written notices of violation or liability or Proceedings under any Environmental Laws that individually or in the aggregate would reasonably be expected to be material to the value of any Asset.
- **4.14 No Cost-Bearing Interests.** Except as set forth in <u>Section 4.14</u> of the Disclosure Schedules, the Assets do not include any interests which obligate Seller to bear a share of drilling, operating or any other costs of production or development.
- **4.15** Condemnation. There is no actual or, or to Seller's Knowledge, threatened taking (whether permanent, temporary, whole or partial) of any part of the Assets by reason of condemnation or the threat of condemnation.

# Article 5 BUYER'S REPRESENTATIONS AND WARRANTIES

As of the Execution Date and the Closing Date, Buyer represents and warrants to Seller that:

- **5.1 Organization.** Buyer is duly formed, validly existing and in good standing under the laws of the jurisdiction of its formation, with full power and authority to enter into this Agreement and perform its obligations under this Agreement.
- **5.2 Qualification.** Buyer is duly qualified to do business and is in good standing in each jurisdiction in which the nature of its business as now conducted makes such qualification necessary, except where the failure to be so qualified or in good standing would not materially hinder or impede the consummation by Buyer of the transactions contemplated by this Agreement.
- 5.3 Power. Buyer has all requisite power and authority to carry on its business as presently conducted, to enter into this Agreement, each Ancillary Agreement to which it is a party and to consummate the transactions contemplated by this Agreement and the Ancillary Agreements and perform its obligations under this Agreement and the Ancillary Agreements. The execution and delivery of this Agreement does not, and the fulfillment of and compliance with the terms and conditions of this Agreement will not: (a) conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, (b) give rise to a right of termination, modification, revocation, cancellation or acceleration of any obligation or to the loss of a benefit under or (c) result in the creation of any claim upon any of the Assets or any Party under, in each case, any provision of (i) Buyer's governing documents, (ii) any applicable Law or (iii) any contract or instrument (including any judgment, decree, order, statute, rule or regulation

applicable to Buyer) to which Buyer or any of its subsidiaries is a party or by which any properties or assets of Buyer or its subsidiaries is bound. No authorization, consent or approval by, or filing with, any Governmental Authority is required for, or in connection with, the authorization, execution, delivery and performance by the Buyer of this Agreement or the consummation of the transactions contemplated by this Agreement and by the Ancillary Agreements; except any matters described in clauses (ii) and (iii) which would not reasonably be expected to have a Material Adverse Effect.

- **5.4 Authorization and Enforceability.** Buyer's execution, delivery and performance of or under this Agreement and the Ancillary Agreements to which it is a party have been duly and validly authorized and approved by all necessary corporate actions. This Agreement constitutes, and each Ancillary Agreement to which Buyer is a party constitutes or will constitute at Closing, Buyer's legal, valid and binding obligation, enforceable in accordance with its terms, subject, however, to the effects of bankruptcy, insolvency, reorganization, moratorium and other applicable Laws for the protection of creditors, as well as to general principles of equity, regardless whether such enforceability is considered in a Proceeding in equity or at Law.
- 5.5 Liability for Brokers' Fees. Buyer has not incurred and will not incur any liability, contingent or otherwise, for investment bankers', brokers' or finders' fees relating to the transactions contemplated by this Agreement for which Seller shall have any responsibility whatsoever.
- **5.6 Litigation.** There are no Proceedings pending, or to Buyer's Knowledge, threatened against Buyer that would challenge or impair the ability of Buyer to enter into this Agreement or the Ancillary Agreements or that would have a Material Adverse Effect upon the ability of Buyer to consummate the transactions contemplated hereby.
- 5.7 Knowledgeable Investor. Buyer is an experienced and knowledgeable investor in the oil and gas business and is sophisticated in the evaluation, purchase, ownership, and operation of oil and gas properties and related facilities. Buyer is not acquiring the Assets in connection with a distribution or resale thereof in violation of federal or state securities Laws and the rules and regulations thereunder.
- **5.8 No Bankruptcy.** There are no bankruptcy Proceedings pending, being contemplated by or threatened against Buyer.

# Article 6 COVENANTS

6.1 Pre-Closing Covenants. From and after the Execution Date and until the earlier to occur of the Closing and the termination of this Agreement in accordance with its terms, and subject to the terms of the Contracts and the Leases, Seller shall administer its Assets in a good and workmanlike manner consistent with its past practices (including, for the avoidance of doubt, the granting of leases and lease renewals in the ordinary course of business) and applicable Law, and shall carry on its business with respect to its interest in the Assets in substantially the same manner as before the Execution Date. Notwithstanding the foregoing, from and after the Execution Date and until Closing, Seller shall not, without Buyer's prior written consent, (a) transfer, sell, mortgage, pledge, abandon or dispose of any of the Assets other than the granting of leases and lease renewals in the ordinary course of business; (b) purchase or otherwise acquire any interests that are similar in type to any of the Interests; (c) amend, modify or terminate any Material Contract or enter into any Contract that would have been a Material Contract if it had been in effect on the Execution Date; or (d) agree, whether in writing or otherwise, to do any of the foregoing; provided that nothing in this Section 6.1 shall be construed to prevent or limit Seller from complying with applicable Laws or the terms of any existing Material Contract or Lease.

- **6.2 Reasonable Efforts.** Each Party agrees that it will not voluntarily undertake any course of action inconsistent with the provisions or intent of this Agreement and will use its Reasonable Efforts to take, or cause to be taken, all action and to do, or cause to be done, all things reasonably necessary, proper, or advisable under applicable Laws to consummate the transactions contemplated by this Agreement, including (a) cooperating in determining whether any consents, approvals, orders, authorizations, waivers, declarations, filings, or registrations of or with any Governmental Authority or third party are required in connection with the consummation of the transactions contemplated by this Agreement; (b) obtaining any consents, approvals, and registrations; (c) causing to be lifted or rescinded any injunction or restraining order or other order adversely affecting the ability of the Parties to consummate the transactions contemplated by this Agreement; (d) defending, and cooperation in defending, all Proceedings challenging this Agreement or the consummation of the transactions contemplated by this Agreement; and (e) executing any additional instruments necessary to consummate the transactions contemplated by this Agreement.
- 6.3 **Amendment of Disclosure Schedules.** From time to time up to the earlier of the Closing Date or termination of this Agreement in accordance with the terms hereof, with respect to the representations and warranties of a Party made in this Agreement, the Parties shall have the right, by written notice to other Party, to promptly supplement or amend the Disclosure Schedules to reflect any event or circumstance that first came into existence after the Execution Date and would have been required to be set forth or described on one or more existing or new Disclosure Schedules if such event or circumstance was in existence on the Execution Date of this Agreement (for the avoidance of doubt, no revision or supplement of the Disclosure Schedules may be effected on account of obtaining knowledge after the Execution Date of an event or circumstance that first came into existence before the Execution Date); provided that any such supplement, amendment or addition shall not be deemed to have (a) cured any inaccuracy in or breach of any representation, warranty or covenant for purposes of determining whether or not the conditions set forth in Sections 8.1 or 8.2 (as applicable) have been met or (b) except with respect to a Subject Schedule Supplement, cured any inaccuracy in or breach of any representation, warranty or covenant for purposes of determining Seller's indemnification obligations pursuant to Section 10.2 (and, for the avoidance of doubt, such liabilities will count against the Deductible). If any matter disclosed pursuant to any schedule supplement by Seller in accordance with this Section 6.3 gives rise to the right of Buyer to terminate this Agreement (such a "Subject Schedule Supplement") and Buyer nevertheless elects to proceed with the Closing, then such matters giving rise to the right of Buyer to terminate that are disclosed pursuant to any such Subject Schedule Supplement shall be deemed waived by Buyer, and Buyer shall not be entitled to make a claim pursuant to Section 10.2(a) with respect to such matters.
- 6.4 Limitations on Representations and Warranties. ALL ORAL INFORMATION OR MATERIALS, DOCUMENTS AND OTHER INFORMATION MADE AVAILABLE TO BUYER BY SELLER OR ANY OF THEIR REPRESENTATIVES AT ANY TIME IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY HAVE BEEN MADE AVAILABLE AS AN ACCOMMODATION AND HAVE BEEN PROVIDED ON AN "AS IS" BASIS. EXCEPT FOR THE EXPRESS AND SPECIFIC REPRESENTATIONS AND WARRANTIES OF SELLER IN ARTICLE 4, THE SELLER OFFICER CERTIFICATES, THE ANCILLARY AGREEMENTS OF SELLER DELIVERED AT THE CLOSING AND THE SPECIAL WARRANTY, AS APPLICABLE, BUYER ACKNOWLEDGES THAT SELLER HAS NOT MADE, AND SELLER HEREBY EXPRESSLY DISCLAIMS AND NEGATES ANY OTHER REPRESENTATION OR WARRANTY (EXPRESS, IMPLIED, AT COMMON LAW, BY STATUTE OR OTHERWISE), INCLUDING THE WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE, AND FURTHER INCLUDING, WITHOUT LIMITATION, ANY REPRESENTATION AND WARRANTIES RELATING TO (A) PRODUCTION RATES, RECOMPLETION OPPORTUNITIES, GEOLOGY, ENGINEERING, DECLINE RATES OR THE QUALITY, QUANTITY OR VOLUME OF THE RESERVES OF OIL AND GAS OR OTHER MINERALS, IF ANY, ATTRIBUTABLE TO THE ASSETS, (B) THE ACCURACY, COMPLETENESS OR MATERIALITY OF ANY INFORMATION, DATA OR OTHER MATERIALS

(WRITTEN OR ORAL) NOW, HERETOFORE OR HEREAFTER FURNISHED TO BUYER BY OR ON BEHALF OF SELLER, AND (C) THE ENVIRONMENTAL CONDITION OF THE ASSETS. SELLER AND BUYER AGREE THAT, TO THE EXTENT REQUIRED BY APPLICABLE LAW TO BE EFFECTIVE, THE DISCLAIMERS OF CERTAIN WARRANTIES CONTAINED IN THIS SECTION ARE "CONSPICUOUS" DISCLAIMERS FOR THE PURPOSES OF ANY APPLICABLE LAW, RULE OR ORDER.

- **6.5 Post-Closing Obligations.** If Closing occurs, Seller and Buyer have the following post-Closing obligations:
- (a) Property Records. Within thirty (30) days after Closing, Seller shall deliver (at Seller's cost and expense) the Records (which may be in the form of original records or electronic copies) to Buyer at the address of Buyer. With respect to any original Records delivered to Buyer, (i) Seller shall be entitled to retain copies of such Records, and (ii) Buyer shall preserve and retain any such original Records for at least seven (7) years beyond the Closing Date. During such seven (7) year period, Seller shall be entitled to obtain access to such Records, at reasonable business hours and upon prior notice to Buyer, so that Seller may make copies of such original Records, at Seller's expense (and Seller shall maintain the strict confidentiality of any Records of which it retains any copies), including as may be reasonable or necessary in connection with any Proceeding or threatened Proceeding against Seller.
- (b) Recording and Filing. Buyer, within thirty (30) days after Closing, shall use Reasonable Efforts to: (i) record the Conveyance and all other instruments that must be recorded to effectuate the transfer of the Assets; and (ii) file for approval with the applicable federal, state or local agencies the Conveyance (to the extent required by applicable Law) and all other federal, state or local transfer documents required to effectuate transfer of the Assets. Buyer shall provide Seller a recorded copy of the Conveyance and other recorded instruments, and approved copies of the Conveyance and other federal, state or local transfer documents, promptly after they are available.
- (c) <u>Special Warranty.</u> Subject to <u>Section 10.3</u>, the Conveyance delivered at Closing shall provide that Seller, will warrant and defend Defensible Title to the Assets against whomsoever lawfully claims the same or any part thereof solely to the extent a claim is brought by, through or under such Seller or any of its Affiliates but not otherwise (the "**Special Warranty**"). To the extent Seller may legally assign such rights and grant such subrogation, Buyer shall have all rights of legal subrogation to third party claims or Seller claims asserted or held by that third party or by Seller arising from or related to the Special Warranty.
- 6.6 Exclusivity. From and after the Execution Date and until the earlier to occur of the Closing and the termination of this Agreement in accordance with its terms: (a) Seller shall not, directly or indirectly through any of its Affiliates or any of its or their respective Representatives, (i) encourage, solicit, initiate, facilitate or continue inquiries regarding bids, offers, inquiries or proposals from any Person (other than Buyer and its designees, agents and all their respective Affiliates) with respect to the sale, transfer or disposition of all or any portion of the Assets ("Acquisition Proposals"), (ii) enter into discussions or negotiations with, or provide any information to, any Person concerning a possible Acquisition Proposal or (iii) enter into any agreements or other instruments (whether or not binding) regarding an Acquisition Proposal; (b) Seller shall immediately cease and cause to be terminated, and shall cause its Affiliates and all of its and their Representatives to immediately cease and cause to be terminated, all existing discussions or negotiations with any Persons with respect to, or that could lead to, an Acquisition Proposal; and (c) Seller shall promptly provide Buyer with notice of any unsolicited Acquisition Proposals received by such Seller.

# Article 7 TAX MATTERS

7.1 Transfer Taxes. All required documentary, filing and recording fees and expenses incurred in connection with the filing and recording of the assignments, conveyances or other instruments required to convey title to the Assets to Buyer shall be borne by Buyer. If any Transfer Taxes are imposed on the transfer of the Assets to Buyer, Buyer shall bear and pay any such Transfer Taxes. Seller and Buyer shall reasonably cooperate in good faith to minimize, to the extent permissible under applicable Law, the amount of any such Transfer Taxes.

#### 7.2 Asset Taxes.

- (a) The Seller shall be allocated all Asset Taxes attributable to (i) any Tax period ending prior to the Effective Time and (ii) the portion of any Straddle Period ending immediately prior to the Effective Time. Buyer shall be allocated and bear all Asset Taxes attributable to (A) any Tax period beginning after the Effective Time and (B) the portion of any Straddle Period beginning at the Effective Time. Buyer shall be responsible for payment to the applicable Governmental Authorities of all Asset Taxes that become due and payable, and the filing of all Tax Returns for Asset Taxes that become due, on or after the Closing Date.
- (b) For purposes of determining the allocations described in Section 7.2(a), (i) Asset Taxes that are based upon income, sales, revenue or similar items (other than Asset Taxes described in clause (ii) or (iii) below) shall be allocated to the period in which the transaction giving rise to such Asset Taxes occurred, (ii) Asset Taxes that are ad valorem, property or similar Asset Taxes imposed on a periodic basis pertaining to a Straddle Period shall be allocated between the portion of such Straddle Period ending immediately prior to the Effective Time and the portion of such Straddle Period beginning at the Effective Time based on the number of days in the applicable Straddle Period that occur before the date on which the Effective Time occurs, on the one hand, and the number of days in such Straddle Period that occur on or after the date on which the Effective Time occurs, on the other hand, and (iii) Asset Taxes that are attributable to the severance or production of Hydrocarbons (other than Asset Taxes described in clause (ii) above) shall be allocated to the period in which the severance or production giving rise to such Asset Taxes occurred. For purposes of (A) the preceding sentence, any exemption, deduction, credit or other item that is calculated on an annual basis shall be allocated pro rata per day between the portion of the Straddle Period ending immediately prior to the Effective Time and the portion of the Straddle Period beginning at the Effective Time; and (B) clause (ii) of the preceding sentence, the period for such Asset Taxes shall begin on the date on which ownership of the applicable Assets gives rise to liability for the particular Asset Tax and shall end on the day before the next such date.
- (c) As provided in Article 2, the Purchase Price shall be adjusted for Asset Taxes allocable to each Party under Section 7.2(a) and Section 7.2(b). To the extent the actual amount of an Asset Tax is not known at the Final Settlement Date, Buyer and Seller shall utilize the most recent information available in estimating the amount of such Asset Tax for purposes of adjusting the Purchase Price under this Section 7.2(c). To the extent the actual amount of an Asset Tax (or the amount thereof paid or economically borne by a Party) is ultimately determined to be different than the amount (if any) that was taken into account for purposes of adjusting the Purchase Price under this Section 7.2(c), Seller shall pay Buyer, or Buyer shall pay Seller, as the case may be, within a commercially reasonable amount of time following written notice of such determination, any amount necessary to cause each of Seller and Buyer to bear its appropriate share of such Asset Tax as determined under this Section 7.2.
- 7.3 Tax Allocation. The Parties agree that the Purchase Price any other amounts treated as consideration for U.S. federal income Tax purposes shall be allocated among the Assets for U.S. federal

and applicable state income Tax purposes in accordance with all applicable Treasury Regulations promulgated under Section 1060 of the Code, and to the extent allowed by applicable Law, in a manner consistent with the Allocated Values. The initial draft of such allocation shall be prepared by Buyer and shall be provided to Seller no later than thirty (30) days after the Final Settlement Date. The Parties shall then cooperate to prepare a final allocation, which also shall be materially consistent with the Allocated Values to the extent allowed by applicable Law. In the event that Seller and Buyer are not able to resolve all of the items in the allocation, all such unresolved items shall be determined by the Accounting Consultant in accordance with the procedures of Section 2.4(c), mutatis mutandis. The final allocation shall be updated to reflect any adjustments to the Purchase Price and any other amounts treated as consideration for U.S. federal income Tax purposes. The final allocation shall be used by Seller and Buyer as the basis for reporting asset values and other items, including preparing Internal Revenue Service Form 8594 (Asset Acquisition Statement under Section 1060), if required. Each Party agrees not to take any position inconsistent with such final allocation unless (a) required by Law or as required by final adjustment or settlement with any Governmental Authority or agency, in which case written notice of such change shall be provided to the other Party, or (b) with the consent of the other Party.

- **7.4 Refunds of Taxes.** Seller shall be entitled to any and all refunds of Taxes allocated to Seller under Section 7.2, and Buyer shall be entitled to any and all refunds of Taxes allocated to Buyer under Section 7.2. If a Party receives a refund of Taxes to which the other Party is entitled under this Section 7.4, the first Party shall promptly pay such amount to the other Party, net of any reasonable costs or expenses incurred by the first Party in procuring such refund.
- **7.5 Cooperation.** Seller and Buyer each agree to furnish or cause to be furnished to the other, upon request, as promptly as practicable, such information and assistance relating to the Assets, including access to books and records, as is reasonably necessary for the filing of all Tax Returns by Buyer or Seller, the making of any election relating to Taxes, the preparation for any audit by any taxing authority and the prosecution or defense of any claim, suit or Proceeding relating to any Tax attributable to the Assets.

# Article 8 CONDITIONS PRECEDENT AND TERMINATION

- **8.1** Conditions to Obligations of Seller. The obligations of Seller to consummate the transactions provided for in this Agreement are subject, at the option of Seller, to the fulfillment on or prior to the Closing of each of the following conditions:
- (a) (i) The Buyer Fundamental Representations shall be true and correct as of the Execution Date and as of the Closing Date as though made on and as of such date (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date) and (ii) the representations and warranties of Buyer contained in this Agreement that are not Buyer Fundamental Representations shall be true and correct as of the Execution Date and as of the Closing Date as though made on and as of such date (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date), except where all such breaches of the representations and warranties described in clause (ii) taken collectively would not have a Buyer Material Adverse Effect;
- (b) Buyer shall have performed or complied in all material respects with all obligations, covenants and agreements contained in this Agreement as to which performance or compliance by Buyer is required prior to the Closing;
- (c) Buyer shall deliver to the Seller a certificate signed by an officer of Buyer in his or her capacity as an officer of Buyer (and not individually), dated as of the Closing Date, certifying that

the conditions specified in <u>Section 8.1(a)</u> and <u>Section 8.1(b)</u> have been fulfilled (the "**Buyer Officer** Certificate");

- (d) No material suit, action, litigation or other Proceeding instituted by any third party shall be pending before any Governmental Authority seeking to restrain, prohibit, enjoin or declare illegal, or seeking substantial damages in connection with, the transactions contemplated by this Agreement, and no statute, rule, regulation, executive order, decree, temporary restraining order, preliminary or permanent injunction, judgment or other order shall have been enacted, entered, promulgated, enforced or issued by any Governmental Authority preventing the consummation of the transactions contemplated by this Agreement;
- (e) Buyer shall be ready, willing and able to deliver to Seller, and shall have delivered to Seller, the closing deliverables to be delivered by Buyer in accordance with Section 9.2; and
- (f) The closing of the transactions contemplated in the Purchase and Sale Agreement dated as of December 6, 2021 among Buyer and Merrimac Properties Partners, LLC and Quarter Horse Energy Partners, LLC (the "Concurrent PSA") shall have occurred concurrently with or prior to the Closing.
- **8.2** Conditions to Obligations of Buyer. The obligations of Buyer to consummate the transactions provided for in this Agreement are subject, at the option of Buyer, to the fulfillment on or prior to the Closing of each of the following conditions:
- (a) (i) The Seller Fundamental Representations shall be true and correct on and as of the Execution Date and the Closing Date as though made on and as of such dates (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date) and (ii) the representations and warranties of Seller contained in this Agreement that are not Seller Fundamental Representations shall be true and correct on and as of the Execution Date and the Closing Date as though made on and as of such dates (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date), except where all such breaches of the representations and warranties described in clause (ii) taken collectively would not have a Seller Material Adverse Effect.
- (b) Seller shall have performed or complied in all material respects all with all obligations, covenants and agreements contained in this Agreement as to which performance or compliance by Seller is required prior to the Closing.
- (c) Seller shall deliver to Buyer a certificate signed by an officer of Seller in his or her capacity as an officer of Seller (and not individually), dated as of the Closing Date, certifying that the conditions specified in Section 8.2(a) and Section 8.2(b) have been fulfilled (each a "Seller Officer Certificate" and, together with the Buyer Officer Certificate, the "Officer Certificates").
- (d) No material suit, action, litigation or other Proceeding instituted by any third party shall be pending before any Governmental Authority seeking to restrain, prohibit, enjoin or declare illegal, or seeking substantial damages in connection with, the transactions contemplated by this Agreement, and no statute, rule, regulation, executive order, decree, temporary restraining order, preliminary or permanent injunction, judgment or other order shall have been enacted, entered, promulgated, enforced or issued by any Governmental Authority preventing the consummation of the transactions contemplated by this Agreement;
- (e) Seller shall be ready, willing and able to deliver to Buyer, and shall have delivered to Buyer, the closing deliverables to be delivered by Seller in accordance with Section 9.2; and

- (f) The closing of the transactions contemplated in the Concurrent PSA shall have occurred concurrently with or prior to the Closing.
- **8.3 Frustration of Closing Conditions.** Neither Buyer nor Seller may rely, either as a basis for not consummating the transactions contemplated in this Agreement or terminating this Agreement in accordance with this Agreement and abandoning the transactions contemplated in this Agreement, on the failure of any condition set forth in Section 8.1(a) Section 8.1(b) or Section 8.1(f) or Section 8.2(a), Section 8.2(b) or Section 8.2(f), as the case may be, to be satisfied if such failure was caused by such Party's breach of any provision of this Agreement (or, as it pertains to the conditions in Section 8.1(f) and Section 8.2(f), the breach by such Party or its Affiliate of the Concurrent PSA) or failure of such Party or its Affiliate to act in good faith. For clarity, for purposes of this Section 8.3, Seller and the "Seller" under the Concurrent PSA are Affiliates of one another.

## **8.4 Termination.** This Agreement may be terminated prior to Closing:

- (a) by the mutual written consent of the Parties;
- (b) by Seller, at its option, if any of the conditions set forth in Section 8.1 have not been satisfied on or before the Scheduled Closing Date and, following written notice thereof from Seller to Buyer specifying the reason such condition is unsatisfied (including any breach by Buyer of this Agreement), such condition remains unsatisfied for a period of ten (10) Business Days after Buyer's receipt of written notice thereof from Seller;
- (c) by Buyer, at its option, if any of the conditions set forth in <u>Section 8.2</u> have not been satisfied on or before the Scheduled Closing Date and, following written notice thereof from Buyer to Seller specifying the reason such condition is unsatisfied (including any breach by any Seller of this Agreement), such condition remains unsatisfied for a period of ten (10) Business Days after Seller's receipt of written notice thereof from Buyer;
- (d) by Seller or Buyer, if the sum of all Title Defect Amounts for Title Defects asserted by Buyer in good faith on or before the Title Claims Date that exceed the Title Threshold exceeds thirty percent (30%) of the unadjusted Purchase Price;
- (e) by Seller or Buyer if Closing shall not have occurred on or before the Scheduled Closing Date; or
- (f) automatically, unless the Parties agree otherwise in writing, upon the termination of the Concurrent PSA;

provided, however, that no Party shall have the right to terminate this Agreement under clauses (b), (c) or (e) above if (i) such Party is, at such time, in material breach of any provision of this Agreement, or (ii) the other Party has filed in good faith (and then solely for so long as such Party continues to diligently pursue) an action before a Governmental Authority seeking specific performance of such first Party's obligation under this Agreement to consummate the Closing as permitted by Section 11.16.

#### 8.5 Effect of Termination and Remedies.

(a) <u>Seller's Remedies</u>. Subject to <u>Section 8.5(c)</u>, prior to Closing, Seller's sole and exclusive remedy for any breach by Buyer of this Agreement is (i) to seek specific performance of Buyer's obligations under this Agreement in accordance with <u>Section 11.17</u> or (ii) to terminate this Agreement in accordance with Section 8.4. Upon such termination, Seller shall be free immediately to enjoy all rights of

ownership of the Assets and to sell, transfer, encumber or otherwise dispose of the Assets to any Person without any restriction under this Agreement. Subject to <u>Section 8.5(c)</u>, prior to Closing, the remedies set forth in this <u>Section 8.5(a)</u> shall be Seller's sole and exclusive remedies for Buyer's default or Seller's termination of this Agreement, and Seller hereby expressly waive and release all other remedies.

- (b) <u>Buyer's Remedies</u>. Subject to <u>Section 8.5(c)</u>, prior to Closing, Buyer's sole and exclusive remedy for any breach by Seller of this Agreement is (i) to seek specific performance of Seller's obligations under this Agreement in accordance with <u>Section 11.17</u> or (ii) to terminate this Agreement in accordance with <u>Section 8.4</u> and Buyer will be entitled to recover its actual damages against Seller. Subject to <u>Section 8.5(c)</u>, prior to Closing, the remedies set forth in this <u>Section 8.5(b)</u> shall be Buyer's sole and exclusive remedies for Seller's default or Buyer's termination of this Agreement, and Buyer hereby expressly waives and releases all other remedies.
- (c) <u>Exceptions</u>. Notwithstanding anything in this Agreement to the contrary, termination of this Agreement shall not prejudice or impair Seller's or Buyer's rights and obligations under the provisions of this <u>Section 8.5</u>, and all of <u>Article 11</u> (other than <u>Section 11.4</u>), which shall survive the termination of this Agreement.
- (d) <u>No Other Remedies</u>. Subject to (and except as otherwise provided in) the foregoing provisions in this <u>Section 8.5</u>, following any termination of this Agreement prior to Closing, no Party shall have any liability or obligation under this Agreement and Seller shall be free immediately to enjoy all rights of ownership of the Assets and to sell, transfer, encumber or otherwise dispose of the Assets to any Person without any restriction under this Agreement; <u>provided</u>, <u>however</u>, in no event shall Seller dispose any of the Assets if (i) there are any actions or claims pending or litigation threatened as a result of any alleged breaches of the Agreement by Seller or (ii) Buyer is seeking specific performance pursuant to the terms of the Agreement.

## Article 9 CLOSING

#### 9.1 Date and Place of Closing.

- Date of Closing. Subject to the satisfaction or waiver of the applicable conditions to Closing, the consummation of the transactions contemplated by this Agreement (the "Closing") shall be held virtually and electronically on December 15, 2021 (such date, the "Scheduled Closing Date"): provided, that, if all conditions to Closing in Article 8 (other than those conditions that are only capable of being satisfied at Closing, but subject to all such conditions being satisfied or waived at Closing) have not been yet waived in writing or satisfied on or before the Scheduled Closing Date, then, subject to Section 8.4 (and without limiting Section 8.4(e)), the Closing shall occur within five (5) Business Days following the date on which all such conditions have been waived in writing or satisfied (other than those conditions that are only capable of being satisfied at Closing, but subject to all such conditions being satisfied or waived at Closing), or on such earlier or later date or at such other place or in such other manner as the Parties agree in writing. The date on which the Closing actually occurs is hereinafter referred to as the "Closing Date." All events of Closing shall each be deemed to have occurred simultaneously with the other, regardless of when actually occurring and each will be a condition precedent to the other. If the Closing occurs, all conditions of Closing shall be deemed to have been satisfied or waived (but Seller's and Buyer's warranties, representations, covenants and indemnities are not waived and will survive the Closing to the extent provided in Section 10.3).
- (b) <u>Place of Closing</u>. Closing will take place remotely by electronic exchange of documents and signatures, or at such other place as mutually agreed upon by the Parties.

# **9.2** Closing Obligations. At Closing:

- (a) <u>Seller and Buyer shall deliver to one another the Officer Certificates, as applicable;</u>
- (b) Seller and Buyer shall execute, acknowledge and deliver, with respect to all of the Assets, including the Mineral Interests, Royalty Interests, Overriding Royalty Interests, a deed and assignment effective as of the Effective Time substantially in the form of Exhibit B (the "Conveyance");
- (c) <u>Seller and Buyer shall execute and deliver an acknowledgement of the Preliminary</u> Settlement Statement;
- (d) Buyer shall cause the Purchase Price, as adjusted in accordance with the Preliminary Settlement Statement, to be paid by wire transfer of immediately available funds to the account(s) designated by the Seller in writing;
- (e) Seller shall deliver to Buyer an executed certificate of non-foreign status that meets the requirements set forth in Treasury Regulations § 1.1445-2(b)(2), substantially in the form attached hereto as Exhibit C (the "FIRPTA Certificate"); and
- (f) Seller and Buyer shall execute and deliver any other agreements, instruments or documents which are expressly required by the other terms of this Agreement to be executed and delivered at the Closing.

# Article 10 ASSUMPTION AND RETENTION OF OBLIGATIONS AND INDEMNIFICATION; DISCLAIMERS

10.1 Buyer's Assumption of Liabilities and Obligations. From and after Closing, subject to Seller's obligations in Section 10.2(a), (a) Buyer shall assume and pay, perform, fulfill and discharge (or cause to be paid, performed, fulfilled and discharged) all claims, costs, expenses, liabilities and obligations (collectively, "Obligations") of Seller, known or unknown, arising from, based upon, relating to or associated with the Assets, or to the ownership or management of the Assets, arising on or after the Effective Time, in each case, other than the Retained Obligations (the "Assumed Liabilities"); provided that the Assumed Liabilities shall not include, and Seller shall retain all (and Buyer shall have no obligation with respect to any) Obligations arising from or related to the Retained Obligations. For the avoidance of doubt, any calculation or determination of any Obligations under this Agreement shall not include any measure of damages or theories of liability expressly waived by the Parties pursuant to Section 10.7.

#### 10.2 Indemnification.

- (a) <u>Seller's Indemnification of Buyer.</u> From and after Closing, subject to the limitations set forth in this <u>Article 10</u> and <u>Section 11.11</u> below, Seller shall be responsible for, pay on a current basis, and **DEFEND, INDEMNIFY AND SAVE AND HOLD HARMLESS** Buyer and its Affiliates, and its and their respective equityholders, partners, members, directors, officers, managers, employees, attorneys, consultants, agents and representatives (collectively, the "**Buyer Indemnitees**"), from and against all Losses and Obligations, whether or not incurred in the investigation or defense of any of the same or in asserting, preserving or enforcing any of their respective rights hereunder, arising directly or indirectly from, based upon, in connection with, related to, or that are associated with:
  - (i) Any breach by any Seller of any of its representations and warranties contained in this Agreement;

- (ii) Any breach by any Seller of any of its covenants or agreements under this Agreement;
  - (iii) Seller Taxes; and
  - (iv) the Retained Obligations.
- (b) <u>Buyer's Indemnification of Seller</u>. From and after the Closing, subject to the limitations set forth in this <u>Article 10</u> and <u>Section 11.11</u> below, Buyer shall be responsible for, pay on a current basis, and **DEFEND, INDEMNIFY AND SAVE AND HOLD HARMLESS** Seller and its Affiliates, their respective equityholders, partners, members, directors, officers, managers, employees, attorneys, consultants, agents and representatives (collectively, the "**Seller Indemnitees**"), from and against all Losses and Obligations, whether or not incurred in the investigation or defense of any of the same or in asserting, preserving or enforcing any of their respective rights hereunder, arising directly or indirectly from, based upon, in connection with, related to, or that are otherwise associated with:
  - (i) any breach by Buyer of any of its representations or warranties contained in this Agreement;
  - (ii) any breach by Buyer of any of its covenants or agreements under this Agreement; and
    - (iii) the Assumed Liabilities.

#### (c) <u>Limitations on Indemnity</u>.

- (i) Subject to and without limiting <u>Section 11.16</u>, following the Closing, except with respect to Special Warranty and Fraud, indemnification under this <u>Section 10.2</u> shall be the sole and exclusive remedy available to each Party (and in lieu of all other remedies) against the other Party for any claims arising out of or based upon the matters set forth in this Agreement and the transactions contemplated hereby, and (except as provided in <u>Section 11.16</u>) no Party shall seek relief against the other Party other than through the indemnification provided in this <u>Section 10.2</u>.
- (ii) Seller shall not have any liability under <u>Section 10.2(a)(i)</u> for any individual Losses arising from breaches of representations and warranties (other than the Seller Fundamental Representations) unless such individual Loss exceeds an amount equal to Five Thousand Dollars (\$5,000.00) (the "**Individual Claim Threshold**"), and then only to the extent that the aggregate amount of all such Losses which meet the Individual Claim Threshold, collectively, exceed Fifty Thousand Dollars (\$50,000.00) (the "**Deductible**"), in which event Seller shall be liable for the aggregate amount of all such Losses.
- (iii) No indemnification may be sought by any Party under this Agreement in respect of any Loss to the extent such Loss is reflected in the calculations of the Preliminary Settlement Statement or as finally determined in connection with the Final Settlement Statement. No indemnified Party shall be entitled to recover more than once for the same Loss.
- (iv) Notwithstanding anything to the contrary contained elsewhere in this Agreement, except in cases where a Party has acted with Fraud, or except for breaches of the Seller Fundamental Representations or Buyer Fundamental Representations, as

applicable, in no event shall Seller have any obligation to indemnify Buyer or Buyer Indemnitees against, or reimburse Buyer or any Buyer Indemnitees for, any Obligations or Losses arising under Section 10.2(a)(i) in excess of an amount equal to thirty percent (30%) of the unadjusted Purchase Price. In no event shall either Party's aggregate liability under this Agreement for any and all indemnification obligations exceed an amount equal to the Purchase Price.

- (v) Any claim for indemnity to which a Seller Indemnitee or Buyer Indemnitee is entitled must be asserted by and through Seller or Buyer, as applicable. The amount of any indemnification provided under Section 10.2(a) and Section 10.2(b) shall be reduced by any amounts actually recovered by the indemnified Party under insurance policies or otherwise from third parties, net of any collection costs (including Taxes), and excluding any insurance underwritten by the indemnified Parties.
- 10.3 Survival of Warranties, Representations and Covenants. All representations, warranties, covenants, and performance obligations set forth in this Agreement shall survive the Closing until fully performed, subject to applicable statutes of limitations, except:
- (a) the representations and warranties of the Seller in <u>Article 4</u> (other than the Seller Fundamental Representations) shall survive Closing for a period of twelve (12) months;
- (b) the representations and warranties of Seller set forth in <u>Section 4.1</u> through <u>Section 4.7</u> (inclusive) and <u>Section 4.9</u> (such representations, the "**Seller Fundamental Representations**") shall survive the Closing without time limit, except that the representations in <u>Section 4.9</u> shall survive for the applicable statute of limitations period plus sixty (60) days;
- (c) the representations and warranties of Buyer set forth in <u>Article 5</u> (other than the Buyer Fundamental Representations) shall survive Closing for a period of twelve (12) months;
- (d) the representations and warranties of Buyer set forth in <u>Section 5.1</u> through <u>Section 5.5</u> (inclusive) and <u>Section 5.8</u> (such representations, the "**Buyer Fundamental Representations**") shall survive the Closing without time limit;
  - (e) the Special Warranty shall survive Closing for a period of thirty-six (36) months;
- (f) any covenants of the Parties to be performed prior to or at Closing shall expire upon and as of Closing;
- (g) the indemnities in <u>Section 10.2(b)(iii)</u> shall survive the Closing for a period of thirty-six (36) months; and
- (h) the indemnities in <u>Section 10.2(a)(iv)</u> shall survive the Closing without time limit; and
- (i) the indemnity in Section 10.2(a)(iii) shall survive until sixty (60) days after the expiration of the applicable statute of limitations (each such period described in clauses (a) through (h), as applicable, the "Survival Period").

All indemnities herein other than those in <u>Section 10.2(b)(iii)</u>, <u>Section 10.2(a)(iv)</u> and <u>Section 10.2(a)(iii)</u> shall terminate as of the termination date of each respective representation, warranty, covenant or performance obligation that is subject to the indemnification thereto (as specified herein), in each case,

except to the extent a claim notice with respect to any breach of the foregoing has been properly delivered before the expiration of the Survival Period applicable thereto alleging a right to indemnification or defense for Obligations or Losses arising out of, relating to, or otherwise attributable to the breach of such representation, warranty, covenant or performance obligation, then, solely to the extent directly related to the matters expressly set forth in such claim notice, such representation, warranty, covenant or performance obligation shall continue to survive until the claims asserted in such claim notice have been fully and finally resolved under the terms of this Agreement or by written agreement (including a settlement agreement) of the Parties.

- 10.4 Reservation as to Non-Parties. Nothing in this Agreement is intended to limit or otherwise waive any recourse Buyer or Seller may have against any non-Party for any Obligations or Losses that may be incurred with respect to the Assets.
- 10.5 Acceptance of Title Condition. EXCEPT AS OTHERWISE SET FORTH IN THIS AGREEMENT (INCLUDING, WITHOUT LIMITATION, ANY REPRESENTATIONS, WARRANTIES AND COVENANTS OF SELLER UNDER THIS AGREEMENT) AND EXCEPT AS TO THE SPECIAL WARRANTY OF DEFENSIBLE TITLE UNDER THIS AGREEMENT OR THE CONVEYANCE DELIVERED AT CLOSING, BUYER WILL ACCEPT THE ASSETS AT CLOSING IN THEIR PRESENT CONDITION, "AS IS AND WHERE IS AND WITH ALL FAULTS." BUYER ACKNOWLEDGES AND AGREES THAT, EXCEPT AS OTHERWISE SET FORTH IN THIS AGREEMENT AND IN THE CONVEYANCE DELIVERED AT CLOSING, SELLER HAVE MADE NO REPRESENTATIONS OR WARRANTIES OF ANY KIND, EXPRESS OR IMPLIED, WRITTEN, ORAL, OR OTHERWISE, AS TO THE ACCURACY OR COMPLETENESS OF THE BACKGROUND MATERIALS OR ANY OTHER INFORMATION RELATING TO THE ASSETS FURNISHED BY OR ON BEHALF OF SELLER OR TO BE FURNISHED TO BUYER OR ITS REPRESENTATIVES, INCLUDING SELLER'S INTERNAL APPRAISALS AND INTERPRETIVE DATA.
- 10.6 Express Negligence Rule. THE DEFENSE, INDEMNIFICATION, HOLD HARMLESS, RELEASE AND ASSUMED LIABILITIES PROVISIONS PROVIDED FOR IN THIS AGREEMENT SHALL BE APPLICABLE AND ENFORCEABLE WHETHER OR NOT THE LIABILITIES, LOSSES, COSTS, EXPENSES, OBLIGATIONS AND DAMAGES IN QUESTION AROSE OR RESULTED SOLELY OR IN PART FROM THE ACTIVE, PASSIVE, CONCURRENT OR COMPARATIVE NEGLIGENCE, STRICT LIABILITY OR OTHER FAULT OR VIOLATION OF LAW OF OR BY ANY INDEMNIFIED PARTY. BUYER AND SELLER ACKNOWLEDGE THAT THIS STATEMENT COMPLIES WITH THE EXPRESS NEGLIGENCE RULE AND IS CONSPICUOUS.
- Waiver of Certain Damages. NOTWITHSTANDING ANYTHING IN THIS 10.7 AGREEMENT TO THE CONTRARY, NO PARTY SHALL BE LIABLE TO THE OTHER PARTY FOR SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES SUFFERED BY SUCH PARTY RESULTING FROM OR ARISING OUT OF THIS AGREEMENT OR THE BREACH THEREOF OR UNDER ANY OTHER THEORY OF LIABILITY, WHETHER TORT, NEGLIGENCE, STRICT LIABILITY, BREACH OF CONTRACT, WARRANTY, INDEMNITY OR OTHERWISE, INCLUDING LOSS OF USE, INCREASED COST OF OPERATIONS, LOSS OF PROFIT OR REVENUE, DIMINUTION IN VALUE, LOST BUSINESS OPPORTUNITY OR BUSINESS INTERRUPTIONS. IN FURTHERANCE OF THE FOREGOING, EACH PARTY RELEASES THE OTHER PARTY AND WAIVES ANY RIGHT OF RECOVERY FOR SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES (INCLUDING ANY CALCULATIONS OF LOSSES USING ANY "MULTIPLES OF PROFITS", "MULTIPLES OF CASH-FLOW" OR OTHER SIMILAR METHODOLOGY) SUFFERED BY SUCH PARTY REGARDLESS OF WHETHER ANY SUCH DAMAGES ARE CAUSED BY THE OTHER PARTY'S NEGLIGENCE (AND REGARDLESS OF WHETHER SUCH NEGLIGENCE IS SOLE, JOINT, CONCURRENT, ACTIVE,

PASSIVE OR GROSS NEGLIGENCE), FAULT, OR LIABILITY WITHOUT FAULT: <u>PROVIDED</u>, <u>HOWEVER</u>, THE FOREGOING SHALL NOT BE CONSTRUED AS LIMITING AN OBLIGATION OF A PARTY TO INDEMNIFY, RELEASE, DEFEND AND HOLD HARMLESS THE OTHER PARTY AGAINST CLAIMS ASSERTED BY THIRD PARTIES, INCLUDING, THIRD PARTY CLAIMS FOR SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES.

- 10.8 Redhibition Waiver. BUYER: (A) WAIVES ALL RIGHTS IN REDHIBITION PURSUANT TO LOUISIANA CIVIL CODE ARTICLE 2475 AND ARTICLES 2520 THROUGH 2548; (B) ACKNOWLEDGES THAT THIS EXPRESS WAIVER SHALL BE CONSIDERED A MATERIAL AND INTEGRAL PART OF THIS SALE AND THE CONSIDERATION THEREOF; AND (C) ACKNOWLEDGES THAT THIS WAIVER HAS BEEN BROUGHT TO THE ATTENTION OF BUYER, HAS BEEN EXPLAINED IN DETAIL AND THAT BUYER HAS VOLUNTARILY AND KNOWINGLY CONSENTED TO THIS WAIVER OF WARRANTY OF FITNESS AND WARRANTY AGAINST REDHIBITORY VICES AND DEFECTS WITH RESPECT TO THE ASSETS.
- 10.9 UTPCPL Waiver. TO THE EXTENT APPLICABLE TO THE ASSETS OR ANY PORTION THEREOF, BUYER HEREBY WAIVES THE PROVISIONS OF THE LOUISIANA UNFAIR TRADE PRACTICES AND CONSUMER PROTECTION LAW (LA. R.S. 51.1401, ET SEQ.). BUYER WARRANTS AND REPRESENTS THAT IT: (A) IS EXPERIENCED AND KNOWLEDGEABLE WITH RESPECT TO THE OIL AND GAS INDUSTRY GENERALLY AND WITH TRANSACTIONS OF THIS TYPE SPECIFICALLY; (B) POSSESSES AMPLE KNOWLEDGE, EXPERIENCE AND EXPERTISE TO EVALUATE INDEPENDENTLY THE MERITS AND RISKS OF THE CONTEMPLATED TRANSACTIONS; AND (C) IS NOT IN A SIGNIFICANTLY DISPARATE BARGAINING POSITION.
- 10.10 Determination of Losses. With respect to any Party's indemnification obligations under this Article 10, solely for purposes of determining the amount of Losses resulting from a breach or inaccuracy of a representation or warranty of the Seller in Article 4 or of the Buyer in Article 5 (including any bringdown of any such representation or warranty in any certificate delivered pursuant to this Agreement), such representations and warranties shall be read without giving effect to any materiality, Material Adverse Effect or similar qualification therein.
- Party in writing (including the full particulars of such claim) of any potential Obligations or Losses of which it becomes aware and for which it is entitled to indemnification from such other Party under this Agreement; provided that the failure to so notify such other Party shall not affect the other Party's indemnification obligations under this Agreement with respect to such claim except to the extent that such other Party is materially prejudiced by such failure. The indemnifying Party is obligated to defend at the indemnifying Party's sole expense any litigation or other administrative or adversarial proceeding against the indemnified Party relating to any claim for which the indemnifying Party has agreed to indemnify and hold the indemnified Party harmless under this Agreement; provided, however, the indemnified Party shall have the right to participate with the indemnifying Party in the defense of any such claim at its own expense.

## Article 11 MISCELLANEOUS

11.1 Expenses. Except as otherwise specifically provided in this Agreement, all fees, costs and expenses incurred by Buyer, on the one hand, or Seller, on the other, in negotiating this Agreement and the Ancillary Agreements or in consummating the transactions contemplated by this Agreement shall be paid by the Party incurring the same.

11.2 Notices. All notices and communications required or permitted to be given under this Agreement, shall be sufficient in all respects if given in writing and delivered personally, sent by electronic mail (so long as such electronic mail is acknowledged by the recipient as having been received; provided that such recipient is affirmatively obligated to promptly acknowledge receipt), sent by bonded overnight courier or mailed by U.S. Express Mail, Federal Express or United Parcel Service Express Delivery or by certified or registered United States Mail with all postage fully prepaid, addressed to the appropriate person at the address for such Person shown below:

#### If to Seller:

c/o SRP Management Services, LLC 3811 Turtle Creek Blvd, Suite 1100 Dallas, Texas 75219 Attn: Matt Veazey, Director

Email: matt.veazey@srp-ok.com Telephone: 214-699-4954

with a copy (which will not constitute notice) to:

Kirkland & Ellis LLP 609 Main Street Houston, Texas 77002 Attn: Rahul D. Vashi, P.C.

Lindsey M. Jaquillard Email: rahul.vashi@kirkland.com

lindsey.jaquillard@kirkland.com

Telephone: 713-836-3600

#### If to Buyer:

PHX Minerals Inc. 1601 NW Expressway Valliance Bank Building, Suite 1100 Oklahoma City, Oklahoma 73118

Attn: Chad Stephens Email: chad@phxmin.com Telephone: (405) 945-6105

with a copy (which will not constitute notice) to:

Jackson Walker LLP 1401 McKinney St., Suite 1900 Houston, Texas 77010 Attn: Kirk Tucker Email: ktucker@jw.com

Telephone: (713) 752.4389

Any notice given in accordance with this Agreement shall be deemed to have been given when delivered to the addressee in person, by electronic mail (provided such electronic mail is acknowledged by

the recipient as described above) or by courier during normal business hours, or upon actual receipt by the addressee after such notice has either been delivered to an overnight courier or deposited in the United States Mail or with Federal Express or United Parcel Service, as the case may be. Any Party may change its contact information for notice by giving written notice to the other Party in the manner provided in this Section 11.2. If the date specified in this Agreement for giving any notice or taking any action is not a Business Day (or if the period during which any notice is required to be given or any action taken expires on a date which is not a Business Day), then the date for giving such notice or taking such action (and the expiration date of such period during which notice is required to be given or action taken) shall be the next day which is a Business Day.

#### 11.3 Governing Law; Jurisdiction; Venue: Jury Waiver; Binding Arbitration.

- (a) This Agreement, the Ancillary Agreements and any other documents delivered pursuant hereto and the legal relations between the Parties shall be governed and construed in accordance with the laws of the State of Texas, without giving effect to principles of conflicts of laws that would result in the application of the laws of another jurisdiction; <u>provided</u> that any dispute pertaining to real property shall be governed by the laws of the state in which such real property is located.
- Subject to and without limiting Section 2.4(c), the Parties agree that any and all disputes or claims by any Party arising from or related to this Agreement that cannot be amicably settled, will be determined solely and exclusively by arbitration in accordance with the Federal Arbitration Act and using the rules of the American Arbitration Association or any successor thereof when not in conflict with such act; provided, however, that nothing contained in this Section 11.3 shall limit any Party's right to bring (i) post-arbitration actions seeking to enforce an arbitration award or (ii) actions seeking injunctive or other similar relief under Section 11.16 in the event of a breach or threatened breach of any of the provisions of this Agreement (or any other agreement contemplated hereby). Arbitration will take place at an appointed time and place in Dallas, Texas, at a mutually agreeable and neutral venue. The Parties to the dispute will attempt in good faith to agree on a single impartial arbitrator. If such Parties cannot agree on a single impartial arbitrator, each Party to the dispute will select one impartial arbitrator, and the two so designated will select a third (3rd) impartial arbitrator. If either Party to the dispute should fail to designate an arbitrator within fourteen (14) days after arbitration is requested, or if the two (2) arbitrators should fail to select a third arbitrator within thirty (30) days after arbitration is requested, then an arbitrator will be selected in accordance with the Commercial Arbitration Rules of the American Arbitration Association. Each arbitrator will have at least ten (10) years' experience in the oil and gas industry in the State of Texas (or, if the dispute pertains to real property, the state where such real property is located). Discovery will be made pursuant to the Federal Rules of Civil Procedure. Final hearing on the matter will be held within nine (9) months after the date the Parties to the dispute agree on a single arbitrator or the selection of the third arbitrator, as applicable (such date, the "Arbitration Commencement Date") and a final decision with a written opinion stating the reasons therefor will be rendered within fourteen (14) days after the conclusion of the testimony at the final hearing, which written opinion shall be final and binding; provided, however, that Seller will have the exclusive right, in their discretion, to require that (A) discovery be completed by a specific date, which date must be at least forty-five (45) days after the Arbitration Commencement Date, or (B) the final hearing be commenced by a date earlier than nine (9) months after the Arbitration Commencement Date, which date must be at least sixty (60) days after the Arbitration Commencement Date. The prevailing Party in the arbitration proceeding shall recover all reasonable and necessary attorneys' fees and costs, including, but not limited to, all attorneys' fees, expert witness fees, court reporter fees, support staff fees, arbitration venue fees, and arbitrator fees. A Party need not be awarded any damages to be considered the prevailing Party, and a Party need not succeed on all or any affirmative claims to be considered the prevailing Party. The Parties irrevocably waive their right to any form of appeal, review or recourse to any state court or other judicial authority, insofar as such waiver may be validly made. Judgment upon an award of the majority of the arbitrators will be binding. Judgment on the award may be entered

and enforced by any court of competent jurisdiction. In no event, may the arbitrators award damages that have been waived under this Agreement. The arbitration process will be kept confidential and such conduct, statements, promises, offers, views and opinions will not be discoverable or admissible in any legal proceeding for any purpose, except to the extent reasonably necessary to enforce the final decision of the arbitrators.

- FOR PURPOSES OF (I) ANY POST-ARBITRATION ACTIONS SEEKING TO (c) ENFORCE AN ARBITRATION AWARD UNDER THIS AGREEMENT OR (II) ACTIONS SEEKING INJUNCTIVE OR OTHER SIMILAR RELIEF UNDER SECTION 11.16, THE VENUE FOR ANY SUCH ACTION BROUGHT UNDER THIS AGREEMENT SHALL BE DALLAS COUNTY, TEXAS. EACH PARTY CONSENTS TO PERSONAL JURISDICTION IN ANY ACTION BROUGHT IN THE UNITED STATES FEDERAL COURTS LOCATED WITHIN DALLAS COUNTY, TEXAS (OR, IF JURISDICTION IS NOT AVAILABLE IN THE UNITED STATES FEDERAL COURTS, TO PERSONAL JURISDICTION IN ANY ACTION BROUGHT IN THE STATE COURTS LOCATED IN DALLAS COUNTY, TEXAS) WITH RESPECT TO ANY DISPUTE, CLAIM OR CONTROVERSY ARISING OUT OF OR IN RELATION TO OR IN CONNECTION WITH THIS AGREEMENT, AND EACH OF THE PARTIES AGREES THAT ANY ACTION INSTITUTED BY IT AGAINST THE OTHER WITH RESPECT TO ANY SUCH DISPUTE, CONTROVERSY OR CLAIM WILL BE INSTITUTED EXCLUSIVELY IN THE UNITED STATES DISTRICT COURT FOR THE NORTHERN DISTRICT OF TEXAS. THE PARTIES HEREBY WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM BROUGHT BY ANY PARTY AGAINST ANOTHER IN ANY MATTER WHATSOEVER ARISING OUT OF OR IN RELATION TO OR IN CONNECTION WITH THIS AGREEMENT. IN ADDITION, EACH PARTY IRREVOCABLY WAIVES ANY OBJECTION, INCLUDING ANY OBJECTION TO THE LAYING OF VENUE OR BASED ON THE GROUNDS OF FORUM NON CONVENIENS, WHICH IT MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY SUCH ACTION IN THE RESPECTIVE JURISDICTIONS REFERENCED IN THIS SECTION.
- 11.4 Further Assurances. From time to time after Closing, Seller and Buyer shall each execute, acknowledge and deliver to the other such further instruments and take such other action as may be reasonably requested in order to accomplish more effectively the purposes of the transactions contemplated by this Agreement.
- 11.5 Amendments: Waivers. This Agreement may not be amended, nor any rights hereunder waived except by an instrument in writing signed by the Party to be charged with such amendment or waiver and delivered by such Party to the Party claiming the benefit of such amendment or waiver. No waiver of any breach or violation or, default under or inaccuracy in any representation, warranty or covenant set forth in this Agreement will be deemed to extend to any prior or subsequent breach, violation, default, inaccuracy or affect in any way any rights arising by virtue of any prior or subsequent occurrence. No delay or omission on the part of any Party in exercising any right, power or remedy under this Agreement will operate as a waiver thereof.
- Agreement or any of its rights under this Agreement without the prior written consent of the other Parties (which consent may be granted or withheld in each such other Party's respective sole discretion); provided, however, that Buyer shall have the right, upon prior written notice to Seller, to assign, to one or more of its Affiliates, Buyer's rights under this Agreement to acquire the Assets at Closing; provided further, that without the prior written consent of the other Parties, no assignment pursuant to this Section 11.6 shall relieve any Party from its obligations under this Agreement.
- 11.7 Counterparts/Fax Signatures. This Agreement may be executed and delivered in one or more counterparts, each of which when executed and delivered shall be an original, and all of which when

executed shall constitute one and the same instrument. The exchange of copies of this Agreement and of signature pages by facsimile or by electronic image scan transmission in .pdf format shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes.

#### 11.8 Construction.

- (a) <u>Capitalized terms used but not otherwise defined in this Agreement have the meanings</u> given to such terms in Annex I.
- (b) The headings of the Articles and Sections of this Agreement are for guidance and convenience of reference only and shall not limit or otherwise affect any of the terms or provisions of this Agreement. The Disclosure Schedules are hereby incorporated in this Agreement by reference and constitute a part of this Agreement. The words "this Agreement," "herein," "hereby," "hereunder" and "hereof," and words of similar import, refer to this Agreement as a whole and not to any particular Article, Section, subsection or other subdivision unless expressly so limited. All references to "\$" or "dollars" shall be deemed references to United States dollars. References made in this Agreement, including use of a pronoun, shall be deemed to include where applicable, masculine, feminine, singular or plural, individuals or entities. The word "or" when used in a list shall not indicate that the listed items are exclusive of each other. The words "include," "includes," and "including" are deemed to be followed by "without limitation" whether or not they are in fact followed by such words or words of similar import. References to any Law, Contract, agreement or other instrument as it may be amended from time to time.
- 11.9 Entire Agreement. This Agreement and the Ancillary Agreements constitute the entire understanding among the Parties, their respective partners, members, trustees, shareholders, officers, directors and employees with respect to the subject matter hereof, superseding all negotiations, prior discussions and prior agreements and understandings relating to such subject matter, including any letter of intent entered into between the Parties.
- 11.10 Successors and Permitted Assigns. This Agreement shall be binding upon, and shall inure to the benefit of, the Parties, and their respective successors and permitted assigns.
- 11.11 Parties in Interest. Nothing in this Agreement shall entitle any Person, other than the Parties, or the Parties' respective related indemnified parties hereunder, to any claim, cause of action, remedy or right of any kind; <u>provided</u>, that only a Party will have the right to enforce the provisions of this Agreement on its own behalf or on behalf of its related indemnified parties (but shall not be obligated to do so).
- 11.12 Severability. In the event any term of this Agreement is found by any court to be void or otherwise unenforceable, the remainder of this Agreement shall remain valid and enforceable, and, to the fullest extent permitted by Law, such offending term or terms shall be replaced with an enforceable term or enforceable terms that as nearly as possible effect the Parties' intent.
- 11.13 Joint Preparation. The Parties acknowledge and agree that this Agreement was jointly prepared by the Parties, and that each Party was afforded the opportunity to consult with counsel in the preparation and negotiation of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties and no presumption or burden of proof shall arise favoring or disfavoring either Party by virtue of the authorship of any of the provisions of this Agreement.

11.14 Publicity. Seller and Buyer shall consult with each other with regards to all publicity and other releases concerning this Agreement and the transactions contemplated hereby and, except as required by applicable Law or the applicable rules or regulations of any Governmental Authority or stock exchange, neither Party shall issue any such publicity or other release without the prior written consent of the other Party hereto (such consent not to be unreasonably withheld, conditioned or delayed). Without the prior written consent of the other Party, neither Buyer nor Seller shall disclose or issue any press release or public announcement that includes the name of the non-disclosing Party; provided, however, the foregoing shall not restrict disclosures in order to comply with applicable securities or other laws, to comply with demands or requests from regulatory authorities, or to comply with existing loan or other agreements binding upon such Party.

11.15 Disclosure Schedules. The matters set forth on the disclosure schedules hereto (collectively, the "Disclosure Schedules") are not necessarily matters that Seller is required to disclose or matters that would constitute a breach of any representation or warranty had such matters not been disclosed. No disclosure by any Seller in the Exhibits or the Disclosure Schedules relating to any possible breach or violation of any contract or applicable Law shall be construed as an admission or indication that such breach or violation exists or has actually occurred.

11.16 Specific Performance. Subject to and without limiting Section 8.5, from and after Closing, if either Seller or Buyer violates or fails or refuses to perform any covenant or agreement made by Seller or Buyer (as applicable) in this Agreement, the non-breaching Party, subject to the terms of this Agreement and in addition to any remedy at Law for damages or other relief permitted under this Agreement, may institute and prosecute an action in any court of competent jurisdiction to enforce specific performance of such covenant or agreement or seek any other equitable relief, without the necessity of proving actual damages or posting of a bond. Each of Seller and Buyer hereby acknowledges and agrees that the rights of the other to consummate the transactions contemplated by this Agreement are special, unique and of extraordinary character and that, if any Seller or Buyer violates or fails or refuses to perform any covenant or agreement made by it in this Agreement, the non-breaching Party may be without an adequate remedy at Law.

11.17 Confidentiality. Subject to Section 11.14, for a period of eighteen (18) months after the Closing Date, Seller shall, and shall cause its Affiliates to, not use or make disclosures to third parties of any confidential or proprietary information relating to Assets except with the prior written consent of Buyer or as required by applicable Law, except to the extent that such information (a) is generally available to the public through no breach by such Seller or its Affiliates of this Section 11.17, (b) is lawfully acquired by such Seller or its Affiliates after the Closing from sources which are not actually known to such Seller to be prohibited from disclosing such information due to confidentiality obligations owed to Buyer, (c) is required for purposes of compliance by such Seller or its Affiliates with Tax or regulatory reporting requirements, (d) is needed in connection with the exercise, enforcement or performance of Seller's rights and obligations hereunder, the Ancillary Agreements or the enforcement, performance or defense of the same, or (e) is disclosed (i) in the course of any trial or other legal Proceeding involving Seller or its Affiliates, (ii) as required by any applicable securities Law or legal process or other Law (including any subpoena, interrogatory, or other similar requirement for such information to be disclosed) or rules of any applicable national stock exchange, or (iii) as is necessary to allow Seller or its Affiliates to manage or operate the Excluded Assets or Seller's or its Affiliates' other assets or Retained Obligations; provided, however, that (A) nothing shall prohibit such Seller or its Affiliates from using such information in the conduct of their respective businesses following the Closing so long as such use is consistent in all material respects with the manner in which such Seller or its Affiliates used such information prior to the Closing and (B) such Seller and its Affiliates may discuss the underlying investment with respect to the Assets and the acquisition or disposition of the Assets with its Representatives and current and potential partners or investors or in connection with legitimate fundraising activities or fund performance reporting with current or prospective investors, lenders or partners.

11.18 No Recourse. Except as provided in this Section 11.18, nothing express or implied in this Agreement or any document, agreement or instrument delivered in relation to this Agreement is intended or shall be construed to confer upon or give any Person, other than the Parties, any right or remedies under or by reason of this Agreement. All claims, obligations, liabilities or causes of action (whether in contract or in tort, in Law or in equity or granted by statute) that may be based upon, in respect of, arise under, out or by reason of, be connected with or relate in any manner to this Agreement, or the negotiation, execution or performance of this Agreement (including any representation or warranty made in, in connection with or as an inducement to, this Agreement) and the transactions contemplated hereby, may be made only against (and such representations and warranties are those solely of) the Parties (the "Contracting Parties"). No Person who is not a Contracting Party, including any past, present or future director, officer, employee, incorporator, member, partner, manager, equityholder, Affiliate, agent, attorney, representative or assignee of, and any financial advisor or lender to, any Contracting Party, or any past, present or future director, officer, employee, incorporator, member, partner, manager, equityholder, Affiliate, agent, attorney, representative or assignee of, and any financial advisor or lender to, any of the foregoing (collectively, the "Nonparty Affiliates"), shall have any liability (whether in contract or in tort, in law or in equity, or granted by statute) for any claims, causes of action, obligations or liabilities arising under, out of, in connection with or related in any manner to this Agreement or the transactions contemplated hereby or based on, in respect of or by reason of this Agreement or its negotiation, execution, performance or breach of this Agreement and the transactions contemplated hereby, and, to the maximum extent permitted by Law, each Contracting Party hereby waives and releases all such liabilities, claims, causes of action and obligations against any such Nonparty Affiliates. Without limiting the foregoing, to the maximum extent permitted by Law, each Contracting Party disclaims any reliance upon any Nonparty Affiliates with respect to the performance of this Agreement or any representation or warranty made in connection with, or as an inducement to, this Agreement. This Section 11.18 shall not be effective as to any Seller to the extent such Seller makes any distribution to its members, including of any portion of the Purchase Price, that would violate any provision of Subchapter VIII of the Delaware Limited Liability Company Act or similar Law.

[Signature Pages to Follow]

IN WITNESS WHEREOF, the Parties have duly executed this Agreement as of the Execution Date.

## **SELLER:**

# Palmetto Investment Partners II, LLC

By: /s/ Ryan Turner
Name: Ryan Turner
Title: Managing Person

By: /s/ William T. Fennebresque

Name: William T. Fennebresque

Title: Managing Person

## **BUYER:**

# **PHX Minerals Inc.**

By: /s/ Chad L. Stephens III

Name: Chad L. Stephens III
Title: President and Chief Executive Officer

#### **ANNEX I**

#### **DEFINED TERMS**

- "AAA" has the meaning set forth in Section 3.2(g)(iv).
- "Accounting Consultant" has the meaning set forth in Section 2.4(c).
- "Acquisition Proposals" has the meaning set forth in Section 6.6.
- "Actual NRA" means the actual Net Royalty Acres determined by the Parties or the Title Consultant, as the case may be, to be owned by Seller in each Tract pursuant to Article 3.
- "Affiliate" means, with respect to any Person, any other Person that directly or indirectly controls, is controlled by or is under common control with, such first Person. For the purposes of this definition, "control" (including, with correlative meanings, the terms "controlling," "controlled by" and "under common control with"), as applied to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of that Person, whether through the ownership of voting securities, by contract or otherwise.
  - "Agreement" has the meaning set forth in the introductory paragraph of this Agreement.
  - "Allocated Value" has the meaning set forth in Section 2.2.
  - "Ancillary Agreements" means the Conveyance and the FIRPTA Certificate.
  - "Arbitration Commencement Date" has the meaning set forth in Section 11.3(b).
- "Asset Taxes" means ad valorem, property, excise, severance, production and similar Taxes based upon the operation or ownership of the Assets or the production of Hydrocarbons or the receipt of proceeds therefrom, but excluding, for the avoidance of doubt, Income Taxes and Transfer Taxes.
- "Assets" means, save, except for and excluding the Excluded Assets, all of Seller's collective right, title and interest in and to the following assets and properties from and after the Effective Time:
- all fee mineral interests and lessor royalty interests in all oil, natural gas, coalbed methane and other liquid or gaseous Hydrocarbons, as well as their respective constituent products (including condensate, casinghead gas, distillate and natural gas liquids), and any other minerals, similar or dissimilar, customarily produced in conjunction therewith (all such substances are defined for purposes of this Agreement as a "Mineral" or the "Minerals"), in, on and under or produced and saved from the lands, tracts and properties described in Exhibit A-1 attached hereto (such lands, tracts, and properties described in Exhibit A-1, the "Lands"), and from all wells located on the Lands (collectively, the "Wells"), together with any royalty interests attributable thereto and any units, lands, tracts or other properties pooled with any of the Lands (collectively, the "Mineral Interests");
- (b) any overriding royalty interests held by Seller burdening oil, gas or other minerals produced, saved or sold from the Lands, including those described in <u>Exhibit A-1</u> attached hereto (collectively, the "Overriding Royalty Interests");

- (c) the non-participating royalty interests burdening the Lands, including those described in <u>Exhibit A-1</u> attached hereto (collectively, the "**Royalty Interests**", together with the Mineral Interests and Overriding Royalty Interests, the "**Interests**");
- (d) any oil, gas and mineral leases that relate to the Interests, including all reversionary rights thereunder, including those described in <u>Exhibit D</u>, which <u>Exhibit D</u> shall be provided by Seller to Buyer, based on information reasonably available in Seller's files, within five (5) calendar days from the Execution Date (collectively, the "**Leases**");
- (e) all division and transfer orders, and, except for the Leases, all written contracts, contractual rights, interests and other written agreements to which any Seller is a party (including as assignee) and under which any of the Interests are bound (collectively, the "Contracts");
  - (f) all Records, except for the Excluded Records;
- (g) (i) all rights of ingress and egress, at all times for the purpose of exploring, drilling, operating for and producing from the Lands for Minerals and removing the same therefrom, and of laying pipelines, storing oil, building tanks, processing and treating plants and facilities, power stations, telephone lines, roads and other structures necessary to produce, save, care for, treat, store, compress, process, and transport said products, attributable to the conveyed interests, and (ii) all rights with respect to the use and occupation of the surface of the Lands and the subsurface depths under the Lands for the sole purpose of development and maintenance of the Interests; and (iii) the right of ingress and egress across any lands owned or controlled by Seller to the extent such ingress and egress across such lands is reasonable to obtain access for the development and maintenance of the Interests;
- (h) all proceeds attributable to any of the Interests or the Leases and attributable to the time period after the Effective Time;
- (i) to the extent transferable (provided that Seller will not be required to pay any fees or incur any liabilities in order to transfer such data), all proprietary and non-proprietary geologic, geophysical and seismic data that relate to the Interests (excluding any interpretive data);
- (j) all permits to the extent primarily relating to or applicable to any of the Interests or Leases and (i) required for the ownership of the Interests or Leases and (ii) transferable pursuant to applicable Law;
- (k) to the extent relating to the Assets and to the extent assignable, all rights and interests of either Seller (i) under any agreement or policy of indemnity or insurance (including any rights, claims or causes of action of either Seller against third parties under any indemnity or hold harmless agreements) and any indemnities received in connection with Seller's prior acquisition of any of the Assets, but in each case only to the extent that such rights or interests arise from obligations or liabilities for which Buyer is responsible under this Agreement, or (ii) relating to claims and causes of action that may be asserted against a third party to the extent such rights and claims arise from obligations or liabilities assumed by Buyer hereunder; and
- (l) except to the extent pertaining to any Retained Obligations, and only to the extent assignable, all rights, claims and causes of action to the extent, and only to the extent, that such rights, claims or causes of action are associated with any of the other Assets and relate to the period from and after the Effective Time.

"Assumed Liabilities" has the meaning set forth in Section 10.1.

- "Business Day" means any weekday other than a weekday on which commercial banks in the city of Oklahoma City, Oklahoma and Dallas, Texas are authorized or required by applicable Law to be closed.
  - "Buyer" has the meaning set forth in the introductory paragraph of this Agreement.
  - "Buyer Fundamental Representations" has the meaning set forth in <u>Section 10.3</u>.
  - "Buyer Indemnitees" has the meaning set forth in Section 10.2(a).
  - "Buyer Material Adverse Effect" means a Material Adverse Effect with respect to Buyer.
  - "Buyer Officer Certificate" has the meaning set forth in Section 8.1(c).
  - "Closing" has the meaning set forth in <u>Section 9.1</u>.
  - "Closing Date" has the meaning set forth in Section 9.1.
  - "Code" has the meaning set forth in Section 4.6.
  - "Concurrent PSA" has the meaning set forth in Section 8.1(f).
  - "Contracting Party" has the meaning set forth in <u>Section 11.18</u>.
  - "Contracts" has the meaning set forth in the definition of "Assets."
  - "Conveyance" has the meaning set forth in Section 9.2(b).
  - "Credit Deficiency Notice" has the meaning set forth in Section 3.2(b).
  - "Credit Notice" has the meaning set forth in Section 3.2(b).
  - "Cure Notice" has the meaning set forth in Section 3.2(f).
  - "Cure Period" has the meaning set forth in Section 3.2(f).
  - "Deductible" has the meaning set forth in Section 10.2(c)(ii).
  - "**Defect Allegation**" has the meaning set forth in Section 3.2(a).
- "Defensible Title" means such title of Seller to the Assets that, as of the Effective Time and as of the Closing Date, is deducible of record or provable by documentation that would be successfully defended if challenged and that, subject to the Permitted Encumbrances, (a) as to each Tract, entitles Seller to not less than the Stated NRA for such Tract and (b) is free and clear of all Encumbrances.
  - "**Deficiency Notice**" has the meaning set forth in Section 3.2(a).
  - "Delayed Closing Date" has the meaning set forth in Section 3.2(f).
  - "Disclosing Credit Party" has the meaning set forth in Section 3.2(b).
  - "Disclosure Schedules" has the meaning set forth in Section 11.15.

"Dispute Notice" has the meaning set forth in Section 2.4(a).

"Effective Time" means, as to each Asset subject to this Agreement, the same date and time that was deemed the "Effective Time" (or such similar definition) in the applicable acquisition agreement or other instrument of conveyance through which the applicable Seller acquired its interest in, and derives its title to, such Asset (each, a "Title Instrument"), as specified in the applicable Title Instrument. "Encumbrance" means any lien, judgment, mortgage, deed of trust, security instrument, pledge or option that is binding upon the Assets, or other preferential arrangement or other similar encumbrance.

"Environmental Laws" means the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. § 9601 et seq.; the Resource Conservation and Recovery Act, 42 U.S.C. § 6901 et seq.; the Federal Water Pollution Control Act, 33 U.S.C. § 1251 et seq.; the Clean Air Act, 42 U.S.C. § 7401 et seq.; the Hazardous Materials Transportation Act, 49 U.S.C. § 5101 et seq.; the Toxic Substances Control Act, 15 U.S.C. §§ 2601 through 2629; the Oil Pollution Act, 33 U.S.C. § 2701 et seq.; the Emergency Planning and Community Right to Know Act, 42 U.S.C. § 11001 et seq.; and the Safe Drinking Water Act, 42 U.S.C. §§ 300f through 300j, and all other applicable Laws of any Governmental Authority having jurisdiction over the property in question addressing (a) pollution or pollution control; (b) protection of natural resources, the environment or biological resources; or (c) the disposal or release or threat of release of hazardous substances, in each case as enacted and in effect on or prior to the Closing Date.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

## "Excluded Assets" means the following assets and properties:

- (a) all proceeds attributable to any of the Assets to the extent attributable to the time period before the Effective Time, including any lease bonuses, royalties and other proceeds and benefits to the extent (i) attributable to the time period before the Effective Time and (ii) received no later than one (1) year following the Closing Date;
- (b) all claims of Seller or any of its Affiliates for refunds of, rights to receive funds from any Governmental Authority, or loss carry forwards or credits with respect to Seller Taxes;
- (c) any refunds of costs or other expenses borne by any Seller or its predecessors in title, insofar as such refunds relate to time periods prior to the Effective Time or to any Retained Obligations;
  - (d) the Excluded Records;
- (e) all claims and causes of action of Seller relating to (i) any Retained Obligation or (ii) production that occurred prior to the Effective Time, and all audit rights arising under any Contracts with respect to the time periods prior to the Effective Time or any Retained Obligations; provided, however, if any third party makes a claim regarding time periods prior to the Effective Time that is or would be an Assumed Liability, then the claims and causes of action against such third party, to the extent relating to such Assumed Liability, that would otherwise be Excluded Assets pursuant to this clause (e), shall become Assets;
- (f) all computer or communications software owned, licensed or used by Seller, other than the Records;
- (g) any logo, service mark, copyright, trade name or trademark of or associated with Seller or any Affiliate of Seller or any business of Seller or of any Affiliate of Seller;
  - (h) motor vehicles and other rolling stock owned by Seller; and

(i) all oil, gas and Hydrocarbons produced, saved or sold from the Assets with respect to all periods prior to the Effective Time.

"Excluded Records" means (a) any files, records, or data (other than title opinions or memoranda relating to the Assets), which Seller believe in good faith are protected by attorney-client privilege or confidentiality agreements, provided that Seller have used Reasonable Efforts to obtain waivers of such confidentiality agreements, (b) all of Seller's corporate minute books, corporate financial records and Income Tax records that relate to Seller's business generally, (c) all files, records or data relating to the marketing process for the Assets including any and all bids received in connection therewith, and (d) (i) Seller's interpretation of any geologic, geophysical and seismic data, and (ii) any non-proprietary geologic, geophysical and seismic data that relate to the Interests, to the extent that (A) such non-proprietary geologic, geophysical and seismic data are not transferable without consideration or without a third party's consent, and (B) such consent was not obtained from such third party or Buyer has not agreed in writing to pay such consideration prior to the time of Closing.

"Execution Date" has the meaning set forth in the introductory paragraph of this Agreement.

"Final Price" has the meaning set forth in Section 2.4(a).

"Final Settlement Date" has the meaning set forth in Section 2.4(a)

"Final Settlement Statement" has the meaning set forth in Section 2.4(a).

"FIRPTA Certificate" has the meaning set forth in Section 9.2(e).

"Fraud" means an intentional and knowing misrepresentation of fact with the intent to deceive solely with respect to the making of the representations and warranties in this Agreement or any instrument or other agreement entered into in connection herewith which constitutes common law fraud under Delaware law.

"GAAP" means United States generally accepted accounting principles as in effect from time to time, consistently applied.

"Governmental Authority" means any federal, state, local, municipal, tribal or other government; any governmental, regulatory or administrative agency, commission, body or other authority exercising or entitled to exercise any administrative, executive, judicial, legislative, regulatory or taxing authority or power, and any court or governmental tribunal, including any tribal authority having or asserting jurisdiction.

"Hazardous Substances" means any pollutants, contaminants, toxins or hazardous or extremely hazardous substances, materials, wastes, constituents, compounds or chemicals that are regulated by, or will, given extent levels, form the basis of remediation liability under, any Environmental Laws (including NORM and other substances) in effect as of the Execution Date and each Closing Date.

"Hydrocarbons" means any oil, gas, casinghead gas, condensate, distillate or other liquid or gaseous hydrocarbon of every kind or description.

"Income Taxes" means any U.S. federal, state or local or non-U.S. income Tax or Tax based on profits, net profits, margin revenues or similar Taxes (including franchise Taxes and any capital gains and net worth Taxes).

"Individual Claim Threshold" has the meaning set forth in Section 10.2(c)(ii).

"Interests" has the meaning set forth in the definition of "Assets."

"**Knowledge**" means the actual knowledge, without any obligation of inquiry, of (x) in the case of Seller, of Matt Veazey or (y) in the case of Buyer, of Ralph D'Amico.

"Lands" has the meaning set forth in the definition of "Assets."

"Law" means any applicable statute, law, rule, regulation, ordinance, order, code, ruling, writ, injunction, decree or other official act of or by any Governmental Authority.

"Leases" has the meaning set forth in the definition of "Assets."

"Losses" means any and all losses, costs, expenses (including court costs, reasonable fees and expenses of attorneys, technical experts and expert witnesses and the cost of investigation), liabilities, damages, diminution of value of any Asset, demands, suits, claims, causes of action and sanctions of every kind and character (including civil fines); <u>provided, however,</u> for the avoidance of doubt, any calculation or determination of Losses shall not include any measure of damages or theories of liability expressly waived by the Parties pursuant to <u>Section 10.7</u>.

"Material Adverse Effect" means, with respect to a Party, any circumstance, change or effect that has had or would be reasonably likely to (a) have, individually or in the aggregate with any other circumstance, change or effect, a material and adverse effect on the ownership or financial condition of the Assets in excess of an amount equal to fifteen percent (15%) of the Purchase Price (with respect to a Seller Material Adverse Effect) or the assets and properties of Buyer (with respect to a Buyer Material Adverse Effect), taken as a whole or (b) materially impair, prevent or delay such Party's timely consummation of the transactions contemplated hereby, but shall exclude any circumstance, change or effect resulting or arising from: (i) any change in general conditions in the industries or markets in which such Party operates; (ii) seasonal reductions in revenues or earnings of such Party in the ordinary course of its business; (iii) any adverse change, event or effect on the global, national or regional energy industry as a whole, including any such change to energy prices or the value of oil and gas assets and properties or other commodities, goods or services, or the availability or costs of hedges; (c) national or international political conditions, including any engagement in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack; (d) changes in GAAP or the interpretation thereof; (e) the entry into or announcement of this Agreement, actions contemplated by this Agreement or the consummation of the transactions contemplated hereby; (f) changes or developments in financial or securities markets or the economy in general; (g) the outbreak or continuation of any disease or epidemic (including COVID-19); (h) natural declines in well performance or reclassification or recalculation of reserves in the ordinary course of business consistent with ordinary, prudent and customary practices in the oil and natural gas exploration and production industry; (i) acts or failures to act of any Governmental Authorities and changes in Law or the interpretation thereof from and after the Execution Date; (j) effects of weather, meteorological events, natural disasters or other acts of God; or (k) any title defects, title losses or encumbrances on title; provided that the exclusions set forth in subsections (b), (d), (f), (g) and (j) shall be effective only insofar as such exclusions do not disproportionately affect the assets and properties of the affected Party as compared to similarly situated oil and gas assets or similarly situated owners of oil and gas assets.

- "Material Contracts" means any of the following Contracts to which any Seller is a party:
- (a) Contracts restricting in any material respect the owner of the Assets from freely engaging in any business or competing anywhere;
- (b) each Contract that burdens any of the Assets and is an indenture, mortgage, loan, credit agreement, sale-leaseback, guaranty of any obligation, bond, letter of credit, or similar financial Contract (other than Permitted Encumbrances) that will not be terminated at or before the Closing;
- (c) any Contract that burdens any of the Assets, covering any futures, hedges, swap, collar, put, call, floor, cap, option or the like that is intended to reduce or eliminate the fluctuations in the prices of commodities, currency, exchange rates or interest rates that will not be terminated at or before the Closing;
  - (d) any Contract the primary purpose of which is to indemnify another Person;
  - (e) Contracts to sell, exchange or otherwise dispose of all or any part of the Assets;
- (f) Contracts involving obligations of, or payments to or from, the owner of the Assets that, in each case, can reasonably be expected to result in aggregate payments by Seller in excess of Fifty Thousand Dollars (\$50,000.00) in the current or any subsequent fiscal year; and
- (g) Contracts between any Seller and any Affiliate of such Seller that will be binding on the Assets after the Closing.
  - "Mineral Interests" has the meaning set forth in the definition of "Assets."
  - "Minerals" has the meaning set forth in the definition of "Assets."
- "Net Mineral Acres" means with regard to an Interest, (a) the number of gross surface acres included in such Interest, <u>multiplied</u> by (b) Seller's percentage ownership interest in the oil and gas fee mineral interests beneath such gross surface acres; <u>provided</u>, <u>however</u>, that if Seller's Net Mineral Acres vary for different formations or areas within a Tract, a separate calculation shall be performed with respect to each such formation or area for the purposes of calculating Net Mineral Acres under this Agreement.
- "Net Royalty Acres" means, with regard to the Interest, the number equal to (a) the number of Net Mineral Acres attributable to such Interest <u>multiplied</u> by (b) the applicable Royalty Rate for such Interest <u>multiplied</u> by (c) eight (8); <u>provided</u>, <u>however</u>, that if an Interest varies as to different target formations or areas within a Tract, a separate calculation shall be performed with respect to each such formation or area for the purposes of calculating Net Royalty Acres under this Agreement. For any unleased tracts, a twelve and one-half percent (12.5%) Royalty Rate shall be used as the applicable Royalty Rate.
  - "Nonparty Affiliates" has the meaning set forth in Section 11.18.
  - "Obligations" has the meaning set forth in Section 10.1.
  - "Officer Certificates" has the meaning set forth in Section 8.2(c).
  - "Overriding Royalty Interests" has the meaning set forth in the definition of "Assets."
  - "Palmetto" has the meaning set forth in the introductory paragraph of this Agreement.
  - "Parties" has the meaning set forth in the introductory paragraph of this Agreement.

"Party" has the meaning set forth in the introductory paragraph of this Agreement.

#### "Permitted Encumbrances" means:

- (a) the terms and provisions of any Contract or Lease of record to the extent that such terms and provisions do not, individually and in the aggregate, operate to reduce Seller's Net Royalty Acres in any Tract below the Stated NRA for such Tract;
- (b) lessors' royalties, non-participating royalties and similar burdens of record that do not, individually or in the aggregate, operate to reduce Seller's Net Royalty Acres in any Tract below the Stated NRA for such Tract;
- (c) preferential rights to purchase and required third party consents to assignment of any Asset that are not applicable to the transactions contemplated by this Agreement;
- (d) rights to consent by, required notices to, filings with or other actions by Governmental Authorities or any other Person in connection with the sale or conveyance of any of the Assets if the same are customarily obtained in the ordinary course of business subsequent to such sale or conveyance;
- (e) any materialman's, mechanics', repairman's, employees', contractors', operators' liens or other similar liens or charges for liquidated amounts arising in the ordinary course of business for amounts not yet delinquent or, if delinquent, that are being contested in good faith, or which will be released prior to Closing;
- (f) any liens for Taxes and assessments not yet delinquent or, if delinquent, that are being contested in good faith in the ordinary course of business and for which Seller has agreed to pay under the terms of this Agreement or which have been prorated under the terms of this Agreement;
- (g) all rights reserved to or vested in any Governmental Authority to control or regulate the Assets in any manner;
  - (h) the lack of executive rights in any of the Lands;
- (i) defects or irregularities of title as to which the relevant statute of limitation or prescription would bar any attack or claim against Seller's title;
- (j) defects in the chain of title arising from the failure to recite marital status, omissions of successors or heirship, or the lack of probate proceedings and defects arising out of lack of corporate or other entity authorization, absent reasonable evidence that the defect results in a third party's actual and valid claim of title to the affected Interest:
- (k) any defects based solely on (i) lack of information, including lack of information in Seller's files, the lack of third party records or unavailability of information from applicable Governmental Authorities, absent reasonable evidence that the defect results in a third party's actual and valid claim of title to the affected Interest or (ii) the absence of certain documents in Seller's files that are referenced by other documents that are located in Seller's files;
- (l) division orders and sales contracts terminable without penalty upon no more than ninety (90) days' notice to the purchaser;

- (m) defects or irregularities arising out of lack of corporate authorization or a variation in corporate name, absent affirmative evidence that such corporate action was not authorized and results in another Person's superior claim of title to the relevant Assets;
  - (n) any matter waived in writing by Buyer;
- (o) defects arising out of a scrivener's error that would not reasonably be expected to adversely affect title to the Assets;
- (p) defects based on a gap in the chain of title of the Asset if such gap exists prior to January 1, 1990, unless such gap is shown to exist in the county or parish records by an abstract of title, title opinion or landman's title chain or runsheet;
- (q) defects resulting from lack of survey, unless a survey is expressly required by the underlying lease or applicable Law; and
- (r) defects based on failure to record releases of liens, production payments or mortgages that have expired by their own terms or the enforcement of which are barred by applicable statutes of limitation; in each case, unless Buyer provided evidence that such defects or irregularities may result in another Person's actual and superior claim of title.
- "Person" means any natural person, corporation, partnership, trust, limited liability company, court, agency, government, board, commission, estate or other entity or authority.
  - "Preliminary Settlement Statement" has the meaning set forth in Section 2.3(a).
- "**Proceeding**" means any proceeding, action, arbitration, litigation, subpoena, or suit (whether civil, criminal, administrative, investigative, or informal) commenced, brought, conducted, or heard by or before, or otherwise involving, any Governmental Authority or arbitrator.
  - "Purchase Price" has the meaning set forth in Section 2.1.
- "Reasonable Efforts" means a Party's commercially reasonable efforts in accordance with reasonable commercial practice.
- "Records" means copies of all files, records, and data in any Seller's possession or control to the extent relating to any of the Assets, including deed, mineral and Lease records, assignments, division order records, title records (including abstracts of title, title opinions and memoranda and title curative documents), Contracts, electronic data files (if any), maps, production records, decline curves and graphical production curves and financial, accounting and Asset Tax records, but excluding all Excluded Records.
- "Representatives" means with respect to any Person, its officers, directors, managers, employees, consultants, agents, financial advisors, attorneys, accountants and other representatives.
- "Retained Obligations" means all Obligations arising from or related to the following: (a) the ownership of the Assets prior to the Effective Time, (b) the amount of all Taxes for which Seller are responsible pursuant to Article 7, (c) the Excluded Assets, and (d) general and administrative expenses (as such term is understood in accordance with GAAP) of the Seller.
  - "Review Period" has the meaning set forth in Section 2.4(a).
  - "Royalty Interests" has the meaning set forth in the definition of "Assets."

- "Royalty Rate" means the applicable royalty interest (a) set forth in the applicable Lease, with respect to any Mineral Interest, minus the proportionate share, if any, of any non-participating royalty interests that burden such Mineral Interest or (b) carved out of the lessee's working interest derived from such Lease, with respect to any Overriding Royalty Interest.
  - "Scheduled Closing Date" has the meaning set forth in <u>Section 9.1</u>.
  - "Securities Act" means the Securities Act of 1933, as amended.
  - "Seller" has the meaning set forth in the introductory paragraph of this Agreement.
  - "Seller Fundamental Representations" has the meaning set forth in Section 10.3.
  - "Seller Indemnitees" has the meaning set forth in Section 10.2(b).
  - "Seller Material Adverse Effect" means a Material Adverse Effect with respect to Seller.
  - "Seller Officer Certificate" has the meaning set forth in Section 8.2(c).
- **"Seller Taxes"** shall mean (a) Income Taxes imposed by applicable Law on Seller and (b) Asset Taxes allocable to Seller pursuant to <u>Article 7</u> (taking into account, and without duplication of, such Asset Taxes effectively borne by Seller as a result of (i) the adjustments to the Purchase Price made pursuant to <u>Section 2.3</u>, and (ii) any payments made from one Party to the other in respect of Asset Taxes pursuant to <u>Article 7</u>).
  - "Seller" has the meaning set forth in the introductory paragraph of this Agreement.
  - "Special Warranty" has the meaning set forth in Section 6.5(c).
  - "Stated NRA" has the meaning set forth in Section 3.2(c)(iii).
  - "Straddle Period" means any taxable period beginning before and ending after the Effective Time.
  - "Subject Schedule Supplement" has the meaning set forth in Section 6.3.
  - "Survival Period" has the meaning set forth in Section 10.3.
- "Tax Returns" means any return, declaration, report, claim for refund, or information return or statement relating to Taxes filed or required to be filed with any Governmental Authority, including any schedules or attachment thereto, and including any amendment thereof.
- "Taxes" means (a) all taxes, assessments, fees and other charges in the nature of a tax imposed by any Governmental Authority, including any federal, state, local or foreign income tax, surtax, remittance tax, presumptive tax, net worth tax, special contribution tax, production tax, severance tax, value added tax, withholding tax, gross receipts tax, profits tax, ad valorem tax, personal property tax, real property tax, sales tax, goods and services tax, transfer tax, use tax, excise tax, franchise tax, occupation tax, payroll tax, employment tax, unemployment tax, disability tax, alternative or add-on minimum tax and estimated tax, and (b) any interest, fine, penalty or additions to tax imposed by a Governmental Authority in connection with any item described in clause (a).
  - "Title Claims Date" has the meaning set forth in Section 3.1.

"Title Consultant" has the meaning set forth in Section 3.2(g)(iv).

"**Title Credit**" means any right, circumstance or condition that operates to increase Seller's Actual NRA in any Tract above the Stated NRA set forth on <u>Exhibit A-1</u> for such Tract.

"Title Credit Amount" has the meaning set forth in Section 3.2(e).

"Title Deductible" has the meaning set forth in Section 3.2(d)(ii).

"Title Defect" means (a) any Encumbrance, defect or other matter that causes (or if not cured, could reasonably be expected to cause) Seller not to have Defensible Title to any of the Assets; or (b) any default by Seller under a Lease or Contract or other agreement that (i) has an adverse effect on the operation, value or use of any Asset, (ii) prevents Seller from receiving the proceeds of production attributable to Seller's interest in any Asset, or (iii) results in cancellation or impairment of Seller's interest in any Asset.

"Title Defect Amount" has the meaning set forth in Section 3.2(c).

"Title Dispute" has the meaning set forth in Section 3.2(g)(i).

"Title Instrument" has the meaning set forth in the definition of "Effective Time".

"Title Threshold" has the meaning set forth in Section 3.2(d)(i).

"Tract" has the meaning set forth in Section 2.2.

"Transfer Taxes" means, collectively, transfer, sales, use, excise, documentary, stamp, gross receipts, goods and services, registration or other similar Taxes relating to the transactions contemplated by this Agreement.

"Wells" has the meaning set forth in the definition of "Assets."

# **EXHIBIT A-1 - TRACTS AND ALLOCATED VALUES**

The following Assets in Caddo Parish, LA:

Tract	Seller	Section	Township-Range	NMA	Royalty	NRA
T-0082179	Palmetto	4	17N-15W	19.70	0.22	34.67
T-0082190	Palmetto	4	17N-15W	0.84	0.22	1.48
T-0082188	Palmetto	4	17N-15W	0.84	0.22	1.48
T-0082165	Palmetto	4	17N-15W	1.94	0.20	3.11
T-0086271	Palmetto	9	17N-15W	1.19	0.20	1.90
T-0086222	Palmetto	9	17N-15W	1.00	0.20	1.61
Total				25.52		44.25

API	Wells	Operator	Status
		ENSIGHT IV ENERGY MGMT	
1701734912	MAYFIELD 9 H; HA RA SUY #1	LLC	PDP
1701736213	PINEHILLS 9-4 HC; HA RA SUY #1-ALT	BLUE DOME OPERATING LLC	PDP
1701736434	PINEHILLS 9-4 HC; HA RA SUY #2-ALT	BLUE DOME OPERATING LLC	DUC
1701736435	PINEHILLS 9-4 HC; HA RA SUY #3-ALT	BLUE DOME OPERATING LLC	DUC
1701736436	PINEHILLS 9-4 HC; HA RA SUY #4-ALT	BLUE DOME OPERATING LLC	DUC
1701736437	PINEHILLS 9-4 HC; HA RA SUY #5-ALT	BLUE DOME OPERATING LLC	DUC

#### **EXHIBIT B**

#### FORM OF CONVEYANCE

# DEED, ASSIGNMENT AND BILL OF SALE

(Special Warranty Deed)

STATE OF LOUISIANA
PARISH OF CADDO

This Deed, Assignment and Bill of Sale (this "Deed") is dated as of the acknowledgments below but is effective for all purposes as to each Tract (as defined below) as of the Effective Date of each respective instrument referenced on <a href="Exhibit A-1">Exhibit A-1</a> attached hereto (the "Effective Time"), from [SELLER], a Delaware limited liability company ("Grantor"), having its principal address at 3811 Turtle Creek Blvd., Suite 1100, Dallas, Texas 75219, to PHX Minerals Inc., an Oklahoma corporation ("Grantee") having its principal address at 1601 NW Expressway, Valliance Bank Tower, Suite 1100, Oklahoma City, Oklahoma 73118. Grantor and Grantee are sometimes referred to herein individually as a "Party" and collectively as the "Parties".

Grantor, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and subject to the terms set forth herein, does by these presents grant, bargain, sell, convey, assign, transfer, set over, and deliver unto Grantee all of Grantor's rights, title, and interest in, to, and under the following (collectively, the "Assets"), except to the extent constituting Excluded Assets, as hereinafter defined:

- (a) all fee mineral interests and lessor royalty interests in all oil, natural gas, coalbed methane and other liquid or gaseous Hydrocarbons, as well as their respective constituent products (including condensate, casinghead gas, distillate and natural gas liquids), and any other minerals, similar or dissimilar, customarily produced in conjunction therewith (all such substances are defined for purposes of this Deed as a "Mineral" or the "Minerals"), in, on and under or produced and saved from each of the respective tracts of land and properties conveyed to Grantor by the respective instruments (each, individually a "Tract") listed and described in Exhibit A-1 attached hereto (such lands, Tracts, and properties described in Exhibit A-1, the "Lands"), together with any royalty interests attributable thereto and any units, lands, tracts, mineral servitudes, or other properties pooled with any of the Lands (collectively, the "Mineral Interests");
- (b) any overriding royalty interests held by Grantor burdening oil, gas or other minerals produced, saved or sold from the Lands, including those described in <u>Exhibit A-1</u> attached hereto (collectively, the "*Overriding Royalty Interests*");
- (c) the non-participating royalty interests burdening the Lands, including those described in Exhibit A-1 attached hereto (collectively, the "*Royalty Interests*", together with the Mineral Interests and Overriding Royalty Interests, the "*Interests*");

- (d) any oil, gas and mineral leases that relate to the Interests, including all reversionary rights thereunder, including those described in <u>Exhibit A-1</u> attached hereto (collectively, the "*Leases*");
- (e) all division and transfer orders, and, except for the Leases, all written contracts, contractual rights, interests and other written agreements to which Grantor is a party (including as assignee) and under which any of the Interests are bound (collectively, the "Contracts");
  - (f) all Records, except for the Excluded Records;
- (g) (i) all rights of ingress and egress, at all times for the purpose of exploring, drilling, operating for and producing from the Lands for Minerals and removing the same therefrom, and of laying pipelines, storing oil, building tanks, processing and treating plants and facilities, power stations, telephone lines, roads and other structures necessary to produce, save, care for, treat, store, compress, process, and transport said products, attributable to the conveyed interests, and (ii) all rights with respect to the use and occupation of the surface of the Lands and the subsurface depths under the Lands for the sole purpose of development and maintenance of the Interests; and (iii) the right of ingress and egress across any lands owned or controlled by Grantor to the extent such ingress and egress across such lands is reasonable to obtain access for the development and maintenance of the Interests;
- (h) all proceeds attributable to any of the Interests or the Leases and attributable to the time period after the Effective Time;
- (i) to the extent transferable (*provided* that Grantor will not be required to pay any fees or incur any liabilities in order to transfer such data), all proprietary and non-proprietary geologic, geophysical and seismic data that relate to the Interests (excluding any interpretive data);
- (j) all permits to the extent primarily relating to or applicable to any of the Interests or Leases and (i) required for the ownership of the Interests or Leases and (ii) transferable pursuant to applicable Law;
- (k) to the extent relating to the Assets and to the extent assignable, all rights and interests of either Grantor (i) under any agreement or policy of indemnity or insurance (including any rights, claims or causes of action of either Grantor against third parties under any indemnity or hold harmless agreements) and any indemnities received in connection with Grantor's prior acquisition of any of the Assets, but in each case only to the extent that such rights or interests arise from obligations or liabilities for which Grantee is responsible under this Deed, or (ii) relating to claims and causes of action that may be asserted against a third party to the extent such rights and claims arise from obligations or liabilities assumed by Grantee hereunder; and
- (l) except to the extent pertaining to any Retained Obligations, and only to the extent assignable, all rights, claims and causes of action to the extent, and only to the extent, that such rights, claims or causes of action are associated with any of the other Assets and relate to the period from and after the Effective Time.

SAVING, EXCEPTING AND RESERVING to Grantor, however, all of the following assets (the "*Excluded Assets*"):

(a) all proceeds attributable to any of the Assets to the extent attributable to the time period before the Effective Time, including any lease bonuses, royalties and other proceeds and benefits to the extent (i) attributable to the time period before the Effective Time and (ii) received no later than one year following the Closing Date;

- (b) all claims of Grantor or any of their Affiliates for refunds of, rights to receive funds from any Governmental Authority, or loss carry forwards or credits with respect to Seller Taxes;
- (c) any refunds of costs or other expenses borne by Grantor or its predecessors in title, insofar as such refunds relate to time periods prior to the Effective Time or to any Retained Obligations;
  - (d) the Excluded Records;
- (e) all claims and causes of action of Grantor relating to (i) any Retained Obligation or (ii) production that occurred prior to the Effective Time, and all audit rights arising under any Contracts with respect to the time periods prior to the Effective Time or any Retained Obligations; *provided, however*, if any third party makes a claim regarding time periods prior to the Effective Time that is or would be an Assumed Liability, then the claims and causes of action against such third party, to the extent relating to such Assumed Liability, that would otherwise be Excluded Assets pursuant to this clause (e), shall become Assets;
- (f) all computer or communications software owned, licensed or used by Grantor, other than the Records;
- (g) any logo, service mark, copyright, trade name or trademark of or associated with Grantor or any Affiliate of Grantor or any business of Grantor or of any Affiliate of Grantor;
  - (h) motor vehicles and other rolling stock owned by Grantor; and
- (i) all oil, gas and Hydrocarbons produced, saved or sold from the Assets with respect to all periods prior to the Effective Time.

TO HAVE AND TO HOLD all and singular the Assets, together with all rights, titles, interests, estates, remedies, powers and privileges thereto appertaining unto Grantee and its successors, legal representatives, and assigns forever, subject to the following:

- 1. This Deed is subject to and delivered pursuant to that certain Purchase and Sale Agreement dated [EXECUTION DATE], by and among Grantor and Grantee (as may be amended from time to time, the "Purchase and Sale Agreement"), and nothing in this Deed shall operate to limit, release, or impair any of Grantor's or Grantee's respective rights, obligations, remedies, or indemnities in the Purchase and Sale Agreement. Capitalized terms used in this Deed shall have the meanings prescribed in this Deed where such capitalized terms are defined; provided, however, that capitalized terms used in this Deed and not otherwise defined shall have the meanings given to such terms in the Purchase and Sale Agreement. Each defined term shall be equally applicable both to the singular and the plural forms of the term so defined. To the extent the terms and provisions of this Deed are in conflict, or inconsistent, with the terms and provisions of the Purchase and Sale Agreement shall control.
- 2. This Deed shall extend to, be binding upon, and inure to the benefit of the Parties and their respective successors and assigns.
- 3. Subject to the terms of the Purchase and Sale Agreement, from and after the date hereof, Grantee assumes and hereby agrees to pay, perform, fulfill and discharge (or cause to be paid, performed, fulfilled and discharged) the Assumed Liabilities; provided that the Assumed

Liabilities shall not include, and Grantor shall retain (and Grantee shall have no obligation with respect to any) the Retained Obligations.

- 4. Grantor, for a period of thirty-six (36) months after the date hereof, hereby warrants and defends Defensible Title to the Assets unto Grantee, its successors and assigns, against whomsoever lawfully claims the same or any part thereof, solely to the extent such claim arises by, through or under such Grantor but not otherwise. Grantee is hereby specifically assigned, and subrogated to, all warranties of title which Grantor may have from predecessors in interest (other than such Grantor or any Affiliate of such Grantor) to the extent applicable with respect to the Assets and to the extent such Grantor or such Affiliates may legally assign such rights and grant such subrogation.
- 5. EXCEPT AS OTHERWISE SET FORTH IN SECTION 3.2 AND ARTICLE 4 OF THE PURCHASE AND SALE AGREEMENT AND EXCEPT AS TO THE SPECIAL WARRANTY OF DEFENSIBLE TITLE UNDER THIS DEED, GRANTEE WILL ACCEPT THE ASSETS AT CLOSING IN THEIR PRESENT CONDITION, "AS IS AND WHERE IS AND WITH ALL FAULTS." GRANTEE ACKNOWLEDGES AND AGREES THAT, EXCEPT AS OTHERWISE SET FORTH IN THIS DEED AND ARTICLE 4 OF THE PURCHASE AND SALE AGREEMENT, GRANTOR HAS MADE NO REPRESENTATIONS OR WARRANTIES OF ANY KIND, EXPRESS OR IMPLIED, WRITTEN, ORAL, OR OTHERWISE, AS TO THE ACCURACY OR COMPLETENESS OF THE BACKGROUND MATERIALS OR ANY OTHER INFORMATION RELATING TO THE ASSETS FURNISHED BY OR ON BEHALF OF GRANTOR OR TO BE FURNISHED TO GRANTEE OR ITS REPRESENTATIVES, INCLUDING GRANTOR'S INTERNAL APPRAISALS AND INTERPRETIVE DATA.
- 6. THIS DEED AND THE LEGAL RELATIONS BETWEEN THE PARTIES SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS, WITHOUT GIVING EFFECT TO PRINCIPLES OF CONFLICTS OF LAWS THAT WOULD RESULT IN THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION, *PROVIDED THAT*, ANY DISPUTE PERTAINING TO REAL PROPERTY SHALL BE GOVERNED BY THE LAWS OF THE STATE OF LOUISIANA.
- 7. SUBJECT TO AND WITHOUT LIMITING SECTION 2.4(C) OF THE PURCHASE AND SALE AGREEMENT, THE PARTIES AGREE THAT ANY AND ALL DISPUTES OR CLAIMS BY ANY PARTY ARISING FROM OR RELATED TO THIS DEED THAT CANNOT BE AMICABLY SETTLED, WILL BE DETERMINED SOLELY AND EXCLUSIVELY BY ARBITRATION IN ACCORDANCE WITH THE FEDERAL ARBITRATION ACT AND USING THE RULES OF THE AMERICAN ARBITRATION ASSOCIATION OR ANY SUCCESSOR THEREOF WHEN NOT IN CONFLICT WITH SUCH ACT; PROVIDED, HOWEVER, THAT NOTHING CONTAINED IN THIS SECTION 7 SHALL LIMIT ANY PARTY'S RIGHT TO BRING (I) POSTARBITRATION ACTIONS SEEKING TO ENFORCE AN ARBITRATION AWARD OR (II) ACTIONS SEEKING INJUNCTIVE OR OTHER SIMILAR RELIEF UNDER SECTION 11.16 OF THE PURCHASE AND SALE AGREEMENT IN THE EVENT OF A BREACH OR THREATENED BREACH OF ANY OF THE PROVISIONS OF THIS DEED

(OR ANY OTHER AGREEMENT CONTEMPLATED HEREBY). IN FURTHERANCE OF THE FOREGOING, THE ARBITRATION PROCEDURES SET FORTH IN <u>SECTIONS 11.3(B)-11.3(C)</u> OF THE PURCHASE AND SALE AGREEMENT ARE INCORPORATED HEREIN BY REFERENCE, *MUTATIS MUTANDIS*, AS THOUGH SUCH PROCEDURES WERE SET FORTH HEREIN IN FULL.

- 8. TO THE EXTENT APPLICABLE TO THE ASSETS OR ANY PORTION THEREOF, GRANTEE HEREBY WAIVES THE PROVISIONS OF THE LOUISIANA UNFAIR TRADE PRACTICES AND CONSUMER PROTECTION LAW (LA. R.S. 51.1401, ET SEQ.). GRANTEE WARRANTS AND REPRESENTS THAT IT: (A) IS EXPERIENCED AND KNOWLEDGEABLE WITH RESPECT TO THE OIL AND GAS INDUSTRY GENERALLY AND WITH TRANSACTIONS OF THE TYPE CONTEMPLATED HEREIN AND IN THE PURCHASE AND SALE AGREEMENT SPECIFICALLY; (B) POSSESSES AMPLE KNOWLEDGE, EXPERIENCE AND EXPERTISE TO EVALUATE INDEPENDENTLY THE MERITS AND RISKS OF THE CONTEMPLATED TRANSACTIONS; AND (C) IS NOT IN A SIGNIFICANTLY DISPARATE BARGAINING POSITION.
- 9. The Exhibits to this Deed are hereby incorporated by reference and constitute a part of this Deed.
- 10. The Parties agree to execute, acknowledge and deliver to the other such further instruments and take such other action as may be reasonably requested in order to accomplish more effectively the purposes of the transactions contemplated by this Deed.
- 11. In the event any term of this Deed is found by any court to be void or otherwise unenforceable, the remainder of this Deed shall remain valid and enforceable, and, to the fullest extent permitted by Law, such offending term or terms shall be replaced with an enforceable term or enforceable terms that as nearly as possible effect the Parties' intent.
- 12. This Deed may be executed and delivered in one or more counterparts, each of which when executed and delivered shall be an original, and all of which when executed shall constitute one and the same instrument.

[Signature and Acknowledgment Pages Follow]

IN WITNESS WHEREOF, the Parties have executed this Deed on the date first set forth above, but this Deed shall be effective for all purposes as of the Effective Time.

	<b>GRANTOR:</b>		
	[SELLER]		
	Ву:		
	Name: Ryan Turner		
	Title: Managing Person	1	
	Witness:		
	Witness:		
STATE OF	§		
COUNTY OF	§ §		
This instrument was ack by Ryan Turner, known to me liability company, who affirms company and that the execution	ed that the foregoing instrum	of <b>[SELLER]</b> , a Defent was signed on b	laware limited behalf of such
	Notary Public	e in and for the State	of

	<u>GRAN</u>	TEE:		
	PHX N	MINERALS INC.		
		Kenna Clapp Director of Land		
	ACKNOV	<u>WLEDGMENT</u>		
STATE OF	§ § 8			
COUNTY OF	§			
This instrument was acknown by Kenna Clapp, known to me to corporation, who affirmed that the and that the execution of this instrument.	be the Dire e foregoing	ector of Land of PH instrument was sig	IX Minerals Inc., gned on behalf of s	an Oklahoma such company
		Notary Public in	and for the State of	of

## **EXHIBIT A-1**

#### **LANDS**

The following described Tracts, insofar and only insofar as the interest acquired by Grantor by virtue of the respective instrument listed below:

[To be updated pursuant to Article 3]

13.	Grantor:
Recor	tive Date:
14.	Grantor:
Recor	tive Date:
15.	Grantor:
Recor	tive Date:
16.	Grantor:
Recor Legal	t <b>ive Date:</b> ded: Description:
17.	Grantor:
Recor	tive Date:

Including without limitation Grantor's interest in the following described wells, insofar and only insofar as attributable to the Tracts:

# [To be updated pursuant to Article 3]

API	Well Name	Operator
1701734912	MAYFIELD 9 H; HA RA SUY #1	ENSIGHT IV ENERGY MGMT LLC
1701736213	PINEHILLS 9-4 HC; HA RA SUY #1-ALT	BLUE DOME OPERATING LLC
1701736434	PINEHILLS 9-4 HC; HA RA SUY #2-ALT	BLUE DOME OPERATING LLC
1701736435	PINEHILLS 9-4 HC; HA RA SUY #3-ALT	BLUE DOME OPERATING LLC
1701736436	PINEHILLS 9-4 HC; HA RA SUY #4-ALT	BLUE DOME OPERATING LLC
1701736437	PINEHILLS 9-4 HC; HA RA SUY #5-ALT	BLUE DOME OPERATING LLC

#### FIRST AMENDMENT TO CREDIT AGREEMENT

THIS FIRST AMENDMENT TO CREDIT AGREEMENT is dated and effective as of December 6, 2021, by and among PHX MINERALS INC., an Oklahoma corporation (referred to herein as the "Borrower"), each lender party hereto (collectively, the "Lenders" and individually, a "Lender"), and INDEPENDENT BANK, as Administrative Agent and L/C Issuer.

#### **RECITALS:**

- A. Borrower, Administrative Agent, and Lenders entered into the Credit Agreement dated as of September 1, 2021 (as amended and as further amended, restated, supplemented and/or otherwise modified prior to the date hereof, the "Existing Credit Agreement"), for the purpose and consideration therein expressed, whereby Lenders made loans to Borrower as therein provided; and
- B. Borrower, Administrative Agent and Lenders desire to amend the Existing Credit Agreement to modify same in accordance herewith;
- NOW, THEREFORE, in consideration of the premises and the mutual covenants and agreements contained herein and in the Existing Credit Agreement, in consideration of the loans and other extensions of credit which may hereafter be made by Lenders to Borrower, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto do hereby agree as follows:

#### ARTICLE I.

#### **DEFINITIONS AND REFERENCES**

- Section 1.1 <u>Terms Defined in the Existing Credit Agreement</u>. Unless the context otherwise requires or unless otherwise expressly defined herein, the terms defined in the Existing Credit Agreement shall have the same meanings whenever used in this Amendment.
- Section 1.2 Other Defined Terms. Unless the context otherwise requires, the following terms when used in this Amendment shall have the meanings assigned to them in this Section 1.2.
  - "Amendment" means this First Amendment to Credit Agreement.
  - "Amendment Documents" means this Amendment, *replacement Notes*, and all other Loan Documents executed and delivered in connection herewith. All of the Amendment Documents shall be deemed to constitute Loan Documents.
  - "Credit Agreement" (or "this Agreement" wherever referred to within the Existing Credit Agreement) means the Existing Credit Agreement as amended by this Amendment, and as the same may hereafter be further amended, restated, modified and/or otherwise supplemented from time to time.

#### ARTICLE II.

#### AMENDMENTS TO EXISTING CREDIT AGREEMENT

- Section 2.1 <u>Borrowing Base Notification</u>. From the date of this Amendment to, but excluding, the next redetermination of the Borrowing Base, as set forth in the Existing Credit Agreement, as amended hereby, the Borrowing Base shall be increased from Twenty-Seven Million Five Hundred Thousand Dollars (\$27,500,000.00) to Thirty-Two Million Dollars (\$32,000,000.00). The foregoing adjustment of the Borrowing Base constitutes the periodic redetermination of the Borrowing Base under Section 2.6(b) of the Existing Credit Agreement for December 1, 2021, and is not an Unscheduled Redetermination. Borrower, Administrative Agent and Lenders hereby agree that this provision satisfies all notification requirements as set forth in the Credit Agreement.
- Section 2.2 <u>Commitment Percentage</u>. The Committed Sum and Commitment Percentage of each Lender under the Revolver Facility are hereby amended to be as set forth in <u>Schedule 2.1</u> to this Amendment; and <u>Schedule 2.1</u> to the Existing Credit Agreement is hereby replaced by <u>Schedule 2.1</u> to this Amendment.

#### ARTICLE III.

#### CONDITIONS OF EFFECTIVENESS

- **Section 3.1** <u>Effective Date</u>. This Amendment shall become effective as of the date hereof when and only when:
- (a) <u>Amendment Documents</u>. Administrative Agent shall have received duly executed and delivered counterparts of each Amendment Document, *including replacement Notes*, in (i) in form, substance and date satisfactory to Administrative Agent and each Lender as required pursuant to the terms of the Credit Agreement, and (ii) in such numbers as Administrative Agent or its counsel may reasonably request.
- (b) <u>No Default</u>. No event shall have occurred and be continuing that would constitute an Event of Default or a Default.
- (c) <u>Increase Fee</u>. Borrower shall pay to Administrative Agent for the account of Lenders, to be divided based on each Lender's percentage of the increase in the Borrowing Base, an increase fee in the amount of \$27,000.00. Such fee shall be fully earned when paid and shall not be refundable for any reason whatsoever.

#### ARTICLE IV.

#### REPRESENTATIONS AND WARRANTIES

- **Section 4.1** Representations and Warranties of Borrower. In order to induce each Lender to enter into this Amendment, Borrower represents and warrants to Administrative Agent and each Lender that:
- (a) All representations and warranties made by Borrower in any Loan Document are true and correct in all material respects (without duplication of any materiality qualifier contained therein) on and as of time of the effectiveness hereof as if such representations and warranties had been made as of the time of the effectiveness hereof (except to the extent that such representation or warranty was made as of a specific date, in which case such representation or warranty shall be true and correct in all material respects (without duplication of any materiality qualifier contained therein) as of such specific date).
- (b) Borrower has duly taken all corporate action necessary to authorize the execution and delivery by it of the Amendment Documents to which it is a party and to authorize the consummation of the transactions contemplated thereby and the performance of its obligations thereunder.
- which it is a party, the performance by it of its obligations under such Amendment Documents, and the consummation of the transactions contemplated by such Amendment Documents, do not and will not (a) conflict with, violate or result in a breach of any provision of (i), to its knowledge, any Law, (ii) its Organizational Documents, or (iii) any material agreement, judgment, license, order or permit applicable to or binding upon it, (b) result in the acceleration of any Indebtedness owed by it, or (c) result in or require the creation of any Lien upon any of its assets or properties except as expressly contemplated or permitted in the Loan Documents. Except (x) as expressly contemplated in the Amendment Documents and (y) such as have been obtained or made and are in full force and effect, to its knowledge, no permit, consent, approval, authorization or order of, and no notice to or filing with, any Governmental Authority or third party is required on the part of or in its respect in connection with the execution, delivery or performance by it of any Amendment Document or to consummate any transactions contemplated by the Amendment Documents.
- (d) This Amendment is, and the other Amendment Documents when duly executed and delivered will be, legal, valid and binding obligations of it, enforceable against it in accordance with their terms except as such enforcement may be limited by bankruptcy, insolvency or similar Laws of general application relating to the enforcement of creditors 'rights and by general principles of equity.

#### ARTICLE V.

#### MISCELLANEOUS

- **Section 5.1** <u>Ratification and Affirmation</u>. Borrower hereby acknowledges the terms of the Existing Credit Agreement, as amended, and ratifies and affirms its obligations under, and acknowledges, renews and extends its continued liability under, each Loan Document to which it is a party and agrees that each Loan Document to which it is a party remains in full force and effect.
- Section 5.2 <u>Survival of Agreements</u>. All of Borrower's various representations, warranties, covenants and agreements in the Amendment Documents shall survive the execution and delivery thereof and the performance thereof, including the making or granting of the Loans and the delivery of the other Loan Documents, and shall further survive until all of the Obligations are paid in full to each Lender and all of Lenders' obligations to Borrower are terminated.
- **Section 5.3** <u>Authorization</u>. The Lenders hereby authorize the Administrative Agent to execute any and all amendments to any Loan Documents deemed necessary by Administrative Agent to evidence the extension of the term of the Loan as described herein.
- **Section 5.4** <u>Interpretive Provisions</u>. Sections 1.2 and 1.3 of the Existing Credit Agreement are incorporated herein by reference herein as if fully set forth.
- Section 5.5 <u>Loan Documents</u>. The Amendment Documents are each a Loan Document, and all provisions in the Existing Credit Agreement pertaining to Loan Documents apply thereto.
- **Section 5.6** Governing Law. This Amendment shall be governed by, and construed in accordance with, the Laws of the State of Texas.
- **Section 5.7** <u>Counterparts; Scans.</u> This Amendment may be separately executed in counterparts and by the different parties hereto in separate counterparts, each of which when so executed shall be deemed to constitute one and the same Amendment. The Amendment Documents may be validly executed by facsimile, scan, or other electronic transmission.

THIS AMENDMENT AND THE OTHER AMENDMENT DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES, THERE ARE NO UNWRITTEN ORAL AGREEMENTS OF THE PARTIES.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed as of the date first above written.

[The remainder of this page has been intentionally left blank.]

# **Signature Page to First Amendment to Credit Agreement**

**BORROWER:** 

### PHX MINERALS INC.,

an Oklahoma corporation

By: /s/ Chad L. Stephens
Name: Chad L. Stephens

President and Chief Executive Officer Title:

### **Signature Page to First Amendment to Credit Agreement**

**ADMINISTRATIVE AGENT:** 

#### INDEPENDENT BANK

By: /s/ Philip M. Mortimer
Name: Philip M. Mortimer, CFA

Vice President Title:

LENDER: INDEPENDENT BANK

By: /s/ Philip M. Mortimer
Name: Philip M. Mortimer, CFA

Vice President Title:

### **Exhibits and Schedules:**

Schedule 2.1 – Lenders, Addresses, Commitment Percentages, and Committed Sums

# **Signature Page to First Amendment to Credit Agreement**

LENDER: MIDFIRST BANK

By: /s/ W. Thomas Portman

Name: W. Thomas Portman
Title: Vice President

### SCHEDULE 2.1 TO LOAN AGREEMENT

### COMMITTED SUMS AND COMMITMENT PERCENTAGES

Lender	<b>Committed Sums</b>	Commitment Percentage	Note Amount
Independent Bank	\$20,000,000.00	62.50000%	\$62,500,000.00
MidFirst Bank	\$12,000,000.00	37.50000%	\$37,500,000.00
TOTAL	\$32,000,000.00	100.000000000%	\$100,000,000.00



FOR IMMEDIATE RELEASE PLEASE CONTACT: Chad L. Stephens 405.948.1560

Website: www.phxmin.com

#### PHX Minerals Inc. Announces Increase to Borrowing Base and Acquisition in the Haynesville

OKLAHOMA CITY, Dec. 9, 2021 /PRNewswire/ -- PHX Minerals Inc. (NYSE: PHX) ("PHX" or the "Company") today announced that it has entered into an amendment to its revolving credit facility ("Credit Facility") under which the borrowing base of the Credit Facility has been increased from \$27.5 million to \$32.0 million in connection with its regularly scheduled semi-annual redetermination. Additionally, PHX has entered into purchase and sale agreements to acquire 426 total net royalty acres in Caddo Parish, La., for \$5,787,272 in cash from two private sellers, subject to customary closing adjustments (the "Acquisition"). The Acquisition will be funded with a combination of cash on hand and borrowings under the Company's Credit Facility. Pro forma the Acquisition, the Company's total debt drawn on the Credit Facility will be approximately \$20.0 million. The Board of Directors of PHX unanimously approved the Acquisition, which is expected to close by Dec. 15, 2021.

Chad Stephens, President and CEO, said, "the Acquisition assets provide almost immediate cash flow from the wells being completed and do a nice job of helping replace the volumes associated with the previously announced divestiture of some of our legacy working interest wellbores. The increase in the borrowing base of our Credit Facility is a reflection of a much stronger PHX with an improving asset base and high-quality prospects."

#### **Acquisitions Highlights**

- Approximately 426 net royalty acres in Caddo Parish, La., focused on the Hayneville play;
- Assets are operated by Blue Dome and include 4 PDP gross wells, 4 gross wells in progress that are expected to come online in the next 30 days and an estimated 4 gross undrilled locations;
- Estimated reserves of 3.8 Bcfe<sup>1</sup>;
- Current net production 0.15 Mmcfe/d<sup>2</sup>; and
- Estimated net production for 2022 totaling between 1,700 to 2,100 Mcfe/d, comprised of 100% natural gas.

**PHX Minerals Inc.** (NYSE: PHX) Oklahoma City-based, PHX Minerals Inc. is a natural gas and oil mineral company with a strategy to proactively grow its mineral position in its core areas of focus. PHX owns approximately 251,000 net mineral acres principally located in Oklahoma, Texas, Louisiana, North Dakota, and Arkansas. Additional information on the Company can be found at www.phxmin.com.

<sup>1</sup> As of 11/01/2021

<sup>&</sup>lt;sup>2</sup> Mmcfe/d is on a 6:1 basis; estimated November 2021 production

#### **Cautionary Statement Regarding Forward-Looking Statements**

This press release includes "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. Words such as "anticipates," "plans," "estimates," "believes," "expects," "intends," "will," "should," "may" and similar expressions may be used to identify forward-looking statements. Forward-looking statements are not statements of historical fact and reflect PHX's current views about future events. Forward-looking statements may include, but are not limited to, statements relating to: PHX's ability to execute its business strategies; the volatility of realized natural gas and oil prices; the level of production on its properties; estimates of quantities of natural gas, oil and NGL reserves and their values; general economic or industry conditions; legislation or regulatory requirements; conditions of the securities markets; the ability to raise capital; changes in accounting principles, policies or guidelines; financial or political instability; acts of war or terrorism; title defects in the properties in which PHX invests; and other economic, competitive, governmental, regulatory or technical factors affecting our properties, operations or prices. Although the Company believes its expectations reflected in these and other forward-looking statements are reasonable, the Company can give no assurance its expectations will prove to be correct. Such forward-looking statements are subject to a number of assumptions, risks and uncertainties, many of which are beyond the control of the Company. These forward-looking statements involve certain risks and uncertainties that could cause the results to differ materially from those expected by the Company's management. Information concerning these risks and other factors can be found in the Company's filings with the Securities and Exchange Commission, including its Annual Reports on Form 10-K and Quarterly Reports on Form 10-Q, available on the Company's website or the SEC's website at www.sec.gov.

Investors are cautioned that any such forward-looking statements are not guarantees of future performance and that actual results or developments may differ materially from those projected in forward-looking statements. The forward-looking statements in this press release are made as of the date hereof, and the Company does not undertake any obligation to update the forward-looking statements as a result of new information, future events or otherwise.

CONTACT: Chad L. Stephens, 405.948.1560, Website: www.phxmin.com